Three Essays on the Electricity Market, Distributed Generation, and Retail Tariff Designs

by

Yiang Guo

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

Department of Economics University of Alberta

© Yiang Guo, 2024

Abstract

The integration of distributed energy resources (DERs) such as solar panels and batteries is crucial for the modernization and sustainability of the electricity grid. These technologies can significantly enhance grid resilience and flexibility by diversifying energy sources and reducing reliance on fossil fuels. The rapid expansion of solar and battery technologies in the past decade, driven by falling costs and supportive policies, underscores their potential to transform the electricity system. Understanding how to integrate these DERs into the electricity grid is of great importance. This thesis aims to explore the barriers to solar and battery integration and investigate how economic mechanisms, such as retail rate designs, can address these challenges and facilitate a smoother transition to a more sustainable energy system.

The first chapter of this thesis investigates whether we can use real-time pricing (RTP) and carbon taxes to reduce carbon emissions associated with battery systems. Behind-the-meter (BTM) or customer-sited battery systems can perform energy arbitrage based on the retail rate they face. Under typical time-of-use pricing, batteries will generally increase emissions because the price signal it faces does not strongly correlate with the emissions signal in the jurisdictions we consider. By incorporating RTP and carbon taxes into the retail tariff structure, we show that batteries can effectively reduce carbon emissions, as the emissions avoided through battery discharge can exceed the emissions incurred from charging and energy losses. However, RTP and carbon taxes will also decrease the private financial value of a solar plus storage system, leading to lower investment levels.

The second chapter of this thesis looks at how transmission congestion will affect locational marginal emissions factors (MEFs) and locational environmental values of solar and batteries. MEFs measure the increase in carbon emissions if the electricity demand increases by 1 MWh. We use a regime-switching model to estimate the MEFs with or without congestion and use the congestion-weighted MEFs to calculate the environmental value of solar and batteries. We find that the emissions reductions from a rooftop solar panel located in a renewablerich region will be reduced substantially when congestion occurs frequently. In contrast, a battery can utilize variations in MEFs with and without congestion, leading to higher emissions reduction potentials.

The third chapter of this thesis investigates whether retail electricity customers will adjust their maximum demand (i.e., highest instantaneous electricity usage) in response to changes in the level of maximum demand charges (MDCs), which charge customers based on their maximum demand rather than kWh usage. MDCs for have been proposed by a number of utilities as a way to recover fixed costs when more customers are self-generating electricity using rooftop solar rather than purchasing power from the utilities. The proponents of MDCs claim that MDCs can motivate customers to reduce peak demand and, therefore, reduce utility expenditures on grid upgrades. Our study finds that a change in the level of MDCs can have a statistically significant and modest effect on the peak demand for medium-large consumers (e.g., a medium-sized grocery store).

There are several policy implications of this thesis. First, new retail tariff designs are essential to integrate rooftop solar and batteries into the electricity market. For example, RTP and carbon taxes can be used to reduce storage-induced carbon emissions and alleviate distortions that exist in prevailing rate structures. Second, the environmental values of solar and batteries have substantial locational variations. Therefore, policies that facilitate these technologies should take into account the locational environmental value. Third, MDCs can be used to reduce the maximum demand for the electrical system, but more evidence is needed to understand their overall effectiveness, especially for large customers.

Preface

This dissertation consists of three empirical articles. I was responsible for the empirical analysis, research methodology development, and manuscript writing. My supervisors were involved in motivation, research methodology development, and manuscript editing.

Acknowledgements

I would like to express my deepest gratitude to my supervisors Dr. David Brown, Dr. Andrew Eckert, and my dissertation committee member Dr. Heather Eckert. Without their continuous inspiration and encouragement, this dissertation could not have been finished. I am very thankful to Dr. Xuejuan Su, Dr. Joseph Marchand, Dr. Martin Alfaro, and Dr. Brendon Andrews, for their insightful comments and suggestions.

Meeting my supervisors has been a fortunate occurrence in my life, as their unwavering support, mentorship, and attention to my education and career have been vital throughout my PhD studies. They are the most exceptional supervisors I have ever encountered, and I am truly grateful for everything I have gained under their guidance.

Furthermore, I would like to express my gratitude to my fellow students Liang Ma, Boris Ortega Moreno, and Asrafuzzaman for their companionship and friendship during this journey.

Lastly, I am indebted to my parents, Nanchu Guo and Ou Yang, as well as my wife, Ying Zhang, for their unconditional love and unwavering support.

Table of Contents

| 1 | The | Emissions Impacts of Behind-the-Meter Battery Storage: The Role | |
|---|------|---|---|
| | of R | teal-Time Pricing | 1 |
| | 1.1 | Introduction | 1 |
| | 1.2 | Real-Time Pricing and Carbon Pricing | 9 |
| | 1.3 | Economic Mechanism | 0 |
| | 1.4 | Data 1 | 4 |
| | 1.5 | Methodology 1 | 6 |
| | | 1.5.1 Optimization Program | 8 |
| | | 1.5.2 DER Technologies | 9 |
| | | 1.5.3 DER Capacity | 0 |
| | | 1.5.4 Marginal Emissions Factor | 1 |
| | | 1.5.5 Tariff Design and Solar Compensation Policy | 6 |
| | | 1.5.6 Summary of Methodology 3 | 2 |
| | 1.6 | Results | 4 |
| | | 1.6.1 Exogenous Results | 4 |
| | | 1.6.2 Endogenous Results | 3 |

| | 1.7 | Robustness Checks | 47 |
|---|--------------|---|-----|
| | | 1.7.1 Price Elasticity | 47 |
| | | 1.7.2 "Real-time" RTP | 51 |
| | 1.8 | Conclusion | 53 |
| | Refe | rences | 56 |
| | 1.9 | Appendices | 62 |
| 2 | Esti gest | mating Locational Marginal Emissions Considering Transmission Con- ion | 69 |
| | 2.1 | Introduction | 69 |
| | 2.2 | Conceptual Framework | 75 |
| | | 2.2.1 Congestion and Locational MEFs | 75 |
| | | 2.2.2 The Environmental Values of DERs | 78 |
| | 2.3 | Texas's Electricity Market | 80 |
| | 2.4 | Data | 84 |
| | 2.5 | Methodology | 86 |
| | | 2.5.1 Congestion | 86 |
| | | 2.5.2 Empirical Strategy | 90 |
| | 2.6 | Results | 100 |
| | | 2.6.1 ERCOT-Wide Marginal Emissions | 100 |
| | | 2.6.2 Locational Marginal Emissions | 106 |
| | 2.7 | Locational Environmental Values | 109 |
| | 2.8 | Conclusion | 114 |
| | Refe | rences | 117 |
| | 2.9 | Appendices | 122 |

| 3 | Con | npetin | g Perspectives of Maximum Demand Charges | 132 |
|---|------|---------|--|-----|
| | 3.1 | Introd | luction | 132 |
| | 3.2 | Review | w of studies of MDC | 137 |
| | | 3.2.1 | System Peak Reduction | 138 |
| | | 3.2.2 | Cost-Causation | 143 |
| | | 3.2.3 | Distributional Effects | 145 |
| | 3.3 | Review | w of Empirical Studies on MDC Response | 148 |
| | 3.4 | Backg | round and Data of the Case Study | 153 |
| | | 3.4.1 | Case Study Objective | 153 |
| | | 3.4.2 | Distribution Tariffs in Lethbridge | 155 |
| | | 3.4.3 | Data | 157 |
| | | 3.4.4 | Effect of MDC Rate on Customer Electricity Bills | 159 |
| | 3.5 | Metho | odology | 161 |
| | | 3.5.1 | Summer-Peak vs Non-Summer-Peak Group | 164 |
| | | 3.5.2 | The Machine Learning Approach | 165 |
| | | 3.5.3 | Before-After Comparison | 170 |
| | 3.6 | Result | S | 173 |
| | | 3.6.1 | Model Performance | 173 |
| | | 3.6.2 | Before-After Maximum Demand Comparison | 174 |
| | 3.7 | Concl | usion | 181 |
| | Refe | erences | | 185 |
| | 3.8 | Apper | ndices | 192 |
| | | | | |

Bibliography

List of Tables

| 1.1 | Exogenous Capacity for Solar PV and Battery Storage | 21 |
|------|---|-----|
| 1.2 | Summary of Variations | 32 |
| 1.3 | Environmental Outcomes | 39 |
| 1.4 | Decomposition of Storage-induced Carbon Emissions | 40 |
| 1.5 | Endogenous Results for CAISO under Different Scenarios | 44 |
| 1.6 | Endogenous Results for ERCOT under Different Scenarios | 46 |
| 1.7 | Elasticity-Adjusted Environmental Outcomes for CAISO | 49 |
| 1.8 | Elasticity-Adjusted Endogenous Results for CAISO | 50 |
| 1.9 | Summary Statistics of the RTP | 52 |
| 1.10 | Environmental Outcomes Comparison | 53 |
| 1.11 | Notation Description | 64 |
| 1.12 | Retail Tariff Designs | 66 |
| 2.1 | Summary Statistics | 87 |
| 2.2 | Effect of Congestion on ERCOT-Wide Marginal Emissions - OLS Estimates | 103 |
| 2.3 | Effect of Congestion on ERCOT-Wide Marginal Emissions - IV Estimates | 104 |
| 2.4 | Annual Carbon Mitigation of DER Technologies | 113 |

| 2.5 | Regional Average MEFs (Congested vs. Uncongested) | 127 |
|-----|---|-----|
| 3.1 | Review of Empirical Studies of Demand Charge Response | 149 |
| 3.2 | Summary Statistics of Electricity Demand | 161 |
| 3.3 | Summary of MAE and MAPE | 174 |
| 3.4 | Panel Fixed Effects Results - 995 Customers | 176 |
| 3.5 | Panel Fixed Effects Results - 996 Customers | 178 |
| 3.6 | Placebo Test - 995 and 996 Customers | 180 |
| 3.7 | Rate Classes in Lethbridge | 192 |
| 3.8 | Daily Maximum Temperature vs. Daily Maximum Demand | 195 |
| 3.9 | Fixed Effects Approach Results | 198 |

List of Figures

| 1.1 | Hypothetical Dispatch Curve During the Middle of the Day | 13 |
|------|--|----|
| 1.2 | Hypothetical Dispatch Curve During the Evening Ramps | 13 |
| 1.3 | Month-Hour MEFs for CAISO (kg CO_2 per kWh) | 25 |
| 1.4 | Month-Hour MEFs for ERCOT (kg CO_2 per kWh) $\ldots \ldots \ldots \ldots \ldots$ | 26 |
| 1.5 | Retail Tariff Designs | 31 |
| 1.6 | Battery Charge and Discharge Decisions | 37 |
| 1.7 | Carbon Reduction under the RTP by Month | 41 |
| 1.8 | Demand Profile for Supermarkets | 65 |
| 1.9 | CAISO Average Weekday LMPs (\$/kWh) | 67 |
| 1.10 | CAISO Average Weekend LMPs ($/kWh$) | 67 |
| 1.11 | ERCOT Average Weekday LMPs ($/kWh$) | 68 |
| 1.12 | ERCOT Average Weekend LMPs ($/kWh$) | 68 |
| 2.1 | Numerical Example Setting | 76 |
| 2.2 | Numerical Example Illustration | 77 |
| 2.3 | Generation and Transmission Infrastructures in ERCOT Load Zones | 83 |
| 2.4 | Congestion Occurrence by Hour | 89 |

| 2.5 | Congestion Decomposition by Load Zones | 90 |
|------|---|----|
| 2.6 | Hourly ERCOT-Wide MEFs during Uncongested and Congested Hours 1 | 05 |
| 2.7 | MEFs in the NORTH Zone During Uncongested and Congested Hours 1 | 07 |
| 2.8 | MEFs in the South Zone During Uncongested and Congested Hours 1 | 08 |
| 2.9 | MEFs in the West Zone During Uncongested and Congested Hours 1 | 09 |
| 2.10 | Congestion-Weighted MEFs for Each Load Zone | 12 |
| 2.11 | Generation Capacities (MW) by Technology | 22 |
| 2.12 | Electricity Generations (MWh) by Technology | 23 |
| 2.13 | Emission Intensities for West Generators | 24 |
| 2.14 | Emission Intensities for North Generators | 25 |
| 2.15 | Emission Intensities for South Generators | 25 |
| 2.16 | Selected Weather Stations | 30 |
| 2.17 | Percentage of Congested Hours by Year | 31 |
| 3.1 | MDC Rates and Corresponding Monthly Electricity Bills | 60 |
| 3.2 | Distribution of Percentage Bill Changes Resulting from the MDC Rate Change1 | 62 |
| 3.3 | Distribution of Summer-Peak Ratio for 995 and 996 Customers | 66 |
| 3.4 | Relationship Between Temperature and Demand for a Summer-Peak Customer1 | 93 |
| 3.5 | Load Profile of a Random 995 Customer | 99 |
| 3.6 | 995 Customers Annual Peaks by Month | 00 |
| 3.7 | 996 Customers Annual Peaks by Month | 01 |
| 3.8 | Daily System Peaks by Hour | 01 |
| 3.9 | 995 Customers Daily Peaks by Hour | 02 |
| 3.10 | 996 Customers Daily Peaks by Hour | 02 |

Chapter 1

The Emissions Impacts of Behind-the-Meter Battery Storage: The Role of Real-Time Pricing

1.1 Introduction

Climate change poses a fundamental and urgent threat to human society (IPCC 2022). Due to more strict environmental regulations and falling costs, renewable energy resources such as wind and solar photovoltaic (PV) have expanded rapidly worldwide (REN21 2024).¹ However, wind and solar PV only generate electricity when the sun shines or the wind blows. This intermittent nature of wind and solar presents a significant challenge to their penetration into the energy sector (Rugolo and Aziz 2012). To accommodate wind and solar, balancing sources, such as battery storage systems, will be needed to balance the fluctuations in electricity supply and demand caused by renewable intermittency.

To promote the adoption of battery storage technologies, a number of policies have been implemented to subsidize customer-sited battery storage technologies (also known as behindthe-meter or BTM battery systems), including California's Self-Generation Incentive Pro-

^{1.} In 2023, wind and solar PV capacities increased by 119 GW and 354 GW worldwide, representing more than 90% of all renewable power additions (REN21 2024).

gram (SGIP) and New York's Reforming the Energy Vision (REV) program (Astoria 2017; CPUC 2023). However, it has been argued that these policies fail to achieve one of the fundamental goals: cutting GHG emissions (CPUC 2017; 2018a). In particular, the deployment of both utility-scale and BTM battery storage systems increases emissions almost everywhere in the US (Hittinger and Azevedo 2015; Fisher and Apt 2017).

Storage-related emissions are determined by three factors: created emissions from charging, avoided emissions from discharging, and energy loss from discharging and discharging (Arciniegas and Hittinger 2018). The first two factors are related to the electricity price signals because batteries are often used to perform energy arbitrage: charge when the energy price is low and discharge when the energy price is high. The energy loss is determined by the technical features of the battery as well as the frequency of charge and discharge. For BTM batteries, the increase in carbon emissions is mainly driven by energy loss (Fisher and Apt 2017). However, with updated battery technology that minimizes energy loss, energy arbitrage can also be an important factor that increases storage-induced emissions.² In particular, energy arbitrage can result in a substantial emission increase in regions that are heavily relying on coal-fired generators (Holland and Mansur 2008). In this case, batteries are often charged at night with cheap but polluting coal-fired energy and discharged when cleaner natural gas energy is used to meet peak demand.

To address storage-induced carbon emissions, the California Public Utilities Commission (CPUC) issued a mandate to reduce storage-induced carbon emissions.³ In response, the

^{2.} For example, one of the most popular BTM batteries, Tesla's PowerWall, has a round-trip efficiency of 90% as of 2023, meaning you can use 90% of the energy you put into the battery as energy output (Smartly Energy 2023). In contrast, some higher-end BTM batteries can achieve a round-trip efficiency of 96% at higher costs (Smartly Energy 2023).

^{3.} In 2018, the CPUC revised its incentive policy for new storage projects under its SGIP (CPUC 2018b). The new policy requires battery owners to reduce carbon emissions by at least 5 kg per kWh of battery in order to recoup the full payment of the SGIP (CPUC 2018b). Under the new policy, each storage project will be reviewed semi-annually to ensure that the project meets the emissions requirements. Without the mandate, a battery system will emit 50 kg per kWh of battery on average (CPUC 2018b). For a medium-sized battery with 100 kWh, carbon emissions, on average, will be reduced by 5.5 metric tons. In comparison, annual carbon emissions for a typical passenger vehicle are 4.6 metric tons (EPA 2023b).

California Solar and Storage Association (CALSSA) has advocated the use of a marketbased mechanism to reduce storage-induced emissions (PV Magazine, 2020). In particular, CALSSA claimed that storage-induced emissions could be reduced substantially by transitioning from the current time-of-use⁴ (TOU) pricing to real-time pricing (RTP). RTP refers to a retail structure that allows the retail electricity price to change very frequently (e.g., hourly) according to the supply/demand balance of the current electricity market (Borenstein 2005). According to CALSSA, the transition from the current TOU design, which charges different prices for peak and low-peak periods, to an RTP based on the wholesale electricity price that varies more frequently, can induce BTM batteries to charge more often from green energy. The stored green energy can then be used to displace fossil fuel generation and reduce carbon emissions. Further, an RTP can increase bill savings and motivate battery investment by providing more arbitrage opportunities than the TOU rate.

Economists have long advocated RTP as it can help avoid deadweight loss associated with electricity consumption and reduce market power (Holland and Mansur 2008). However, whether it can bring additional environmental benefits for batteries is still an open question. Our paper aims to answer how RTP will affect the overall environmental outcome of battery systems coupled with solar. In particular, our paper focuses on two RTP designs: a RTP based on the wholesale electricity price and a RTP based on both the wholesale electricity price and a RTP based on both the wholesale electricity price and a carbon price. The RTP plus carbon price case is interesting because it can achieve the socially optimal operation of batteries by ensuring that the time-varying social marginal cost of electricity production is reflected in the electricity rates.⁵

Under the prevalent compensation policy, the compensation rate for solar generation is closely tied to the prevailing retail prices. Because current retail prices are usually well above

^{4.} Under a TOU rate, customers will pay varying rates throughout the day, within pre-set time periods, based on the cost of providing electricity (Faruqui and George 2005).

^{5.} Section 1.2 provides a detailed discussion on how RTP and carbon pricing can enhance the overall efficiency of electricity markets.

the wholesale rates that RTP is based upon, implementing RTP could significantly reduce the private financial values of solar panels.⁶ Thus, a move to RTP can reduce investment in solar plus storage systems that have the potential to reduce carbon emissions. Overall, RTP and carbon taxes can have ambiguous effects on the environmental outcomes of a solar plus battery system. A formal analysis is needed to quantify these effects.

In this paper, we simulate the impact of RTP and carbon pricing on carbon emissions resulting from battery systems. We compare the environmental outcomes between the RTP plus carbon pricing tariff design and several alternative retail tariff designs, including two representative TOU rates and a standard RTP without a carbon tax. Using an optimization algorithm called the DER-CAM (Distributed Energy Resources Consumer Adoption Model), we simulate the optimal operation and investment decisions for solar plus storage systems under RTP and alternative retail rate designs. In particular, we look at two cases: a case with exogenous solar and battery capacities and a case where we endogenously solve for the optimal capacities of solar and batteries. The exogenous case allows us to focus on emissions resulting from the charge and discharge decisions of batteries. The endogenous case investigates the environmental effects of solar plus storage systems in the long run, allowing for flexibility in both the operation and investment decisions.

To understand how battery operation will affect emissions, we use generation and emissions data to measure marginal emissions factors (MEFs) at different hours of the day and months of the year. The MEF captures the change in emissions for a 1 MWh change in demand. The MEFs can be used to calculate the changes in emissions resulting from changes in electricity demand caused by battery charging and discharging.

We examine the effects of RTP on two different regions to understand how the results vary with different generation mixes. Specifically, we focus on two independent system operators:

^{6.} Some jurisdictions proposes to use a RTP plus a markup

The Electric Reliability Council of Texas (ERCOT) and The California Independent System Operator (CAISO). These two regions are interesting because they have a high penetration of renewable energy resources but different renewable portfolios. In particular, ERCOT has considerable wind capacity, while CAISO is abundant in solar energy resources and has a growing portfolio of wind generators.⁷ Natural gas-fired plants generally meet marginal electricity demand in both jurisdictions, with the natural gas fleet being relatively more efficient and clean in CAISO (Potomac Economics 2023b; CAISO 2023a). While coal-fired generators play a limited role in CAISO, they are still relied upon in ERCOT for generation at night to meet marginal demand (Potomac Economics 2023b).

Our paper focuses on large commercial customers, who are the primary adopters of the BTM battery system in terms of system capacity (CPUC 2018a).⁸ In addition, commercial customers, along with industrial customers, are usually the primary targets for the RTP (Borenstein 2005).⁹ Throughout the analysis, our paper uses the demand profile of super-markets as an example, but our findings are robust to other types of commercial customers.

In the short run, we find that RTP, absent carbon pricing, could lead to lower carbon emissions compared to the TOU rates, but the reductions are not substantial. While RTP can reduce carbon emissions through energy arbitrage, it also increases the use of batteries, leading to higher energy losses during charge and discharge.¹⁰ In line with previous studies e.g., Hittinger and Azevedo 2015; Fisher and Apt 2017, we find that adding batteries to a solar system often increases carbon emissions under both RTP and TOU rates. Energy losses

^{7.} Solar capacity in ERCOT is rapidly expanding. In 2021, solar capacity account for 40% of all capacity additions in ERCOT (ERCOT 2022). However, solar production in ERCOT remains relatively limited during our sample period.

^{8.} Non-residential customers account for 14% of battery storage projects but 81% of battery storage capacity in California (CPUC 2018a). In particular, the CPUC highlights four types of commercial customers as the leading adopters of battery storage systems: supermarkets, hotels, offices, and schools (CPUC 2018a).

^{9.} RTP usually targets large commercial and industrial customers because the benefits from load shifting can usually outweigh the technology costs of meter upgrades (Borenstein 2005).

^{10.} The battery energy losses can be measured by the round-trip efficiency (RTE) of the battery, which is the percentage of electricity charged into the battery that can be retrieved for discharge (Fisher and Apt 2017).

are the primary source of storage-induced emissions in both cases. Therefore, the round-trip efficiency of the battery, which determines the magnitude of energy loss, plays a crucial role in storage-induced emissions.

Combining TOU with carbon taxes will not affect storage-induced carbon emissions, as it does not influence battery charge and discharge decisions.¹¹ However, we show that combining RTP with a sufficiently stringent carbon price can lead to emissions reduction from solar plus battery systems compared to a solar-only system because the emissions reductions from energy arbitrage outweigh the energy losses from battery operations. Emission reductions are more substantial in Texas, as its energy mix (with wind and coal) provides a higher MEF difference between charge and discharge. Through arbitrage, dirtier coal-fired generation is displaced by cleaner wind generation, leading to substantial emissions reduction. In addition, there are substantial seasonal effects on how RTP plus carbon taxes will affect storage-induced carbon emissions. In particular, RTP plus carbon taxes induce considerably more emissions reductions during early spring in CAISO and summer in ERCOT with high solar and wind generation.

In the long run, we find that RTP is generally insufficient to promote the adoption of solar plus storage technologies. The average rate of RTP is considerably lower than alternative rates (e.g., TOU rates). Therefore, purchasing electricity directly from the utility becomes more cost effective than self-generating electricity. As solar is compensated at the retail rate, the private financial value of solar reduces, and consumers are less motivated to install solar PV systems. As a result, under the current compensation policy, TOU rates often

^{11.} Under RTP, carbon taxes can be passed on to customers based on the real-time emissions signals. However, constrained by the retail tariff structure, we assume that carbon taxes will be passed on to customers according to peak and off-peak periods under the TOU. As such, customers will pay an additional charge (\$ per kWh) for carbon taxes that vary by periods, reflecting the average MEFs specific to that period. Consequently, carbon taxes will not alter the TOU structure, but will change the peak/off-peak rates and the rate differences across periods. Since batteries will always charge during off-peak periods and discharge during peak periods, carbon taxes will not impact storage charge and discharge decisions under TOU. See the subsection on "Carbon Taxes" in Section 1.5.5 for a detailed discussion.

outperform RTP regarding emissions reductions when investment decisions are considered. However, this is because TOU rates are over-compensating solar customers and will lead to inefficiently high solar investments.¹² We find that consumers facing RTP are willing to install solar PV systems when the costs of solar are reduced by at least 20%. In addition, the costs of batteries need to be reduced by 50% to 60% until energy arbitrage becomes profitable under the RTP plus carbon tax case.¹³ Solar and battery adoption under RTP could be achieved through substantial government subsidies, reductions in solar and battery costs, or an increase in the social cost of carbon.

Our study extends the existing literature in three ways. First, our study evaluates the role of a socially optimal retail tariff design (RTP plus carbon tax) in the operation and investment of batteries from an environmental perspective. Our paper is the first to analyze such tariff design with BTM battery storage, whereas previous studies mainly focus on existing retail tariffs, such as demand charges¹⁴ and TOU pricing.¹⁵ For example, Fisher and Apt (2017) simulates the environmental outcomes of batteries under TOU rates in different regions, highlighting the fact that BTM batteries may increase emissions under existing tariffs in most jurisdictions. In comparison, we focus on RTP and carbon tax combinations because they represent the most efficient retail tariff designs and serve as the first-best solution to address environmental externalities. Specifically, we aim to assess the environmental effects of this socially optimal retail tariff design within a framework

^{12.} Felder and Athawale (2014) discusses why compensating solar at retail price is over-generous for solar customers. Brown and Sappington (2017b) shows that compensating solar at the prevailing retail electricity rate will reduce aggregate welfare. Borenstein (2017) shows that the solar compensation rate is much higher than the avoided costs, and compensating solar based on avoided costs will significantly reduce the private financial value of solar.

^{13.} A number of studies also show that the compensation for storage is not sufficient to justify the cost. For example, Muchlenbachs and Brown (2023) show that a very high benefit of avoiding outages is needed to justify the current expense of batteries.

^{14.} Demand charges are charged based on the maximum amount of power consumed during a short time interval during a billing period (Berg and Savvides 1983).

^{15.} Hittinger and Azevedo (2015) examines utility-scale batteries that can perform energy arbitrage based on wholesale electricity rates, which is similar to the RTP in our study. The key difference between our study and that of Hittinger and Azevedo (2015) is that we focus on BTM batteries and compare RTP with other retail tariff designs.

characterized by the rapid expansion of renewable energy. Our results shed light on the ongoing policy discussions regarding retail tariff reforms and how to use market mechanisms to reduce carbon emissions.¹⁶

Second, our study looks at regions with high renewable penetration, and we estimate the associated MEFs considering renewables. Previous studies (e.g., Hittinger and Azevedo 2015; Fisher and Apt 2017) examining the environmental outcomes of batteries generally assume that renewable energy resources will not be used to meet marginal demand, and thus do not affect MEFs. Consequently, batteries charging and discharging will always occur when non-renewable energy resources are on the margin. While this assumption holds true in regions with low penetration of renewable energy, it overlooks the impact of renewable energy resources in regions with high renewable penetration, potentially leading to incorrect MEF measurements. By examining the role of renewable energy sources, our study provides insight into how penetration levels of renewable energy sources will affect storage-induced carbon emissions based on MEFs.

Third, our paper differs from previous studies in considering both the operation and investment decisions of a solar plus storage system. In particular, our paper highlights the interactions between the economic values of solar and battery by looking at their investment decisions simultaneously. In contrast, most studies (e.g., Hittinger and Azevedo 2015; Fisher and Apt 2017) that look at storage-induced carbon emissions focus on battery operations in the short run, holding solar and battery capacities constant.

This paper is organized as follows. Section 1.2 provides a discussion on RTP and carbon pricing. Section 1.3 describes the economic mechanism of our analysis. Section 1.4 provides details of the data used in our study. Section 1.5 describes the methodology. Section 1.6 presents the primary results. Section 1.7 presents the results of two robustness checks.

^{16.} See section 1.2 for more details.

Section 1.8 concludes.

1.2 Real-Time Pricing and Carbon Pricing

Economists have long advocated using market-based mechanisms such as retail rate design and carbon pricing to address carbon emissions and enhance the overall efficiency of the electricity market (Bilich et al. 2019). In particular, it has been argued that retail electricity prices should reflect the time-varying costs of electricity, and real-time pricing (RTP) is the most efficient tariff structure to achieve this objective (Borenstein 2005). In the absence of RTP, consumers will consume more electricity during peak periods and less electricity during off-peak periods relative to the optimal quantity. In addition, absent RTP, more capacity is needed to meet higher peak demand than the first-best amount of capacity in the long run (Borenstein 2005). In addition, by increasing the flexibility of electricity demand, RTP can reduce market power (Holland and Mansur 2008).

A move to RTP can bring additional environmental benefits if the RTP provides incentives for batteries to shift energy from low-cost, low-emission hours to high-cost, high-emission hours through energy arbitrage. This could occur in regions with high renewable penetration because renewable energy sources such as solar have zero marginal costs and zero emissions (Antweiler and Muesgens 2021). For example, batteries can store cheap energy from solar when there is an excessive supply of solar energy and then discharge to replace dirtier and more expensive energy resources when the solar is offline.

However, more broadly, without carbon pricing, it is likely that a move to RTP will increase emissions in regions with lower renewable penetration and higher coal generation.¹⁷

^{17.} See Hittinger and Azevedo (2015) for a similar case with utility-scale batteries. In general, RTP in these regions will cause batteries to charge when coal-fired generation is on the margin because of lower electricity rates at night. Furthermore, RTP will often increase the frequency of charging, leading to higher energy losses.

Absent carbon pricing, customers face a set of distorted electricity prices that can lead to over-consumption of electricity generated from emitting sources. Although carbon pricing has been widely implemented among energy producers, carbon taxes are often passed through to consumers based on total electricity consumption instead of total emissions, leading to inefficient responses toward emissions (Fabra and Reguant 2014). It is important to include a mechanism that prices the social costs of carbon (such as a carbon tax) into the time-varying retail electricity prices (Weisbach and Metcalf 2009). Taken together, this emphasizes the value of both RTP and carbon pricing to ensure that the time-varying social marginal cost of electricity production is reflected in the electricity rates.

1.3 Economic Mechanism

This section describes how batteries can impact carbon emissions through energy arbitrage. Batteries can also impact emissions due to energy losses, but we are interested in energy arbitrage because it offers the potential for batteries to reduce carbon emissions. In a typical restructured wholesale electricity market, the wholesale electricity rate is determined by an auction system in which each power plant submits a bid indicating how many MWh they are willing to supply at various price levels. Then, subject to the transmission constraints, the least cost generators are called upon (i.e., dispatched) to produce electricity until supply equals demand. At each node in the transmission system, the market clearing price, also known as the locational marginal price (LMP), is determined by the marginal cost of supplying one more MWh of electricity to that node (Wolak 2021).

Figure 1 illustrates a hypothetical dispatch curve adapted based on the generation fleet of CAISO. Each bar presents a particular generation technology (e.g., solar, wind). The width of the bar measures the available generation capacity, and the height of the bar measures the marginal costs of the corresponding generation technology. The curve plots the average emissions intensity for each generation technology. In a perfectly competitive market, generators will bid at their marginal costs. The wholesale electricity price thus equals the marginal costs of the marginal generation technology that intersects with the demand curve.

Renewable energy resources such as solar and wind have near-zero marginal costs and are usually dispatched automatically. However, these intermittent energy sources only generate power when the sun shines or the wind blows. Wholesale electricity prices could be zero (or even negative) when solar or wind power is on the margin. However, when solar and wind generate less power and are no longer on the margin, the wholesale electricity price will be determined by generation technologies with higher marginal costs, such as natural gas. Differences in wholesale electricity prices create arbitrage opportunities for batteries.

As shown in Figure 1.1, with a high penetration of solar energy sources, batteries can charge during the middle of the day when the wholesale electricity price drops to zero. Then, as shown in Figure 1.2, the battery can discharge during evening ramps when the wholesale electricity price becomes higher, as open-cycle (i.e., simple-cycle) natural gas units are on the margin. As a result, the demand curve would shift from D1 to D2 in the middle of the day. As solar energy has zero carbon emissions, this increase in demand will not increase emissions. In the evening ramp, battery discharge can reduce the demand from D2 to D1. If batteries have a round-trip efficiency of 80%, then for every kWh charged into the battery, 0.8 kWh can be discharged. The discharge reduces the required electricity generation from open-cycle gas turbines and associated carbon emissions. The avoided carbon emissions are given by multiplying the discharged kWh with the emissions intensity (kg CO_2 per kWh) of the marginal technology (open-cycle).

This basic economic mechanism, where renewable energy is used to displace a dirtier source, can be applied to both CAISO and ERCOT. However, the magnitude of the emissions reductions may differ due to a number of reasons. First, the emissions reduction from energy arbitrage is determined by the marginal emissions during battery charging and discharging. Since solar and wind have near-zero emissions, if batteries are charged from these sources, the emissions reductions will be determined by the generation technologies used during battery discharging. For example, batteries can be used to displace coal-fired generators in ERCOT compared to natural gas generators in CAISO, leading to a greater reduction in emissions in ERCOT. Second, emission reductions can differ because different renewable energy sources have different probabilities of appearing on the margin. Energy arbitrage is most profitable when renewable energy sources are deployed on the margin, which occurs when renewable generation can meet local demand and is facing curtailments.¹⁸ Because ERCOT has higher renewable curtailment rates than CAISO¹⁹, a battery in ERCOT is anticipated to bring more emissions reductions than CAISO. Third, ERCOT and CAISO have different seasonal patterns in terms of electricity demand and renewable generation. As a result, emission reductions can also exhibit seasonal effects.

^{18.} This can occur in two scenarios. First, renewable generation may be sufficient to meet system demand, resulting in the curtailment of excess renewable generation and a MEF of zero throughout the system. Second, renewable generation may meet local demand, but due to transmission congestion, it cannot be transmitted to meet demand in other regions. In this case, the local MEF will be zero.

^{19.} According to ERCOT (2022) and CAISO (2023a), ERCOT curtailed 5% of wind generation and 10% of utility-scale solar generation in 2022, while CAISO's solar curtailment rate was around 5%.



Figure 1.1: Hypothetical Dispatch Curve During the Middle of the Day



Figure 1.2: Hypothetical Dispatch Curve During the Evening Ramps

In our context, RTP has the potential to reduce carbon emissions in CAISO and ERCOT because these regions have high renewable penetration. The potential to reduce carbon emissions is greater when solar and wind are more likely to be on the margin. Generation capacities and market demand jointly determine whether solar and wind will be on the margin. Therefore, RTP has a more considerable impact on emissions during times with higher wind and solar output and lower market demand.

In areas with fewer renewables, or in hours where renewable generation has a low probability of being on the margin, RTP could increase emissions. For example, batteries could be used to charge at night when cheap coal units are on the margin and discharged when cleaner natural gas is on the margin.²⁰ As a result, the increased emissions during charging outweigh the avoided emissions during discharging. Combined with energy loss from batteries, the emissions could increase substantially.

1.4 Data

We collect data from several sources on generation, emissions, retail tariffs, demand profiles, and solar radiation. First, we utilize hourly emissions and generation data from the Environmental Protection Agency (EPA)'s Continuous Emissions Monitoring System (CEMS) for the years 2018, 2019, and 2020. CEMS includes data on hourly emissions and generation for all fossil fuel generators in the US with a capacity larger than 25 MW (EPA 2023a). To account for electricity generation from non-emitting sources, we complement the dataset with hourly data for wind and solar generations from each grid operator (CAISO 2023b; ERCOT 2023a).²¹

Second, we use real-world retail tariffs from the Utility Rate Database (URDB) to design representative retail tariffs for each jurisdiction. The URDB provides rate structure information for more than 3,700 US utilities (Open EI 2023). In each jurisdiction, we use

^{20.} Hittinger and Azevedo (2015) shows that this is the case for most jurisdictions in the US.

^{21.} Our dataset does not include hydropower, biomass, geothermal, and small fossil-fuel generators due to data limitations. A detailed discussion of the data limitations is provided in Section 5.4.

URDB to collect possible retail tariffs from local utility companies for medium commercial customers with a maximum demand range of 250 kW to 500 kW (Open EI 2023). To ensure that the collected retail tariffs are representative, we focus on large utility companies and active utility plans in 2020. In CAISO, we focus on the three major investor-owned utilities (IOUs): Pacific Gas and Electric (PGE), San Diego Gas and Electric (SDGE), and Southern California Edison (SCE). In ERCOT, we look at the ten largest utility companies by market share that have available information for their commercial plans (ElectricityMatch 2023).²² In addition, following Borenstein and Bushnell (2022), we exclude any retail tariffs for specific customer types by removing any tariff with a name containing "agricultural," "rural," and "traffic." This leaves us with a more standard collection of retail tariffs for commercial customers.

The retail tariffs we collect are generally composed of three parts: a fixed charge, a demand charge, and a volumetric charge (typically TOU). Our analysis focuses on the TOU rate as it is adopted by most utilities for large commercial and industrial consumers and is a time-varying rate. For simplicity, we remove retail tariffs that contain tiered rates, as tiered pricing often lacks the time-varying feature of electricity pricing and does not justify the installation of batteries for energy arbitrage purposes.²³

Third, we use the Commercial and Residential Hourly Load Profiles for all TMY3 (Typical Meteorological Year 3) locations in the United States provided by the Open Energy Data Initiative (NREL 2014). This dataset contains simulated hourly electricity consumption data for 16 commercial building types across the US, representing approximately 70% of all

^{22.} The URDB only contains utilities that provide "bundled" services, which means the utilities that sell electricity also own and operate the distribution network (Borenstein and Bushnell 2022). However, in ERCOT, many retailers only provide the "energy" service. In this case, we manually collected retail tariff information for large utility companies that are omitted in the URDB: TXU Energy, Direct Energy, and Reliant Energy (ElectricityMatch 2023).

^{23.} A tiered rate is a price structure in which one can use a certain amount of electricity at a lower price. Once a certain threshold is reached, a higher rate will apply. It is usually used to promote energy conservation and support low-income consumers (Borenstein 2016).

commercial buildings in the US (Deru et al. 2011). For expositional purposes, our analysis uses supermarkets as the representative demand profile. The demand profiles in different locations in a state are aggregated to the state level by taking the average. Appendix C presents the representative demand profile of supermarkets.

In addition, we use EnerNOC's (2013) Green Button database, which provides 5-minute electricity consumption data for 100 commercial/industrial sites across the US, to quantify the reliability of the synthetic data from the TMY3 dataset (EnerNOC 2013). The correlation between synthetic data and real smart meter data is 0.8548 for supermarkets. The results (see Appendix C) suggest that the synthetic demand profiles can reasonably reflect real-world data.

Fourth, we use solar radiation data in large population centers from the TMY3 dataset to calculate the average energy output of solar panels (NREL 2014). The TMY3 dataset provides solar radiation information for 1,020 locations throughout the US. For simplicity, in each region, we use the average solar radiation in the three largest cities in California and Texas to calculate the energy output from solar. The assumption is reasonable because most commercial BTM storage systems are located in these large population centers (CPUC 2018a).

1.5 Methodology

We use the Distributed Energy Resources Customer Adoption Model (DER-CAM) to investigate the impact of RTP and carbon tax on the GHG emissions resulting from solar plus battery systems. DER-CAM is a decision support tool developed by Lawrence Berkeley National Laboratory (LBNL) to make optimal operation and investment decisions for DER technologies (LBNL, 2020). The optimization program can determine the optimal charge and discharge decisions that maximize customer bill savings with exogenous DER capacities,

subject to several financial and technological constraints. In addition, it can simultaneously solve the optimal investment and operation decisions for solar and storage. Section 1.5.1 illustrates how DER-CAM solves the optimization problem.

For a given demand profile and retail tariff design, we utilize DER-CAM to simulate the optimal operation and investment decisions for DER technologies and account for regional solar radiation, DER investment costs, and technological characteristics. Then, we use estimated MEFs to understand the environmental outcomes of RTP and carbon taxes compared to a more traditional TOU pricing structure. We first consider the case with exogenous solar and battery capacities. This allows us to isolate the impact of retail tariffs and carbon tax on carbon emissions resulting from charge and discharge decisions. Then, we simultaneously solve the operation and investment decisions to understand how retail tariff designs and carbon taxes would affect carbon emissions in the long run.

To understand how results could vary with generation mixes, our study covers CAISO and ERCOT, two of the leading jurisdictions in terms of renewable generation in the US (NREL 2020). In 2020, 37% of the energy consumed in CAISO came from natural gas, which generally served as the margin fuel. Solar and wind constitute 13% and 11% of energy, respectively (CAISO 2023a). In ERCOT, natural gas represented 46% of the energy generated in 2020, followed by wind (23%) and coal (18%) (Potomac Economics 2023b).

Renewable energy resources in CAISO and ERCOT have frequently been curtailed due to oversupply or local system congestion. According to EIA (2021a), in 2020, around 5% of total solar generation is curtailed in CAISO, within which 31.15% is system-wide curtailment due to oversupply. In ERCOT, 5% of all wind generation and 10% of solar generation were curtailed in 2020, mainly due to local congestion (Potomac Economics 2023b). As a reference, the Midcontinent Independent System Operator (MISO) has a 5% wind curtailment rate as compared to total generation and wind can appear on the margin for over 40% of the time (Li et al. 2017). The results suggest that wind and solar are frequently on the margin and can significantly affect system MEFs.

It is important to acknowledge that we assume that the demand profiles remain constant as we adjust the retail tariff. We carry out robustness checks to evaluate the implications of this assumption. The details of this analysis are described in Section 1.7.1.

1.5.1 Optimization Program

DER-CAM solves the optimal investment and operation decisions by minimizing total costs, subject to several technological constraints.²⁴ Total costs include electricity bills and technology costs (investment and operation costs) for solar and storage systems. The electricity bills contain fixed and variable charges, minus the energy credits earned from selling excess solar power back to the grid. A detailed discussion on how customers are compensated for exporting excess solar power is discussed in Section 1.5.5.

In the exogenous setting, the capacities for solar and battery storage are predetermined, whereas DER-CAM chooses the optimal charge and discharge decisions to maximize customer bill savings. In the endogenous setting, DER-CAM solves both the investment and operation decisions for the solar plus storage systems. The solar/storage capacity configuration is chosen to obtain a global optimization (minimize total costs) based on optimized charge and discharge decisions. This optimization problem is formulated in DER-CAM as a Mixed Integer Linear Program (Cardoso et al. 2017).

For simplicity, DER-CAM solves the problem based on hourly demand profiles represented by three day-types (weekdays, weekends, and high-demand days) for each month of a representative year. As a result, our primary analysis only captures the average trend of RTP and

^{24.} See Appendix B and Boampong and Brown (2020) for additional details on DER-CAM.

omits changes within hours or across days for a given month. It is important to note that this will reduce the variance of RTP, leading to underestimated arbitrage opportunities. Consequently, it could lead to underestimated battery installation and energy arbitrage incentives. As a robustness check, we design an algorithm based on the same objective function and constraints as the exogenous case to understand how our results could differ if we allow RTP to vary hourly and daily (see Section 1.7.2). Given the complexity of the endogenous case, our algorithm is designed exclusively for the exogenous scenario. To maintain consistency across both situations, we employ DER-CAM as our benchmark algorithm for both cases.

While solving the optimization problem, DER-CAM uses MEFs to calculate the total carbon emissions for each retail structure, which is calculated as follows:

$$Emission = \sum_{m} \sum_{d} \sum_{h} D_{mdh} M E F_{mh}$$
(1.1)

where m, d, and h represent month of the year, day type, and hour of the day, respectively. D_{mdh} , therefore, represents the electricity demand for a specific month m, day type d, and hour h. Section 1.5.4 provides a detailed discussion on how MEFs are estimated.

1.5.2 DER Technologies

This section outlines the basic features of solar and battery technologies in our analysis. Following Boampong and Brown (2020), we assume that the battery has a round-trip efficiency of 0.8 (i.e., a 20% of energy loss between charge and discharge), a charge and discharge rate of 0.3 (i.e., a maximum of 30% of total capacity can be charged and discharged per hour). In addition, we assume that the battery has a life span of 10 years and solar photovoltaic has a life span of 25 years, reflecting the most cost-effective battery technology (Mongird et al. 2019).

The costs of solar and storage have dropped significantly in the past ten years. Following Cardoso et al. (2017) and IRENA (2017), we assume that the battery has a capital cost of \$260 per kW, which reflect a base capital cost of \$560 per kW and a government subsidy of \$300 kW (from SGIP).²⁵ We follow Feldman et al. (2021) and assume that solar PV has a variable cost of \$1,750 per kW. In addition, we follow Mongird et al. (2019) and assume that battery has annual maintenance costs equal to 0.3% of the total capital expenditure, while solar PV has annual maintenance costs equal to 0.5% of the total capital expenditure. In addition, the discount rate is assumed to be 3%. Different assumptions on technology features, capital costs, and discount rates could slightly alter the investment decisions for solar and storage systems. However, the qualitative conclusions are likely to persist.

1.5.3 DER Capacity

In our exogenous setting, consumers choose the optimal charge and discharge decisions with predetermined solar and storage capacities. We must, therefore, assume an exogenous amount of solar plus storage capacities to conduct the analysis. Consumers may install solar PV or a solar plus storage system for various reasons, such as managing maximum demand²⁶, maximizing electricity bill savings, or using storage as an emergency backup during power outages (Boampong and Brown 2020).

Solar PV has been shown to achieve a high net present value under commonly employed compensation policies. In 2020, the payback period for solar PV systems was around six years in the US (SEIA 2022). As a result, for large electricity consumers, the size of solar PV is mainly constrained by the roof space. The average roof space size for a supermarket is approximately 4,000 square meters, which permits a maximum PV system of 599 kW (Kurdgelashvili et al. 2016). We assume that consumers will install the maximum PV capacity in the exogenous setting.²⁷

^{25.} While ERCOT does not have a state-wide subsidy for BTM storage, we assume the same government subsidy to CAISO and ERCOT to ensure parity.

^{26.} Customers can use a battery system to reduce maximum demand through load shifting and reduce demand charges.

^{27.} Boampong and Brown (2020) assume that customers will choose a PV capacity that can offset 50%

There is no simple way to determine the exogenous battery capacity, as different consumers may install batteries for different reasons. Following Boampong and Brown (2020), we determine the exogenous battery capacity by assuming that consumers will choose a storage capacity that can completely flatten the demand profiles in hours where utility purchase is necessary. We first calculate the solar generation in each hour based on our exogenous solar capacity. Then, we calculate the average net load \bar{L} and the net load in each hour with no solar export L_t . The battery capacity equals the sum of the positive deviations between L_t and \bar{L} . The exogenous capacities for solar and battery are shown in Table 1.

| | CAISO | ERCOT |
|----------------------|-------|-------|
| Solar PV (kW) | 599 | 599 |
| Battery Storage (kW) | 418 | 429 |

Table 1.1: Exogenous Capacity for Solar PV and Battery Storage

1.5.4 Marginal Emissions Factor

The marginal emissions factor (MEF) measures the carbon abatement potential from changes in electricity generation. In particular, MEFs measure the increase in emissions resulting from a 1 MW increase in market demand (or electricity generation) and can be used to evaluate energy efficiency policies and behavioral changes that shift energy load.

In the literature, MEFs are measured with various time frames. For example, Graff Zivin et al. (2014) estimate MEFs for each hour of a day for the entire sample period (three years). To capture potential seasonal effects, we estimate month-hour MEFs for each hour of the day and month of the year. Following Graff Zivin et al. (2014) and Callaway et al. (2018),

of their peak day usage. Their resulting exogenous capacity is similar to what we assumed. For example, following their assumption, supermarkets in California and Texas will install a solar PV of 531 kW and 601 kW, respectively. Our results are robust to different assumptions of PV capacity.

we estimate the following model for each region:

$$E_t = \sum_{m=1}^{12} \sum_{h=1}^{24} \beta_{hm} \cdot HOUR_h \cdot MONTH_m \cdot G_t + \sum_{m=1}^{12} \sum_{h=1}^{24} \alpha_{hm} \cdot HOUR_h \cdot MONTH_m + \varepsilon_t$$
(1.2)

where E_t and G_t measure hourly emissions and hourly electricity generation, respectively. As described in Section 4, the emissions data from the CEMS only contains carbon emissions from fossil fuel generators located within the jurisdiction territory with a generation capacity of 25 MW and above. G_t is the hourly generation from these fossil fuel generators, plus renewable generation from solar and wind sources located within the jurisdiction territory . $HOUR_h$ and $MONTH_m$ are dummy variables for hour h of the day and month m of the year, respectively. The fixed-effects term α_{hm} captures the average levels of emissions observed in hour h and month m. The month-hour MEFs are captured by β_{hm} . We use generation rather than total electricity demand because our emissions data do not entirely match aggregate demand. For example, by using the CEMS data, we ignore emissions associated with energy imports.

Due to data availability, our approach has several limitations. First, the hourly generation contains only thermal, solar and wind energy sources, which excludes other marginal fuel types such as small fossil fuel, biomass, and hydro. As presented in Siler-Evans et al. (2012), biomass and small fossil fuel plants only represent a small percentage of the overall generation fleet. For example, in Texas, biomass generation accounts for only 0.1% of the total generation, whereas small fossil fuel plants excluded by the CEMS take 1.4%. In CAISO, these excluded plants account for 2.8% of total generation. In addition, biomass power is usually used as baseload power (Matek and Gawell 2015). Therefore, these plants have minimal impact on the MEFs. As discussed in Siler-Evans et al. (2012) and Callaway et al. (2018), assessing hydropower is particularly challenging. For example, a battery discharge will reduce hydro generation when hydropower is on the margin. In this case, the MEF will be estimated as zero. However, a hydro reservoir could reserve the water for later use instead of

spilling the additional water. As a result, instead of reducing zero emissions at hour A, the battery discharge at hour A may reduce future emissions at hour B. Depending on how hydropower operates and when hydropower is on the margin, hydropower could have different effects on the MEFs.

Second, we assume that all consumption in a region is met by generation units from the same area, which rules out energy exchange across regions. For example, since CAISO imports energy from neighboring areas (with coal-fired power plants) to meet demand, the MEFs during peak demand hours could be underestimated.

Third, we assume that there are no line losses. Line losses account for approximately 9% of the total generation (Graff Zivin et al. 2014). By excluding line losses from the analysis, we could slightly underestimate the MEFs because we measure the increase in emissions resulting from a 1 kW generation instead of a 1 kW consumption. Because line losses can vary by hours and locations, accounting for line losses can be challenging.

Our methodology differs from previous studies (e.g., Siler-Evans et al. 2012; Graff Zivin et al. 2014; Callaway et al. 2018) in two ways. First, to understand the effects of RTP in greater time granularity, we estimate hour-month MEFs instead of annual or seasonal MEFs. Second, we estimate MEFs in an environment where renewable energy sources could be on the margin, whereas previous studies assume that renewable energy sources rarely appear on the margin. However, with the significant deployment of renewables and developments of the transmission system and dispatching mechanism, renewable energy sources have been deployed frequently on the margin (Li et al. 2017). For example, between 2014 and 2016, the wind was on the margin for more than 40% of the time in MISO (Li et al. 2017). In CAISO and ERCOT, renewable energy sources have been curtailed frequently, indicating their presence on the margin (CAISO 2023a; Potomac Economics 2023b). Omitting these energy sources will significantly overestimate the MEFs in hours during renewable generation.
For example, if only emitting generation sources are used to estimate the MEFs in CAISO, the MEFs are mostly constant across hours because only natural gas generators are used to meet marginal demand. We find that this can lead to an 60% overestimation of MEFs in the middle of the day in spring.

Our estimates of the MEF are shown in Figures 1.3 and 1.4. The MEFs shown in the figures are measured as the change in carbon emissions (kg) in response to a change in electricity generation (kWh). Figure 1.3 shows the estimated MEFs of CAISO in each month-hour of a representative year. In CAISO, peak electricity demand occurs in summer, and MEFs in the summer months are higher as peaking natural gas plants generally serve the margin. In spring and summer, renewable energy sources, especially solar, play an essential role in electricity generation in CAISO (CAISO 2023a). As a result, MEFs during solar generating hours are considerably lower in the spring months because system demand is relatively lower in spring. Therefore, solar has a greater chance of appearing on the margin in spring than in summer. The high-demand hours in CAISO occur from 4 PM to 9 PM, and the MEFs are generally higher during this period. Therefore, in the spring months, storage has the potential to reduce emissions by shifting clean energy to high-demand hours. However, in other months of the year, the relatively flat MEFs provide minimal opportunities for emissions reduction.



Figure 1.3: Month-Hour MEFs for CAISO (kg CO₂ per kWh)

Figure 1.4 shows the estimated MEFs of ERCOT in each month of a representative year. ERCOT reaches peak electricity demand during winter due to prevalent electric heating and relies on coal plants to meet peak demand, resulting in very high emissions during the winter months (White et al. 2021). During winter, MEFs are higher during the off-peak period than the peak period, as coal-fired generators are used for baseload generation, whereas cleaner natural gas generators are used to meet peak demand. In most of the other months, MEFs are affected by wind generation at night. In these months, MEFs showed higher variances than CAISO, which leaves more potential for emissions reduction through load shift.



Figure 1.4: Month-Hour MEFs for ERCOT (kg CO₂ per kWh)

1.5.5 Tariff Design and Solar Compensation Policy

Solar Compensation Policy

Most states have implemented a net-metering policy in the US to encourage solar PV adoption (Brown and Sappington 2017b). Under net metering, consumers can export excess solar energy to the utility in exchange for energy credits. These energy credits can be used to offset electricity purchases from the same TOU period when the export occurs, usually on a 1-to-1 basis (at the prevailing retail rate). Different utilities have different policies on how long these energy credits can be stored. For example, energy credits can be stored for 12 months in California (CPUC 2022). In contrast, energy credits earned by exporting solar energy in a month cannot be carried over to the next month in Texas (Quick Electricity 2022).²⁸ When the total solar export exceeds the total utility purchase, any surplus export will not be compensated or will be compensated at a lower rate.²⁹

To ensure comparability, we establish the same compensation rule for the TOU rate in ERCOT and CAISO. We assume that solar export is compensated for at the retail rate in the corresponding TOU period. In addition, the energy credits can be stored for the entire representative year. Lastly, following the FERC order 2222, we assume that net surplus export can be exported to the wholesale market and earn the wholesale rate at the corresponding hour (FERC 2021). Our analysis uses the retail net-metering policy as the benchmark case.

Retail Tariff Designs

The representative retail tariffs are designed as follows. We start by collecting the retail tariffs in a region and extracting the fixed and demand charges from each retail tariff. Then, we calculate the total electricity bill for our representative customer for each retail tariff we collected.³⁰

The TOU rates in our analysis are designed such that a utility can recover the average revenue made by the real-world retail tariffs from the representative customer, absent any DER investment. We create two representative TOU rates for each demand profile with peak-to-off-peak ratios equal to 2 and 5, respectively. For simplicity, the TOU rates in our analysis contain a peak and an off-peak period. To determine the peak period for each region

^{28.} Texas does not have a state-wide net-metering policy, but many utilities offer solar buyback plans that work the same as net-metering.

^{29.} Utilities in Texas do not offer compensation for this net surplus export. In California, this net surplus will be compensated for at the net surplus compensation rate, calculated based on the average energy market rate over the past 12 months. This rate is generally less than 4 cents per kWh in California. In general, any excess solar generation above consumer demand is not profitable.

^{30.} We first calculate the monthly kWh consumption for each TOU period and the monthly maximum demand based on the representative demand profile. Then, we calculate the monthly volumetric charges by multiplying the kWh consumption by the corresponding rates and monthly demand charges by multiplying the maximum demand by the demand charge rate. Lastly, we add the monthly fixed payment.

in our study, we use K-mean clustering³¹ to cluster demand into two groups using the system demand profile for CAISO and ERCOT during weekdays, respectively. The results indicate that the peak period in CAISO is between 4 PM and 9 PM (weekends excluded), whereas the peak period in ERCOT is between 1 PM and 6 PM (weekends excluded).³²

Given the peak and off-peak periods, the TOU rates can be calculated with the following equations

$$TotalBill = D_{peak} \cdot rate_{peak} + D_{off} \cdot rate_{off} + fixed \tag{1.3}$$

$$rate_{peak} = s \cdot rate_{off} \tag{1.4}$$

whereas D_{peak} and D_{off} are kWh consumption during peak and off-peak periods, respectively. $rate_{peak}$ and $rate_{off}$ are the energy rates during the peak and off-peak periods, respectively. s is the peak-to-off-peak ratio (2 or 5). *fixed* is the fixed charge, which is calculated taking the average of all fixed charges of the collected retail tariffs.

The RTP in our paper is designed as the average LMP for all trading hubs in the dayahead market in a region, plus a fixed charge. The LMP of a trading hub is the average of all LMPs in the corresponding trading zone. The fixed charge ensures that the utility company recovers the exact costs as the TOU rates, absent any DER investment. We calculate the fixed charge by subtracting the total hourly electricity charges from the total electricity bill. Based on our retail tariff designs, absent any DERs, the total electricity bills are identical across: (i) the observed real-world tariff, (ii) the TOU tariff, and (iii) the RTP tariff. This establishes an equitable starting point for comparing TOU and RTP.

^{31.} K-mean clustering is a common method for dividing data into smaller groups based on a similarity measure (Green et al. 2014). In our setting, the peak and off-peak periods are determined so that the distance between the hourly demand and the average demand of a period is minimized for each hour within that period.

^{32.} Our peak period hours are similar to the peak period hours chosen by the real-world utilities. For example, three major utility companies in California offer TOU rates with peak periods between 4 PM and 9 PM. The peak period of the ERCOT region generally spans from 1 PM to 7 PM, but may vary with the season.

It is important to note that our TOU rates are constructed using the prevailing retail prices that include demand charges. In the RTP structure, these charges are lumped into fixed charges. Although it will not affect the charge and discharge decisions for batteries in our exogenous case, it could affect the compensation rate of solar and investment decisions for DER technologies. In particular, the solar compensation rate under the TOU rates could be overestimated, leading to higher investment levels.

Carbon Taxes

In order to examine the effects of carbon taxes on the environmental outcome brought about by solar plus storage systems, we include carbon taxes on each retail tariff we described above. For the TOU rate, we calculate the carbon emissions generated from each period and multiply them by the corresponding carbon price to obtain the total carbon taxes for each period. Then, we calculate the per kWh carbon costs by dividing total carbon taxes by the kWh consumption for each TOU period. Lastly, we add the per kWh carbon costs to the existing peak and off-peak period energy rate to obtain the new TOU rates with carbon taxes.³³ For the RTP, we multiply the carbon price by the MEF for each hour in each month and add this per kWh carbon cost to the existing RTP rate.

We assume two levels of carbon tax: a normal level and a high level. The levels of carbon tax are determined by the social cost of CO_2 in 2020, estimated at IWG (2021). The normal level carbon tax is \$51 per metric ton of CO_2 , which is the average social cost of CO_2 . The high level of carbon tax is \$152 per metric ton of CO_2 , which is the estimate of the social cost of CO_2 in the 95th percentile.

^{33.} For example, suppose a TOU tariff has two periods, peak and off-peak. Without a carbon tax, the peak and off-peak rates are p_1 and p_2 ($p_1 > p_2$), respectively. Suppose that the carbon tax is c\$ per kg, the MEF during peak and off-peak periods are e_1 and e_2 , respectively. Because the MEF measures the increase in emissions resulting from the increase in electricity demand. The TOU plus carbon tax rates during the peak and off-peak periods will be $p_1 + e_1 * c$ and $p_2 + e_2 * c$, respectively. In this setting, as long as $p_1 + e_1 * c > p_2 + e_2 * c$, the batteries will charge during period 1 and discharge during period 2 to maximize arbitrage gains.

Our carbon tax designs assume that the current tariff designs do not feature a carbon pricing policy. As California has a carbon cap and trade system, the current tariff design, absent carbon taxes, already includes carbon pricing, since energy sellers tend to "pass through" most of the carbon costs to end-users (Fabra and Reguant 2014). As a result, in CAISO, the carbon tax level in our analysis could be higher than the actual social cost of carbon, and the effects of carbon taxes on emissions reductions could be overestimated. However, the current carbon pricing policies (e.g., cap and trade) tend to underestimate the cost of carbon significantly. For example, in 2020, the estimated social cost of carbon is around \$50, but the carbon allowance price in California is around \$17 (CARB 2022). Thus, our rate designs should reflect the social cost of carbon reasonably well.

Figure 1.5 shows the final retail tariffs (with different levels of carbon tax) analyzed in our paper. As an illustration, we only show the TOU2 rate (the TOU rate with a peak-tooff-peak ratio of 2). Compared with ERCOT, the average TOU rates in CAISO are much higher, reflecting the high energy costs in CAISO. The overall pattern of RTP with different levels of carbon taxes is similar in CAISO because of a high correlation between the MEFs and the RTP. As a result, the arbitrage opportunities with or without carbon taxes do not differ significantly. In contrast, carbon taxes could greatly impact the charge and discharge patterns in ERCOT due to higher variances of MEFs. In addition, the considerable divergence in the RTP indicates that the battery can be more profitable in ERCOT. Additional information on retail tariff designs is shown in Appendix D.



(a) CAISO



⁽b) ERCOT



Notes: RTP, RTP-CT, and RTP-CT-H stand for RTP with no, normal, and high carbon tax levels, respectively. TOU2 and TOU2-CT are TOU rates with no and normal carbon tax levels, respectively. For expositional purpose, we do not show TOU2-CT-H, TOU5, TOU5-CT, TOU5-CT-H on the graph.

1.5.6 Summary of Methodology

We consider two retail rate designs: RTP and TOU. We compare the environmental outcomes of RTP with the TOU rate to investigate the effects of retail tariffs on carbon emissions and how the results would vary with various settings of carbon tax. Under each tariff structure and carbon tax setting, we consider three solar and storage cases: a benchmark case where there is no DER technology, a solar-only case, and a solar plus storage case. The outcome differences between the RTP and TOU case capture the effects of different retail rate tariffs on DERs. The differences between the solar-only and solar-plus-storage cases capture the storage-induced carbon emissions. The different tariff structures and variations analyzed in our study are summarized in Table 1.2.

| Regions: | CAISO, ERCOT |
|--------------------------------------|--|
| Tariff Structure: | RTP, TOU |
| Carbon Tax Level ((CO_2) : | \$0, \$51, \$152 |
| Solar & Storage Installation: | No solar and storage, solar only, solar plus storage |

 Table 1.2:
 Summary of Variations

Throughout the analysis, we use supermarkets as the representative demand profile. However, our results are robust to alternative demand profiles. To further illustrate, we first consider the TOU rate. With pre-determined solar and battery capacities, demand profile will not affect the charge and discharge decisions for batteries. This is because, under net metering, customers are compensated at the prevailing retail rate, meaning that exporting energy to the grid has the same financial effect as using the energy on-site.³⁴ In this circum-

^{34.} Under net metering policy, customers can export energy at time t_1 and use the credit to offset energy use at t_2 , as long as both t_1 and t_2 belong to the same period. Essentially, solar generation at any point during a period can be considered to be generation for the entire period. This is true as long as solar generation during a period does not exceed the total consumption for that period.

stance, customers will charge batteries during off-peak periods and discharge them during peak periods, regardless of the demand profile. For the same reason, only the aggregate energy consumption in each TOU period (and the peak-to-off-peak demand ratio) will affect the investment decisions for solar and battery in the endogenous case. This is because once self-generation completely offsets self-energy use in a period, additional energy exports will not be compensated at the prevailing retail rate.³⁵ This constraint can be binding in the endogenous case, leading to different solar and battery investment decisions.³⁶

In terms of RTP, demand profiles do not influence battery charge and discharge decisions in the exogenous case. Similarly to the TOU case, this is because the solar compensation rate is the same as the retail rate under RTP. Batteries will charge when the energy rate is low and discharge when the energy rate is high, regardless of the demand profile. In the endogenous case, since the solar compensation rate always equals the RTP³⁷, even if the energy export exceeds the energy usage at the corresponding hour, the investment decisions for solar and batteries will also remain unaffected by the demand profile.

It should be noted that adding carbon taxes to the RTP could affect the charge and discharge decisions of the battery because the solar compensation rate (RTP) will differ from the avoided energy rate when self-use (RTP+CT). However, this will only occur with a sufficiently high carbon tax level that fundamentally alters the ordinal sequence of the energy rate. For example, if the lowest rate changes from 3 AM to 2 AM. Similarly, adding carbon

^{35.} Refer to Section 1.5.5 for the compensation policy regarding surplus solar exports.

^{36.} Consider two extreme conditions: Customer A, who only uses electricity during off-peak periods, and Customer B, who only uses electricity during peak periods. Suppose that solar power only generates electricity during off-peak periods. Even if solar panels and batteries are sufficiently cheap, Customer A will only install solar panels. The solar capacity is determined by the total energy usage of this customer. On the other hand, Customer B could install both solar and battery systems and use the battery to shift energy from off-peak periods to peak periods. When solar panels and batteries are sufficiently cheap, this customer will install capacities such that the energy shifted to the peak period just offsets his total energy usage in the peak period.

^{37.} Under RTP, solar energy is compensated at the RTP rate when there are no surplus exports. When energy exports exceed consumption, customers can sell the surplus solar energy to the grid at the wholesale electricity price, as stipulated by FERC Order 2222.

taxes to TOU tariffs may affect charge and discharge decisions only if it fundamentally changes the TOU structure such that the energy rate during the off-peak period is higher than the peak period, which is unlikely to occur.

Lastly, batteries will exhibit identical charge and discharge patterns under TOU in the exogenous case, regardless of their carbon tax levels. This is because adding a carbon tax to the TOU will only affect the peak and off-peak energy rates, and batteries will still charge during off-peak periods and discharge during peak periods. Therefore, only the base TOU case is presented in the results to avoid redundancy.

1.6 Results

1.6.1 Exogenous Results

This section presents the results with exogenous solar and storage capacities specified in Section 1.5.3. This allows us to focus on the charge and discharge decisions of the storage systems and isolate the environmental effects resulting directly from battery operations. As our main qualitative results are robust to demand profiles of different customer types, we use supermarkets as the representative demand profile.

Figure 1.6 presents the results by retail tariffs during peak days. A positive/negative kWh indicates that the battery is charging/discharging at the corresponding hour. The vertical lines show the TOU peak periods. Under TOU pricing, customers will follow the same charge and discharge pattern regardless of the peak-to-off-peak rate ratios to maximize bill savings. Therefore, only one TOU rate is presented in the figure. As shown in Figure 6, batteries will be charged to maximum capacity under TOU rates before entering the peak period, as this is the only arbitrage opportunity under the TOU pricing. The battery will start to release

power as soon as the peak period starts to minimize energy storage losses.³⁸

Under RTP, batteries will target the RTP rate ramps. For example, the batteries in both CAISO and ERCOT tend to discharge in the mornings and evenings when the RTP rate increases. When there is a carbon tax, batteries will adjust the charge and discharge pattern in part on the basis of the MEFs. This will reduce CO_2 emissions in hours where the RTP rate and the MEFs are less correlated.

Figure 1.6(a) presents the battery charge and discharge pattern for CAISO. Under the TOU rate, batteries are charged to maximum capacity at 3 PM before entering the peak period. The battery will start to release power as soon as the peak period starts to minimize energy storage losses. Under the RTP rate, batteries will target the two RTP rate ramps in the morning (7 AM) and evening (19 PM and 20 PM). When there is a carbon tax, customers are discouraged from charging batteries overnight because MEFs are relatively high (due to baseload natural gas units) despite low RTP rates since it is less profitable to charge overnight as compared to the no carbon tax setting. Higher carbon taxes further reduce the incentive of overnight charging, but a normal carbon tax is sufficient to capture most of the reduction.

Compared to CAISO, the majority of ERCOT's renewable generation comes from wind energy sources, which produce most of the energy overnight (1 AM to 4 AM). As a result, MEFs in ERCOT do not correlate with TOU peak hours or the RTP rate during the day. Figure 1.6(b) presents the battery charge and discharge decisions for supermarkets in ER-COT. The early TOU peak period overlaps with solar generation hours. During these hours, power generated from solar is used for both self-consumption and battery charging, and excessive power will be discharged during late peak hours. Similarly, the RTP peak overlaps

^{38.} The energy losses (or "self-discharge") identified here occur when energy stored in batteries is reduced due to internal chemical reactions. Unlike the energy losses represented by the round-trip efficiency (which occurs during charge and discharge), it represents an energy loss during storage. It is expressed as a percentage of the charge lost over a certain period (Divya and Østergaard 2009).

with solar generation under the RTP rate, leading to delayed battery discharge. In addition, lower RTP rates and lower emissions intensity from wind generation overnight encourage overnight battery charging and discharging during early mornings.

A major difference between CAISO and ERCOT is that solar produces more energy during the peak period in ERCOT. As a result, the residual demand (demand minus on-site solar generation) is lower in the ERCOT, leading to lower battery discharges at the evening ramp. This implies a potential substitution effect between solar PV and batteries in the ERCOT.



(a) CAISO



(b) ERCOT

Figure 1.6: Battery Charge and Discharge Decisions

The environmental outcomes for CAISO and ERCOT under different retail tariff designs are presented in Table 1.3. All percentage changes are relative to the solar-only case, which measures the environmental effects of batteries. We do not present the results for TOU-CT because the batteries will operate the same under TOU, regardless of the rate difference between peak and off-peak periods. Under the current (TOU) retail rate design, we find that adding on storage to an existing solar-only system tends to increase emissions, which is in line with previous studies (e.g., Hittinger and Azevedo 2015). As MEFs during charge and discharge are similar, the increases in emissions can be mainly attributed to the 80% roundtrip efficiency. Compared to the TOU rate, we find that RTP can reduce carbon emissions by approximately 1%. However, storage will still increase emissions under the RTP when there is no carbon tax. Carbon taxes are vital to reduce the emissions of batteries. In particular, the high variance of MEFs can bring about significant emission reduction through arbitrage. For example, carbon taxes can reduce 4.9% of carbon emissions in ERCOT, whereas the emissions reduction in CAISO is only 3.0%. We also find that carbon taxes have diminished effects on emissions reduction as long as the carbon tax rate is sufficiently high to motivate arbitrage between high MEF and low MEF hours. When the carbon tax is sufficiently high, the ability to reduce emissions through storage technologies is mainly limited by the MEF difference across hours but can be enhanced by increasing the round-trip efficiency of the battery.

Table 1.4 presents the decomposition of storage-induced carbon emissions. The storage-induced carbon emissions reflect changes in emissions under a solar plus storage system compared to a stand-alone solar system. Emissions from energy losses measure the increase in emissions due to a round-trip efficiency of less than 100%. It is calculated by multiplying the energy loss during battery charge and discharge by the corresponding MEFs.³⁹ Emissions from energy arbitrage measure the difference between increased emissions from battery

^{39.} DER-CAM assumes a battery with an RTE of 80% will face a 10% energy loss during charging and a 10% energy loss during discharging.

| | CAISO | | ERCOT | |
|---------------------------|-----------------|---------------|-----------------|---------------|
| | $\rm CO_2~(kg)$ | Change $(\%)$ | $\rm CO_2~(kg)$ | Change $(\%)$ |
| Base | 668,660 | _ | 915,662 | — |
| Solar Only | 208,960 | — | 283,296 | — |
| TOU - Solar + Storage | $215,\!154$ | 3.0 | 292,471 | 3.2 |
| RTP - Solar + Storage | $213,\!583$ | 2.2 | 288,872 | 2.0 |
| RTP CT - Solar + Storage | $208,\!165$ | -0.4 | 274,097 | -3.2 |
| RTP HCT - Solar + Storage | 202,730 | -3.0 | 269,326 | -4.9 |

 Table 1.3:
 Environmental Outcomes

Notes: (1) All percentage changes are relative to the solar-only case. (2) RTP, RTP CT, and RTP HCT stand for RTP with no, normal, and high carbon tax levels, respectively.

charge and avoided emissions from discharge. It is calculated by deducting the emissions from the energy loss from the total storage-induced carbon emissions.

As shown in Table 1.4, energy loss is the main cause of increased storage-induced emissions under TOU rates, as in Fisher and Apt (2017). The RTP can reduce emissions through energy arbitrage, but the RTP will also increase emissions from energy loss due to more frequent battery charges and discharges. Carbon taxes can substantially reduce carbon emissions from energy arbitrage and can often reduce emissions from energy loss as carbon taxes reduce the amount of charge and discharge in certain hours with high MEFs. Therefore, RTP plus carbon taxes can reduce storage-induced carbon emissions substantially, especially with a higher variance of MEFs.

It is worth noting that we assume a round-trip efficiency of 80%. If the round-trip efficiency reaches 90% (with more expensive battery technology), the emissions from energy losses could be reduced by 50%. In this case, RTP can reduce storage-induced carbon emissions more substantially.

| | CAISO | ERCOT |
|---------------------------|-----------------------------|-----------------------------|
| | Storage-induced CO_2 (kg) | Storage-induced CO_2 (kg) |
| TOU - Solar + Storage | 6,194 | 9,175 |
| energy arbitrage | -144 | 396 |
| energy loss | 6,338 | 8,779 |
| RTP - Solar + Storage | 4,623 | 5,576 |
| energy arbitrage | -8,089 | -7,932 |
| energy loss | 12,712 | 13,508 |
| RTP CT - Solar + Storage | -795 | -9,199 |
| energy arbitrage | -12,029 | -18,021 |
| energy loss | 11,234 | 8,822 |
| RTP HCT - Solar + Storage | -6,230 | -13,970 |
| energy arbitrage | -15,688 | -23,059 |
| energy loss | $9,\!458$ | 9,089 |

 Table 1.4:
 Decomposition of Storage-induced Carbon Emissions

Notes: RTP, RTP CT, and RTP HCT stand for RTP with no, normal, and high carbon tax levels, respectively.

There are substantial seasonal effects on environmental outcomes under different retail tariffs. Figure 1.7 presents the emissions reduction of the RTP as compared to a TOU pricing for CAISO and ERCOT, respectively. In CAISO, the RTP reduces the most carbon emissions during the spring months but will increase emissions from July to January.⁴⁰ In contrast, in ERCOT, the RTP will reduce emissions in the summer months and increase emissions from December to April. Whether RTP will increase or decrease carbon emissions is determined by whether renewables could serve as the marginal technology. Specifically, it depends on both the renewable output and the system demand. RTP could reduce emissions in months where both the RTP rates and MEFs are low due to the high renewable output and low system demand.



Figure 1.7: Carbon Reduction under the RTP by Month

The exogenous setting provides several key findings. First, with sufficient renewable penetration, RTP can reduce emissions compared to TOU rates through energy arbitrage. However, without carbon taxes, adding batteries to existing solar systems will still lead to increased carbon emissions under the RTP with a round-trip efficiency of 80%. This is because RTP increases the frequency of battery charge and discharge, increasing emissions as a result of energy losses. Increasing the round-trip efficiency of batteries can effectively reduce

^{40.} Although solar output is the highest in July and August in CAISO, they appear less on the margin due to higher demand. Therefore, the emissions reduction from energy arbitrage is lower in these months.

emissions under RTP. RTP plus carbon taxes can reduce carbon emissions in most cases, given sufficient arbitrage opportunities.

Second, whether batteries can reduce carbon emissions in a solar plus storage system is determined by both the round-trip efficiency of the system and the variance of the MEFs. If the variance of MEFs is low, avoided emissions from energy arbitrage (given a price signal that shifts load from high to low emission intensity hours) might be insufficient to cover the energy loss. Therefore, a mandate that requires emissions reductions for batteries could affect the profitability of the storage system in most of the regions in the US where natural gas plants serve as the marginal plants for most hours.

Third, charging batteries with solar energy does not necessarily lead to emission reductions if solar energy can be used to replace a dirtier generation source on the margin. Whether it will reduce carbon emissions will be determined by a number of factors, including the RTE of the battery and what resources are on the margin in the evening when the sun sets and the battery is discharged.

There are several reasons why RTP and carbon taxes do not substantially reduce carbon emissions substantially compared to TOU in CAISO. First, the RTP and MEFs in our analysis are averaged over an hourly and monthly basis. In a real-time setting, the correlation between MEFs and RTP and the variance of MEFs can be more substantial, leading to more considerable emissions reductions through load shifting.⁴¹ For example, on a very sunny day in May, the MEFs could be close to zero when solar panels generate electricity. Therefore, shifting the load from early evenings to solar generating hours will lead to much higher emissions reductions than solar export. Second, the MEFs are flat most of the months, reducing the potential for emissions reduction. Third, in the case of CAISO, the peak hours of TOU are highly correlated with high RTP periods, providing similar price signals.

^{41.} Section 1.7.2 provides a robustness check for "real-time" RTP. Our results hold under both averaged and "real-time" RTP. However, due to data limitations, a robustness check for "real-time" MEFs is not available.

In addition, the RTP rates and MEFs are highly correlated, which increases the effects of a stand-alone RTP rate but reduces the effects of a carbon tax on charge and discharge decisions.

1.6.2 Endogenous Results

This section presents the results with endogenous solar and storage capacities that minimize total electricity and DER investment costs. We first look at the endogenous results for RTP in CAISO. Table 1.5(a) shows the optimized solar and storage investment decisions in CAISO for RTP at zero, normal, and high carbon tax levels, respectively. Under net metering, solar is compensated at the prevailing retail rate, which is less profitable under the RTP rate than the TOU rate. As a result, customers are much less likely to install solar under the RTP rate, causing higher carbon emissions. Under our investment and technological assumptions, solar panels will be profitable with a compensation rate of no less than 6 cents per kWh. Thus, solar panels will be installed only in the high carbon tax setting with an average RTP rate of 10 cents per kWh. Once the compensation rate is above the cut-off value calculated in Appendix A, consumers will install the maximum allowed capacity for solar PV, binding only by the roof space. Under RTP, storage investment is not profitable regardless of carbon tax levels. This means that the retail rate differences across hours are insufficient to support a profitable arbitrage. This finding contributes to the existing body of evidence indicating that storage remains economically unviable, especially when its role in preventing outages is not considered (Muchlenbachs and Brown 2023).

It is worth noting that these results are reached under a "smoothed" RTP based on three representative day types. Therefore, the arbitrage opportunities are less than in a case with a "real-time" RTP that varies by hour and day. Section 1.7.2 provides a discussion on how this could affect endogenous results.

| Scenario | Carbon Tax (\$/tonne) | Solar Capacity (kW) | Storage Capacity (kW) | Costs (\$) | ${{f CO}_2} {{f Emissions}} {{f (kg)}}$ |
|-----------------|--------------------------|------------------------|--------------------------|---------------|---|
| (a) RTP | | | | | |
| Solar | 0 | 0 | 0 | \$192,212 | 668,660 |
| Solar + Storage | 0 | 0 | 0 | \$192,212 | 668,660 |
| Solar | \$51 | 0 | 0 | \$226,314 | 668,660 |
| Solar + Storage | \$51 | 0 | 0 | \$226,314 | 668,660 |
| Solar | \$152 | 599 | 0 | \$247,892 | 172,939 |
| Solar + Storage | \$152 | 599 | 0 | \$247,892 | 172,939 |
| (b) TOU2 | | | | | |
| Solar | 0 | 599 | 0 | \$121,871 | 150,177 |
| Solar + Storage | 0 | 599 | 0 | \$121,871 | $150,\!177$ |
| Solar | \$51 | 599 | 0 | \$131,093 | $150,\!177$ |
| Solar + Storage | \$51 | 599 | 0 | \$131,093 | $150,\!177$ |
| Solar | \$152 | 599 | 0 | \$149,356 | $150,\!177$ |
| Solar + Storage | \$152 | 599 | 0 | \$149,356 | $150,\!177$ |
| (c) TOU5 | | | | | |
| Solar | 0 | 599 | 0 | \$147,302 | 150,177 |
| Solar + Storage | 0 | 599 | 2101 | \$134,398 | 169,036 |
| Solar | \$51 | 599 | 0 | \$161,135 | $150,\!177$ |
| Solar + Storage | \$51 | 599 | 2115 | \$133,913 | 173,440 |
| Solar | \$152 | 599 | 0 | \$188,531 | $150,\!177$ |
| Solar + Storage | \$152 | 599 | 2120 | \$132,891 | 173,484 |

 Table 1.5:
 Endogenous Results for CAISO under Different Scenarios

Notes: (1) TOU2 and TOU5 are TOU pricing with peak-to-off-peak ratios of 2 and 5, respectively. (2) Costs are annual electricity costs that include both annualized investment costs for DER and annual electricity purchase costs. (3) CO₂ Emissions are annual CO₂ emissions based on customer consumption profile and DER operations.

Tables 1.5(b) and 1.5(c) show the optimized solar and storage investment decisions in

CAISO for TOU2 and TOU5, respectively. Under the TOU rate, customers have incentives to install solar and storage technologies to cover part or all of the electricity usage during the peak period. When solar generates most of its energy during the off-peak period, the investment decision for solar is mainly determined by the compensation rate during the off-peak period. In addition, whether batteries are installed could affect the size of the solar PV since battery charges increase the electricity demand during solar generation hours.⁴² As shown in Table 1.5(b), under the TOU2 rate, consumers are willing to install solar PV because the off-peak rate can sufficiently cover the capital costs of solar. The optimal solar capacity is bound by the rooftop space. Due to a low peak-to-off-peak ratio, installing battery storage is not profitable under any carbon tax levels.

Table 1.5(c) shows the endogenous results under the TOU5 rate. Even with a lower off-peak compensation rate, installing solar PV is still profitable. Unlike the TOU2 case, installing battery storage is profitable due to higher peak-to-off-peak rate differences. In particular, the battery capacity is selected so that all peak period consumptions are covered. As a result, carbon emissions increased by approximately 15% as compared to the no storage (TOU2) cases.⁴³

Table 1.6(a) shows the endogenous results for ERCOT under the RTP. Due to the lower compensation rate, consumers have incentives to install solar PV only if there is a carbon tax. Similarly to CAISO, storage is not profitable under all circumstances. These results suggest that relying on RTP to promote storage could be infeasible without government subsidies. The arbitrage opportunities under the RTP are not sufficient to cover its capital

^{42.} Under the net-metering policy, the size of the solar PV can be limited by either the rooftop space or the total electricity consumption during the off-peak period. The latter constraint exists because once total solar generation exceeds total consumption, the compensation rate will drop to a level where additional solar capacity is not profitable. When batteries exist, total electricity consumption during the off-peak period will increase due to battery charging, thus increasing the optimal size of the solar PV, provided that the rooftop space constraint is not binding.

^{43.} Without the rooftop space constraint, batteries do not necessarily increase emissions as they can induce larger solar PV capacities.

 costs .

| Scenario | Carbon Tax (\$/tonne) | Solar Capacity (kW) | Storage Capacity (kW) | Costs (\$) | ${{{f CO}_2}\atop{{f Emissions}}}$ |
|-----------------|--------------------------|------------------------|--------------------------|---------------|------------------------------------|
| (a) RTP | | | | | |
| Solar | 0 | 0 | 0 | \$114,780 | 915,662 |
| Solar + Storage | 0 | 0 | 0 | \$114,780 | 915,662 |
| Solar | \$51 | 599 | 0 | \$144,054 | 296,139 |
| Solar + Storage | \$51 | 599 | 0 | \$144,054 | 296,139 |
| Solar | \$152 | 599 | 0 | \$172,960 | 296,139 |
| Solar + Storage | \$152 | 599 | 0 | \$172,960 | 296,139 |
| (b) TOU2 | | | | | |
| Solar | 0 | 599 | 0 | \$93,429 | 286,190 |
| Solar + Storage | 0 | 599 | 0 | \$93,429 | 286,190 |
| Solar | \$51 | 599 | 0 | \$104,169 | 286,190 |
| Solar + Storage | \$51 | 599 | 0 | \$104,169 | 286,190 |
| Solar | \$152 | 599 | 0 | \$172,960 | 286,190 |
| Solar + Storage | \$152 | 599 | 0 | \$172,960 | 286,190 |
| (c) TOU5 | | | | | |
| Solar | 0 | 599 | 0 | \$77,166 | 286,190 |
| Solar + Storage | 0 | 599 | 0 | \$77,166 | 286,190 |
| Solar | \$51 | 599 | 0 | \$81,231 | 286,190 |
| Solar + Storage | \$51 | 599 | 0 | \$81,231 | 286,190 |
| Solar | \$152 | 599 | 0 | \$89,280 | 286,190 |
| Solar + Storage | \$152 | 599 | 2431 | \$58,695 | $375,\!985$ |

Table 1.6: Endogenous Results for ERCOT under Different Scenarios

Notes: (1) TOU2 and TOU5 are TOU pricing with peak-to-off-peak ratios of 2 and 5, respectively. (2) Costs are annual electricity costs that include both annualized investment costs for DER and annual electricity purchase costs. (3) CO₂ emissions are annual CO₂ emissions based on the customer consumption profile and DER operations.

Tables 1.6(b) and 1.6(c) present the optimized solar and storage investment decisions in ERCOT for TOU2 and TOU5, respectively. In ERCOT, solar can generate electricity during both off-peak and peak periods, leading to higher bill savings. Similarly to the CAISO case, installing solar PV is profitable under all retail tariff settings. In addition, as mentioned in Section 1.6.1, there exists a substitution effect between solar PV and batteries in ERCOT as solar generation can be used to offset peak period electricity purchase from the utility completely. As a result, installing batteries becomes unprofitable in most cases. An exception is the high carbon tax rate case under TOU5. This happens because even if solar exports are compensated at the wholesale rate, the profits from arbitrage can still cover the high capital costs of battery storage. As a result, emissions increased by about 30% compared to a non-battery case.

The endogenous case suggests that RTP alone cannot provide enough bill savings to justify the installation of solar and storage technologies at the prevailing costs. In this case, a high level of carbon taxes is needed to support the penetration of DER technologies under RTP. In addition, a high peak-to-off-peak ratio in TOU pricing does not necessarily lead to higher capacities of solar and storage and higher emissions reductions. Under our assumptions, the TOU2 rate generally performs better than RTP or TOU5 in terms of emission reduction.

1.7 Robustness Checks

1.7.1 Price Elasticity

In the default setting, we assume that electricity is perfectly inelastic. In reality, both the short- and long-term price elasticity of demand for electricity can differ from zero and have been extensively estimated in the literature (Andruszkiewicz et al. 2019). For example, (Burke and Abayasekara 2018) estimate the short-run price elasticities in the US from 2003

to 2015 and report an average elasticity of -0.11 for residential customers and -0.05 for commercial customers.⁴⁴

As a robustness check, we relax the perfect inelastic assumption and assume a price elasticity of -0.1. For each retail tariff in our study, we derive the new demand profile by calculating the elasticity-adjusted demand in each hour.⁴⁵ For simplicity, we assume that the initial demand profile is simulated at a flat rate, which is derived from Equation (1.2). We use the new demand profiles to examine endogenous and exogenous cases for a supermarket in CAISO.

Table 1.7 presents the adjusted environmental outcomes for the exogenous case. Compared to the benchmark case (Table 1.3), the TOU rates generate lower emissions as they lead to less electricity consumption during the peak period, which is highly correlated with high emissions intensity hours. In contrast, RTP rates lead to much higher emissions as consumers consume more electricity in both peak and off-peak periods due to lower volumetric charges in both periods.⁴⁶ As in the benchmark case, carbon taxes can further reduce emissions by aligning price signals with emissions. Carbon taxes will also raise the energy rate level and lead to lower carbon emissions because customers will consume less electricity. As shown in Table 1.7, elasticity has minimal effect on storage-induced carbon emissions. This is because a change in electricity demand will have a minimal impact on the charge and discharge decisions of the battery.

Table 1.8 shows the elasticity-adjusted endogenous results for CAISO under RTP. As shown in Table 1.8(a), with elasticity, customers under RTP tend to consume more energy,

^{44.} The short-run price elasticities for commercial and residential customers estimated by the EIA is -0.08 and -0.13, respectively (EIA 2021b).

^{45.} The elasticity-adjusted demand D_e equals $D + \Delta D$, where D is the original demand, and $\Delta D = -0.1(R/\Delta R)D$. We assume that the initial demand profile is simulated under a flat rate R, which can be calculated by equating $rate_{peak}$ and $rate_{off}$ with R in equation (2) from section 5.5.2. ΔR is the difference in electricity rate between R and the rate from our retail rate design.

^{46.} Borenstein and Bushnell (2022) suggest that the electricity prices in California are too high compared to the social marginal costs of electricity.

| Scenario | CO_2 Emissions (kg) | Change |
|---------------------------|-----------------------|--------|
| TOU2 - Solar Only | 208251 | |
| TOU2 - Solar + Storage | 214531 | 3.0% |
| TOU5 - Solar Only | 202123 | |
| TOU5 - Solar + Storage | 207887 | 2.9% |
| RTP - Solar Only | 254294 | |
| RTP - Solar + Storage | 259115 | 1.9% |
| RTP CT - Solar Only | 241324 | |
| RTP CT - Solar + Storage | 243608 | 0.9% |
| RTP HCT - Solar Only | 215638 | |
| RTP HCT - Solar + Storage | 217573 | 0.9% |

Table 1.7: Elasticity-Adjusted Environmental Outcomes for CAISO

Notes: (1) All changes are relative to the solar-only case of the same retail rate design. (2) RTP, RTP CT, and RTP HCT stand for RTP with no, normal, and high carbon tax levels, respectively.

resulting in higher emissions. As a result, emissions will increase by 6.7%, 4.8%, and 0.5% under the zero, normal, and high carbon taxes, respectively. This suggests that a very high carbon tax is needed to reduce the energy consumption increase brought about by the lower RTP rates.⁴⁷

^{47.} It is worth noting that the current retail rate in CAISO is higher than marginal costs, rendering it inefficient. Lower RTP rates could potentially increase emissions, but could generally improve the overall efficiency of electricity consumption (Borenstein and Bushnell 2022).

| Scenario | Carbon Tax (\$/tonne) | Solar Capacity (kW) | Storage Capacity (kW) | Costs (\$) | ${{{\bf CO}_2}\atop{{{f Emissions}}}}_{({ m kg})}$ |
|-----------------|--------------------------|------------------------|--------------------------|---------------|--|
| (a) RTP | | | | | |
| Solar | 0 | 0 | 0 | \$195,616 | 713,994 |
| Solar + Storage | 0 | 0 | 0 | \$195,616 | 713,994 |
| Solar | \$51 | 0 | 0 | \$230,270 | 701,024 |
| Solar + Storage | \$51 | 0 | 0 | \$230,270 | 701,024 |
| Solar | \$152 | 599 | 0 | \$269,613 | 381,140 |
| Solar + Storage | \$152 | 599 | 0 | \$269,613 | 381,140 |
| (b) TOU2 | | | | | |
| Solar | 0 | 599 | 0 | \$150,901 | 145,799 |
| Solar + Storage | 0 | 599 | 0 | \$150,901 | 145,799 |
| Solar | \$51 | 599 | 0 | \$166,619 | 145,799 |
| Solar + Storage | \$51 | 599 | 0 | \$166,619 | 145,799 |
| Solar | \$152 | 599 | 0 | \$193,824 | 145,799 |
| Solar + Storage | \$152 | 599 | 0 | \$193,824 | 145,799 |
| (c) TOU5 | | | | | |
| Solar | 0 | 599 | 0 | \$151,310 | 145,799 |
| Solar + Storage | 0 | 599 | 1139 | \$134,714 | 159,230 |
| Solar | \$51 | 599 | 0 | \$164,063 | 145,799 |
| Solar + Storage | \$51 | 599 | 1098 | \$137,641 | 166,198 |
| Solar | \$152 | 599 | 0 | \$184,160 | 292,604 |
| Solar + Storage | \$152 | 599 | 1065 | \$141,897 | 164,458 |

Table 1.8: Elasticity-Adjusted Endogenous Results for CAISO

Notes: (1) TOU2 and TOU5 are TOU pricing with peak-to-off-peak ratios of 2 and 5, respectively. (2) Costs are annual electricity costs that include both annualized investment costs for DER and annual electricity purchase costs. (3) CO₂ emissions are annual CO₂ emissions based on elasticity-adjusted customer consumption profiles and DER operations.

Tables 1.8(b) and 1.8(c) show similar results. Compared to the benchmark case, elasticity

can affect environmental outcomes through two channels. First, when customers react to the energy rates, they consume less during the peak period. Therefore, emissions will decrease when the peak period coincides with high emission-intensity hours. Second, since storage is used to target peak periods under the TOU, consumers are less motivated to install storage when they consume less during the peak period. Because storage tends to increase emissions under the TOU rates, emissions under the TOU rates will be reduced with a more negative elasticity.

Overall, when consumers react to energy rates, they will consume more energy under RTP, leading to higher emissions. Under TOU, consumers will consume less energy during the peak period and will install batteries with smaller sizes, leading to lower emission levels.

Our calculations are done without the influence of solar or battery technologies on consumer behaviors. Consumers may change or shift their electricity demand in response to these technologies. For example, a consumer may consume more energy during solar generation hours to avoid net surplus exports or consume more energy in general since energy becomes cheaper, known as the rebound effect (Beppler et al. 2023). Our analysis does not consider these effects, but considering the relatively low elasticity of demand for commercial customers, we anticipate that the results will be similar to our analysis in Section 1.7.1.

1.7.2 "Real-time" RTP

As DER-CAM only allows representative RTP to be averaged over a month-hour basis, our primary analysis tends to underestimate the arbitrage opportunities (e.g. from price spikes) brought about by the RTP. This section uses a "real-time" RTP that varies hourly to examine how our baseline results might deviate from the real-world outcomes. As in the benchmark case, we assume that the battery is operating under perfect foresight: the customer has perfect information about the demand and RTP in the future. Unlike the benchmark case where the RTP is averaged over three representative days (week, weekend, peak), this section uses the "real-time" RTP, holding other factors constant.

To carry out the analysis, we develop an algorithm following the same objective function and constraints described in Section 1.5.1. In particular, we develop a mixed-integer linear programming model to simulate the charge and discharge decisions of a battery using consecutive hourly inputs. While our algorithm performs the same as DER-CAM for exogenous cases, we rely on DER-CAM to study the endogenous cases. We use LMPs in April 2019 in CAISO to construct two sets of RTP: a representative RTP averaged hourly over the three representative days and a "real-time" hourly RTP that varies by hour and day. We use April 2019 in CAISO because April has a significant variance of LMPs, and the arbitrage opportunities could be significantly underestimated in the benchmark case. The summary statistics for the two sets of RTP are shown in Table 1.9. The standard deviation of the real-time RTP is slightly higher than in the benchmark case, but not by a significant margin.

| Variable | Mean | Std. Dev. | Min | Max |
|-------------------|-------|-----------|--------|-------|
| RTP - benchmark | 0.037 | 0.017 | 0.011 | 0.124 |
| RTP - "real-time" | 0.037 | 0.021 | -0.004 | 0.124 |

 Table 1.9:
 Summary Statistics of the RTP

The environmental results are shown in Table 1.10. Surprisingly, the difference between the benchmark and the "real-time" cases is minimal. This reflects that the RTP averaged over three representative days can reasonably represent "real-time" pricing well. This could be due to two reasons. First, in the day-ahead market, the variance of LMPs is slight, and the LMP patterns are relatively robust across days. In other words, a price spike is very unlikely to happen in the day-ahead market. Second, arbitrage opportunities during days with subnormal levels of LMPs are well captured by the "peak" representative day.

| | Total Carbon (kg) | Total Bill (\$) |
|----------------------|-------------------|-----------------|
| Solar-only | 17,173 | 2,453 |
| RTP - benchmark | 17,098 | 2,070 |
| RTP - "real-time" | 17,053 | 2,043 |
| RTP CT - benchmark | 16,280 | $2,\!905$ |
| RTP CT - "real-time" | 16,233 | 2,940 |

 Table 1.10:
 Environmental Outcomes Comparison

1.8 Conclusion

Our paper investigates whether RTP and a carbon tax could be used to cut storage-induced carbon emissions compared to the more traditional TOU rates. We find that with better timing of charging and discharging, RTP can reduce carbon emissions compared to a TOU rate. However, emission reductions are often insignificant when TOU rates are carefully designed with batteries that have an 80% RTE. In general, RTP alone cannot eliminate the emissions resulting from the use of batteries. This is because the increase in emissions from energy loss largely offsets the reduction in emissions from energy arbitrage. However, when the battery RTE increases to 90%, RTP has the potential to eliminate storage-induced emissions through energy arbitrage even without a carbon tax. When the battery RTE is at 80%, a carbon tax is essential for the elimination of storage-induced carbon emissions.

The effectiveness of RTP plus carbon taxes in reducing emissions through energy arbitrage is determined by the variance of the MEFs. In most regions where renewable penetration is low and natural gas plants generally serve in the margin, RTP is likely to not incur emission reductions. RTP and carbon tax can reduce the most substantial emissions when the region has both renewable energy resources and coal-fired power plants. In this circumstance, removing the net-metering compensation policy would only slightly increase carbon emissions.

Taking investment decisions into consideration, we find that RTP is generally insufficient to promote the adoption of solar plus storage technologies. Only with a very high carbon tax level will consumers be willing to install solar and batteries under the RTP. As a result, TOU rates often provide better environmental results than RTP. Under RTP, solar and battery costs must be reduced by 20% and 50%, respectively, to incentivize consumers to adopt them.

These findings provide important policy insights for retail rate reforms. Our analysis highlights the importance of using RTP and carbon taxes to provide real-time price signals to reduce carbon emissions from batteries. In addition, we show that government subsidies will be required to promote the adoption of DER technologies if RTP is implemented. This implies that it is not cost effective to subsidize batteries at the current cost level from an environmental point of view, considering the relatively low social gains from the reduction of emissions.⁴⁸

There are three main limitations to our analysis that require future research. First, the RTP and MEFs used in our analysis are averaged on an hourly basis on three typical days. As a result, the effects of RTP on the charge and discharge decisions could be underestimated.⁴⁹

Second, we only employ two simple retail rate structures, RTP and TOU rates. Other commonly used rate structures (such as demand charges and tiered charges) are left for future studies.

^{48.} It is worth noting that battery storage systems can provide indirect environmental benefits by enhancing grid reliability, particularly in managing the intermittency of renewable energy sources. However, this is beyond the scope of our study.

^{49.} The robustness check from section 1.7.2 is conducted to deal with the issue, but the MEFs are still not real-time.

Third, our paper is static and only considers the current level of renewable penetration. As more adoptions of solar and storage occur, RTP could be lower, and MEFs tend to be flatter. As a result, the incentives for adopting solar plus storage systems could drop significantly.

Fourth, our paper focuses exclusively on investment decisions driven by energy arbitrage opportunities arising from time-varying retail tariff designs. In reality, customers may choose to install batteries for various reasons. In particular, customers might install batteries to avoid maximum demand charges and prevent outages (Boampong and Brown 2020; Muehlenbachs and Brown 2023). These different incentives should be explored further in future research.

References

- Andruszkiewicz, Jerzy, Józef Lorenc, and Agnieszka Weychan. 2019. "Demand Price Elasticity of Residential Electricity Consumers with Zonal Tariff Settlement Based on Their Load Profiles." *Energies* 12, no. 22 (November 13, 2019): 4317. ISSN: 1996-1073, accessed June 2, 2024. https://doi.org/10.3390/en12224317. https://www.mdpi.com/ 1996-1073/12/22/4317.
- Antweiler, Werner, and Felix Muesgens. 2021. "On the long-term merit order effect of renewable energies." *Energy Economics* 99 (July): 105275. ISSN: 01409883, accessed June 2, 2024. https://doi.org/10.1016/j.eneco.2021.105275. https://linkinghub.elsevier. com/retrieve/pii/S0140988321001808.
- Arciniegas, Laura M., and Eric Hittinger. 2018. "Tradeoffs between revenue and emissions in energy storage operation." *Energy* 143 (January): 1–11. ISSN: 03605442, accessed June 2, 2024. https://doi.org/10.1016/j.energy.2017.10.123. https://linkinghub.elsevier.com/retrieve/pii/S0360544217318145.
- Astoria, Ross. 2017. "On the radicality of New York's Reforming the Energy Vision." The Electricity Journal 30, no. 5 (June): 54–58. ISSN: 10406190, accessed June 2, 2024. https://doi.org/10.1016/j.tej.2017.04.018. https://linkinghub.elsevier.com/retrieve/ pii/S1040619017301070.
- Beppler, Ross C., Daniel C. Matisoff, and Matthew E. Oliver. 2023. "Electricity consumption changes following solar adoption: Testing for a solar rebound." *Economic Inquiry* 61, no. 1 (January): 58–81. ISSN: 0095-2583, 1465-7295, accessed June 2, 2024. https://doi.org/10.1111/ecin.13031. https://onlinelibrary.wiley.com/doi/10.1111/ecin. 13031.
- Berg, Sandford, and Andreas Savvides. 1983. "The theory of maximum kW demand charges for electricity." *Energy Economics* 5, no. 4 (October): 258–266. ISSN: 01409883, accessed September 15, 2023. https://doi.org/10.1016/0140-9883(83)90030-0. https://linkinghub.elsevier.com/retrieve/pii/0140988383900300.
- Bilich, Andy, Elisheba Spiller, and James Fine. 2019. "Proactively planning and operating energy storage for decarbonization: Recommendations for policymakers." *Energy Policy* 132 (September): 876–880. ISSN: 03014215, accessed June 2, 2024. https://doi.org/10.1016/j.enpol.2019.06.033. https://linkinghub.elsevier.com/retrieve/pii/S0301421519304033.
- Boampong, Richard, and David P. Brown. 2020. "On the benefits of behind-the-meter rooftop solar and energy storage: The importance of retail rate design." *Energy Economics* 86 (February): 104682. ISSN: 01409883, accessed October 17, 2023. https://doi.org/10.1016/j.eneco.2020.104682. https://linkinghub.elsevier.com/retrieve/pii/S0140988320300219.

- Borenstein, Severin. 2005. "The Long-Run Efficiency of Real-Time Electricity Pricing." The Energy Journal 26 (3): 93–116.
- Borenstein, Severin. 2016. "The economics of fixed cost recovery by utilities." The Electricity Journal 29, no. 7 (September): 5–12. ISSN: 10406190, accessed September 15, 2023. https://doi.org/10.1016/j.tej.2016.07.013. https://linkinghub.elsevier.com/ retrieve/pii/S1040619016301130.
- Borenstein, Severin. 2017. "Private Net Benefits of Residential Solar PV: The Role of Electricity Tariffs, Tax Incentives, and Rebates." Journal of the Association of Environmental and Resource Economists 4 (S1): S85–S122. ISSN: 2333-5955, 2333-5963, accessed June 2, 2024. https://doi.org/10.1086/691978. https://www.journals.uchicago.edu/ doi/10.1086/691978.
- Borenstein, Severin, and James B. Bushnell. 2022. "Do Two Electricity Pricing Wrongs Make a Right? Cost Recovery, Externalities, and Efficiency." *American Economic Journal: Economic Policy* 14, no. 4 (November 1, 2022): 80–110. ISSN: 1945-7731, 1945-774X, accessed October 30, 2023. https://doi.org/10.1257/pol.20190758. https: //pubs.aeaweb.org/doi/10.1257/pol.20190758.
- Brown, David P., and David E.M. Sappington. 2017b. "Designing Compensation for Distributed Solar Generation: Is Net Metering Ever Optimal?" *The Energy Journal* 38, no. 3 (May): 1–32. ISSN: 0195-6574, 1944-9089, accessed June 2, 2024. https://doi.org/ 10.5547/01956574.38.3.dbro. https://journals.sagepub.com/doi/10.5547/01956574.38. 3.dbro.
- Burke, Paul J., and Ashani Abayasekara. 2018. "The Price Elasticity of Electricity Demand in the United States: A Three-Dimensional Analysis." *The Energy Journal* 39, no. 2 (March): 123–146. ISSN: 0195-6574, 1944-9089, accessed June 2, 2024. https: //doi.org/10.5547/01956574.39.2.pbur. https://journals.sagepub.com/doi/10.5547/ 01956574.39.2.pbur.
- CAISO. 2023a. 2022 Annual Report On Market Issues and Performance.
- CAISO. 2023b. Wind and Solar Summary. Accessed May 5, 2023. http://oasis.caiso.com/ mrioasis/logon.do.
- Callaway, Duncan S., Meredith Fowlie, and Gavin McCormick. 2018. "Location, Location, Location: The Variable Value of Renewable Energy and Demand-Side Efficiency Resources." Journal of the Association of Environmental and Resource Economists 5, no. 1 (January): 39–75. ISSN: 2333-5955, 2333-5963, accessed June 2, 2024. https://doi.org/10.1086/694179. https://www.journals.uchicago.edu/doi/10.1086/694179.
- **CARB.** 2022. Cap-and-Trade Program Data. Accessed May 31, 2023. https://ww2.arb.ca. gov/our-work/programs/cap-and-trade-program/cap-and-trade-program-data.

- Cardoso, G., M. Stadler, S. Mashayekh, and E. Hartvigsson. 2017. "The impact of ancillary services in optimal DER investment decisions." *Energy* 130 (July): 99–112. ISSN: 03605442, accessed June 2, 2024. https://doi.org/10.1016/j.energy.2017.04.124. https://linkinghub.elsevier.com/retrieve/pii/S0360544217306904.
- CPUC. 2017. 2016 SGIP Advanced Energy Storage Impact Evaluation.
- **CPUC.** 2018a. 2017 SGIP Advanced Energy Storage Impact Evaluation.
- CPUC. 2018b. Revised Self-Generation Incentive Program Greenhouse Gas Staff Proposal.
- **CPUC.** 2022. "Customer-Sited Renewable Energy Generation." Accessed April 30, 2022. https://www.cpuc.ca.gov/industries-and-topics/electrical-energy/demand-side-management/net-energy-metering.
- **CPUC.** 2023. Self-Generation Incentive Program Handbook.
- Deru, M., K. Field, D. Studer, K. Benne, B. Griffith, P. Torcellini, B. Liu, et al. 2011. U.S. Department of Energy Commercial Reference Building Models of the National Building Stock. NREL/TP-5500-46861, 1009264. February 1, 2011. Accessed June 6, 2024. https://doi.org/10.2172/1009264. http://www.osti.gov/servlets/purl/1009264pitlfN/.
- Divya, K.C., and Jacob Østergaard. 2009. "Battery energy storage technology for power systems—An overview." *Electric Power Systems Research* 79, no. 4 (April): 511–520. ISSN: 03787796, accessed June 2, 2024. https://doi.org/10.1016/j.epsr.2008.09.017. https://linkinghub.elsevier.com/retrieve/pii/S0378779608002642.
- EIA. 2021a. "California's curtailments of solar electricity generation continue to increase." Accessed April 30, 2022. https://www.eia.gov/todayinenergy/detail.php?id=49276.
- EIA. 2021b. Price Elasticity for Energy Use in Buildings in the United States.
- **ElectricityMatch.** 2023. "Texas Electricity Providers." Accessed April 30, 2023. https://electricitymatch.com/providers/.
- EnerNOC. 2013. ENERNOC 2012 Commercial Energy Consumption Data. Accessed April 30, 2023. https://trynthink.github.io/buildingsdatasets/show.html?title_id=enernoc-2012-commercial-energy-consumption-data.
- **EPA.** 2023a. Continuous Emissions Monitoring System. Accessed May 5, 2023. https://www.epa.gov/emc/emc-continuous-emission-monitoring-systems.
- **EPA.** 2023b. Tailpipe Greenhouse Gas Emissions from a Typical Passenger Vehicle.
- ERCOT. 2022. ERCOT Fuel Mix Report. Accessed May 5, 2024. https://www.ercot.com/ gridinfo/generation.
- ERCOT. 2023a. Fuel Mix Report. Accessed May 5, 2023. https://www.ercot.com/gridinfo/generation.

- Fabra, Natalia, and Mar Reguant. 2014. "Pass-Through of Emissions Costs in Electricity Markets." American Economic Review 104, no. 9 (September 1, 2014): 2872–2899. ISSN: 0002-8282, accessed June 2, 2024. https://doi.org/10.1257/aer.104.9.2872. https: //pubs.aeaweb.org/doi/10.1257/aer.104.9.2872.
- Faruqui, Ahmad, and Stephen George. 2005. "Quantifying Customer Response to Dynamic Pricing." *The Electricity Journal* 18, no. 4 (May): 53–63. ISSN: 10406190, accessed September 15, 2023. https://doi.org/10.1016/j.tej.2005.04.005. https://linkinghub. elsevier.com/retrieve/pii/S1040619005000412.
- Felder, Frank A., and Rasika Athawale. 2014. "The Life and Death of the Utility Death Spiral." The Electricity Journal 27, no. 6 (July): 9–16. ISSN: 10406190, accessed June 5, 2024. https://doi.org/10.1016/j.tej.2014.06.008. https://linkinghub.elsevier.com/ retrieve/pii/S1040619014001407.
- Feldman, David, Vignesh Ramasamy, Ran Fu, Ashwin Ramdas, Jal Desai, and Robert Margolis. 2021. U.S. Solar Photovoltaic System and Energy Storage Cost Benchmark (Q1 2020). NREL/TP-6A20-77324, 1764908, MainId:26270. January 1, 2021. Accessed June 6, 2024. https://doi.org/10.2172/1764908. https://www.osti. gov/servlets/purl/1764908/.
- **FERC.** 2021. "FERC Order No. 2222 Explainer: Facilitating Participation in Electricity Markets by Distributed Energy Resources." Accessed April 30, 2022. https://www.ferc.gov/ferc-order-no-2222-explainer-facilitating-participation-electricity-markets-distributed-energy.
- Fisher, Michael J., and Jay Apt. 2017. "Emissions and Economics of Behind-the-Meter Electricity Storage." *Environmental Science & Technology* 51, no. 3 (February 7, 2017): 1094–1101. ISSN: 0013-936X, 1520-5851, accessed June 5, 2024. https://doi.org/10. 1021/acs.est.6b03536. https://pubs.acs.org/doi/10.1021/acs.est.6b03536.
- Graff Zivin, Joshua S., Matthew J. Kotchen, and Erin T. Mansur. 2014. "Spatial and temporal heterogeneity of marginal emissions: Implications for electric cars and other electricity-shifting policies." Journal of Economic Behavior & Organization 107 (November): 248–268. ISSN: 01672681, accessed June 5, 2024. https://doi.org/10.1016/ j.jebo.2014.03.010. https://linkinghub.elsevier.com/retrieve/pii/S0167268114000808.
- Green, Richard, Iain Staffell, and Nicholas Vasilakos. 2014. "Divide and conquer? Assessing k-means clustering of demand data in simulations of the British electricity system." *IEEE Transactions on Engineering Management* 61 (2): 251–260.
- Hittinger, Eric S., and Inês M. L. Azevedo. 2015. "Bulk Energy Storage Increases United States Electricity System Emissions." *Environmental Science & Technology* 49, no. 5 (March 3, 2015): 3203–3210. ISSN: 0013-936X, 1520-5851, accessed June 5, 2024. https://doi.org/10.1021/es505027p. https://pubs.acs.org/doi/10.1021/es505027p.
- Holland, Stephen P, and Erin T Mansur. 2008. "Is Real-Time Pricing Green? The Environmental Impacts of Electricity Demand Variance." The Review of Economics and Statistics.
- IPCC. 2022. Global Warming of 1.5°C: IPCC Special Report on Impacts of Global Warming of 1.5°C above Pre-industrial Levels in Context of Strengthening Response to Climate Change, Sustainable Development, and Efforts to Eradicate Poverty. 1st ed. Cambridge University Press, June 9, 2022. ISBN: 978-1-00-915794-0 978-1-00-915795-7, accessed June 5, 2024. https://doi.org/10.1017/9781009157940. https://www.cambridge.org/ core/product/identifier/9781009157940/type/book.
- **IRENA.** 2017. Electricity storage and renewables: costs and markets to 2030. OCLC: 1393122335. Abu Dhabi: International Renewable Energy Agency.
- **IWG.** 2021. Technical Support Document: Social Cost of Carbon, Methane, Washington, D.C. (United States): The Interagency Working Group.
- Kurdgelashvili, Lado, Junli Li, Cheng-Hao Shih, and Benjamin Attia. 2016. "Estimating technical potential for rooftop photovoltaics in California, Arizona and New Jersey." *Renewable Energy* 95 (September): 286–302. ISSN: 09601481, accessed June 5, 2024. https://doi.org/10.1016/j.renene.2016.03.105. https://linkinghub.elsevier.com/ retrieve/pii/S0960148116302890.
- Li, Mo, Timothy M. Smith, Yi Yang, and Elizabeth J. Wilson. 2017. "Marginal Emission Factors Considering Renewables: A Case Study of the U.S. Midcontinent Independent System Operator (MISO) System." *Environmental Science & Technology* 51, no. 19 (October 3, 2017): 11215–11223. ISSN: 0013-936X, 1520-5851, accessed June 5, 2024. https://doi.org/10.1021/acs.est.7b00034. https://pubs.acs.org/doi/10.1021/acs. est.7b00034.
- Matek, Benjamin, and Karl Gawell. 2015. "The Benefits of Baseload Renewables: A Misunderstood Energy Technology." The Electricity Journal 28, no. 2 (March): 101– 112. ISSN: 10406190, accessed June 6, 2024. https://doi.org/10.1016/j.tej.2015.02.001. https://linkinghub.elsevier.com/retrieve/pii/S104061901500024X.
- Mongird, K, V Viswanathan, P Balducci, J Alam, V Fotedar, V Koritarov, and B Hadjerioua. 2019. "Energy Storage Technology and Cost Characterization Report" [in en].
- Muehlenbachs, Lucija, and David Brown. 2023. The Value of Electricity Reliability: Evidence from Battery Adoption. Accessed June 5, 2024. https://doi.org/10.2139/ssrn. 4509955. https://www.ssrn.com/abstract=4509955.
- NREL. 2014. Commercial and Residential Hourly Load Profiles for all TMY3 Locations in the United States. Accessed April 30, 2023. https://catalog.data.gov/dataset/ commercial-and-residential-hourly-load-profiles-for-all-tmy3-locations-in-the-unitedstate-bbc75.

- **NREL.** 2020. 2018 Renewable Energy Grid Integration Data Book. Golden, CO (United States): National Renewable Energy Lab.(NREL).
- **Open EI.** 2023. *The Utility Rate Database*. Accessed April 30, 2023. https://openei.org/wik i/Utility_Rate_Database#:~:text=The%20Utility%20Rate%20Database%20(URDB, of%20Energy%20Energy%20Information%20Administration..
- **Potomac Economics.** 2023b. 2022 State of the Market Report for the ERCOT Electricity Markets.
- Quick Electricity. 2022. "Solar Buyback Plans and Net Metering Incentives in Texas." Accessed April 30, 2022. https://quickelectricity.com/2018-solar-panel-incentivestexas-net-metering-buyback-programs/.
- **REN21.** 2024. Renewables 2024 Global Status Report Collection, Global Overview.
- Rugolo, Jason, and Michael J. Aziz. 2012. "Electricity storage for intermittent renewable sources." *Energy & Environmental Science* 5 (5): 7151. ISSN: 1754-5692, 1754-5706, accessed June 5, 2024. https://doi.org/10.1039/c2ee02542f. https://xlink.rsc.org/?DOI=c2ee02542f.
- **SEIA.** 2022. Solar Market Insight Report 2021 Year in Review. Washington, D.C. (United States): Solar Energy Industries Association.
- Siler-Evans, Kyle, Inês Lima Azevedo, and M. Granger Morgan. 2012. "Marginal Emissions Factors for the U.S. Electricity System." *Environmental Science & Technology* 46, no. 9 (May 1, 2012): 4742–4748. ISSN: 0013-936X, 1520-5851, accessed June 6, 2024. https://doi.org/10.1021/es300145v. https://pubs.acs.org/doi/10.1021/es300145v.
- Smartly Energy. 2023. "The Tesla Powerwall: How Do the Alternatives Measure Up?" Accessed May 5, 2024. https://smartlyenergy.com/blog/the-tesla-powerwall-how-do-the-alternatives-measure-up.
- Weisbach, David A., and Gilbert E. Metcalf. 2009. "The Design of a Carbon Tax." SSRN Electronic Journal, ISSN: 1556-5068, accessed June 5, 2024. https://doi.org/10. 2139/ssrn.1327260. http://www.ssrn.com/abstract=1327260.
- White, Philip R., Joshua D. Rhodes, Eric J.H. Wilson, and Michael E. Webber. 2021. "Quantifying the impact of residential space heating electrification on the Texas electric grid." *Applied Energy* 298 (September): 117113. ISSN: 03062619, accessed June 5, 2024. https://doi.org/10.1016/j.apenergy.2021.117113. https://linkinghub.elsevier. com/retrieve/pii/S0306261921005559.
- Wolak, Frank A. 2021. "Wholesale electricity market design." In Handbook on Electricity Markets, edited by Jean-Michel Glachant, Paul L. Joskow, and Michael G. Pollitt. Edward Elgar Publishing, November 12, 2021. ISBN: 978-1-78897-995-5 978-1-78897-994-8, accessed June 5, 2024. https://doi.org/10.4337/9781788979955.00010. https: //www.elgaronline.com/view/edcoll/9781788979948/9781788979948.00010.xml.

1.9 Appendices

Appendix A: Investment Decisions

The investment decisions can be illustrated by the following example. Suppose a consumer chooses the optimal size of a solar plus storage system that minimizes the bill payments. The customer faces a time-of-use pricing of two periods, whereas the rates he pays in period 1 and period 2 is p_1 and p_2 , respectively. For simplicity, we assume that the solar panel only generates electricity in period 1.

The electricity demanded by the customer in each period is d_1 and d_2 per day. The utility company has a net-metering policy for solar energy. If the solar output is lower than d_1 , the customer is compensated at the retail rate, p_1 . If solar output exceeds d_1 , the excess power will be compensated for at the net surplus compensation rate e where $e < p_1$.

The variable cost of solar and battery is VC_p and VC_s per kW. The initial investment costs are financed by the customer with a discount rate of *i*. The payback period is equal to the lifetime of the solar panel (t_p) and the batteries (t_s) . Therefore, the marginal cost of solar (C_p) and battery (C_s) per kW per day can be calculated as:

$$C_{p,s} = VC_{p,s} \cdot \frac{i}{365(1 - \frac{1}{(1+i)^{t_{p,s}}})}$$

Suppose the capacity of solar and battery is K_p and K_s , respectively. The daily solar output can be described as σK_p , where σ is the average peak-sun hours. When there is no battery technology, the daily marginal benefits of solar energy equal $p_1\sigma$ if $K_p \leq d_1/\sigma$, or $e\sigma$ if $K_p > d_1/\sigma$. As a result, the customer will install a solar panel with capacity d_1/σ as long as $p_1 \geq \frac{C_p}{\sigma} > e^{.50}$ Similarly, a customer will install a solar panel up to the physical constraint if $e \geq \frac{C_p}{\sigma}$.

^{50.} The rate p_1 should be no smaller than 4.8 cents under the following assumptions. The variable cost of solar is \$1,750. The average peak-sun hours is 4. The discount rate is 3%. The maximum lifetime of a solar panel is 25 years. This cut-off value is much smaller than the usual retail rate, indicates strong motivations for installing solar under the TOU rates.

Suppose that a battery has a round-trip efficiency of r. If $p_2 > p_1$, the customer can store a unit of energy at p_1 and sell it at rp_2 , and vice versa. The daily marginal benefits of storage can be given as $rp_2 - p_1$ if $K_s \leq d_2/\Delta\theta$ where $\Delta\theta$ is the difference between the max and min battery charge status. If the TOU rate only applies on weekdays (260 out of 365 days), the customer will install a battery with a capacity of $d_2/\Delta\theta$ if $rp_2 - p_1 \geq \frac{365}{260}C_s$. Moreover, as storage increases the electricity demanded in period 1, it could also increase the solar capacity by d_1/σ . Thus, the total marginal benefit is the sum of the two effects.

Our simple model can be modified to include situations where solar can generate during both peak and off-peak period, and when there is no net-metering policy. For example, suppose that solar can generate electricity during both peak and off-peak period, and the solar output during peak and off-peak period is $\sigma_1 K_p$ and $\sigma_2 K_p$, respectively. The marginal benefits of solar is the output weighted compensation rate. For example, when $\sigma_i K_p < d_i$ for $\forall i = 1, 2$, the marginal benefit is given by $\sigma_1 p_1 + \sigma_2 p_2$. Suppose $\sigma_1 > \sigma_2$ and $d_1 < d_2$, the marginal benefits will drop to $\sigma_1 e_1 + \sigma_2 p_2$ when $\sigma_1 K_p \ge d_1$ and $\sigma_2 K_p < d_1$. If $\sigma_2 K_p \ge$ d_1 , the marginal benefits will drop further to $\sigma_1 e_1 + \sigma_2 e_2$. The investment decision for solar (whether solar generation should cover the peak-period electricity consumption, offpeak-period consumption, or both) is therefore determined by whether the output weighted compensation rates exceed the cut-off value.

The above example can be further modified to analyze a case where there is no netmetering. Suppose the hourly electricity consumption d_{mh} , energy rate p_{mh} , and hourly peak-sun hours σ_{mh} varies with month m = 1, 2, ..., 12 and hour h = 1, 2, ..., 24. Once solar generation exceeds the electricity consumption in a hour, the net surplus solar exports are compensated at a lower rate e_{mh} . Suppose $K_{mh} = \frac{dmh}{\sigma_{mh}}$ is the minimum solar capacity needed to completely offset electricity consumption in month m and hour h. We can rank K_{mh} by size and denote them as $K_1, K_2, ..., K_n, ..., K_N$ where $K_1 < K_2 < ... < K_N$. The marginal benefits is thus $B_n = \sum_{i=1}^n \sigma_i e_i + \sum_{i=n+1}^N \sigma_i p_i$ when $K_p = K_n$. The capacity is chosen at K_{n^*} so that $B_{n^*} > 365 \cdot C_p > B_{n^*+1}.$

Appendix B: DER-CAM Algorithm

This section outlines the objective function and key constraints for the optimization algorithm in our endogenous setting following Boampong and Brown (2020). In our analysis, the algorithm is used to minimize total costs:

$$C = \sum_{m} Fixed + \sum_{m} \sum_{d} \sum_{h} Demand_{m,d,h} \cdot Rate_{m,d,h}$$
$$+ \sum_{j} [(VCC_{j} \cdot CAP_{j}) \cdot An_{j} + CAP_{j} \cdot OMP_{j}]$$
$$- \sum_{m} \sum_{d} \sum_{h} Export_{m,d,h} \cdot Compensation_{m,d,h}$$

| Notation | Variable | | | |
|------------------------|---|--|--|--|
| Fixed | Monthly Fixed Charge | | | |
| $Demand_{m,d,h}$ | Hourly Electricity Demand at Hour h , Day d , and Month m | | | |
| $Rate_{m,d,h}$ | Hourly Electricity Rate at Hour h , Day d , and Month m | | | |
| VCC_{j} | Variable Capital Cost of Technology j | | | |
| CAP_{j} | Capacity of Technology j | | | |
| An_j | Annuity Factor | | | |
| OMP_j | Operational Cost of Technology j | | | |
| $Export_{m,d,h}$ | Solar Exports at Hour h , Day d , and Month m | | | |
| $Compensation_{m,d,h}$ | Solar Compensation Rate at Hour h , Day d , and Month m | | | |

Table 1.11: Notation Description

subject to the following constraints:

1. Supply equals demand at any time.

2. Solar output at each hour is determined by the solar radiation in the corresponding hour, solar panel efficiency, and capcity of solar system.

3. Battery charge and discharge constraints:

(a) the current battery charge status equals the net charge plus the charge status from last period minus enregy losses.

(b) the current battery charge must be higher than the minimum charge status and lower than the maximum charge status required by the battery.

(c) hourly charge and discharge rate is limited by the charge and discharge rate.

Appendix C: Demand Profiles



Figure 1.8: Demand Profile for Supermarkets

Appendix D: Retail Tariff Designs

| Load Profile | CAISO | ERCOT | |
|---------------------|-------|-------|--|
| Energy Rate | 0.114 | 0.063 | |
| TOU2 - Peak | 0.188 | 0.103 | |
| TOU2 - Off-peak | 0.094 | 0.051 | |
| TOU5 - Peak | 0.307 | 0.163 | |
| TOU5 - Off-Peak | 0.061 | 0.033 | |
| TOU2 CT - Peak | 0.222 | 0.145 | |
| TOU2 CT - Off-peak | 0.111 | 0.072 | |
| TOU5 CT - Peak | 0.363 | 0.230 | |
| TOU5 CT - Off-Peak | 0.073 | 0.046 | |
| TOU2 CTH - Peak | 0.290 | 0.228 | |
| TOU2 CTH - Off-peak | 0.145 | 0.114 | |
| TOU5 CTH - Peak | 0.473 | 0.362 | |
| TOU5 CTH - Off-Peak | 0.095 | 0.072 | |
| Avg RTP Rate | 0.039 | 0.039 | |
| Avg CT - RTP Rate | 0.060 | 0.065 | |
| Avg CTH - RTP Rate | 0.101 | 0.117 | |

 Table 1.12:
 Retail Tariff Designs

Appendix E: Average Month-hour RTP



Figure 1.9: CAISO Average Weekday LMPs (kWh)



Figure 1.10: CAISO Average Weekend LMPs (\$/kWh)



Figure 1.11: ERCOT Average Weekday LMPs (\$/kWh)



Figure 1.12: ERCOT Average Weekend LMPs (\$/kWh)

Chapter 2

Estimating Locational Marginal Emissions Considering Transmission Congestion

2.1 Introduction

Electricity generation is one of the most carbon-intensive sectors in the US, accounting for 25% of all carbon emissions (EPA 2024). Transitioning to renewable generation resources such as wind and solar is critical to decarbonizing the energy sector. Because renewable energy resources are often located in remote regions, substantial investments in transmission capacity are needed to deliver energy from renewable-rich remote regions to population centers. However, the expansion of transmission capacity is often too slow to catch up with the rapid expansion of renewable generation, causing severe transmission congestion in certain regions (Joskow 2020).¹

In a restructured electricity market, generators are called upon to supply electricity based on ascending order of price offers to minimize the costs of supplying electricity through a

^{1.} For example, in Texas, where wind penetration is high, the electric system had at least one congested transmission line for 75% of hours in 2022, resulting in a total congestion loss of over 5 billion dollars (Potomac Economics 2023b). In California, which is rich in solar energy, transmission congestion occurred 50% of hours, resulting in a congestion loss of over 1.7 billion US dollars (CAISO 2023a).

procurement auction system. The generator with the lowest price offers will be called upon first, followed by the next lowest-priced generator, until the electricity supply meets demand. Transmission congestion can affect this sequence of supply (also known as the "merit order") as energy generated in congested areas cannot be used to meet supply in uncongested areas, and vice versa. For example, when a population center (powered by coal and natural gas) cannot receive wind energy generated in a remote region due to transmission congestion, a nearby fossil fuel generator must be called upon to meet local electricity demand. Consequently, transmission congestion can have a locational impact on emissions. In particular, it can affect the locational marginal emissions factors (MEFs), which measure the change in locational emissions for an incremental change in local demand. MEFs are a critical parameter in the evaluation of policies and technologies that can shift energy loads, such as distributed energy resources (DER) or electric vehicles.²

In addition, large load centers, such as bitcoin mining operations and data centers, have increasingly impacted the electricity grid. In particular, where to locate these load centers has become a topic of considerable debate (Rhodes et al. 2021; Stoll et al. 2023). For instance, it has been argued that placing load centers in regions with high renewable penetration can significantly reduce carbon emissions and increase the economic value of renewable energy resources (Stoll et al. 2023). Understanding locational MEFs is crucial for assessing the environmental impact of these load centers.

To illustrate how transmission congestion affects MEFs, consider two regions: a remote region powered by wind energy and a densely populated region powered by coal and natural gas. When there is no transmission constraint between the two regions, generators will be dispatched based on the merit order, and the two regions will share the same MEFs because the same generator can respond to a demand increase in either region. However, when

^{2.} DER technologies are customer-sited technologies that can generate and store energy on or near the site of an end user (Brown and Sappington 2017a). In this paper, DER technologies refer to rooftop solar panels and behind-the-meter (BTM) batteries.

congestion occurs, the merit order will split into two parts as local demand can only be met by local generators, leading to different MEFs. For example, in the absence of congestion, when the supply of wind energy in the remote region exceeds local demand, the excess energy will be delivered to the populated region. When there is a constraint to the outflow of wind energy, the extra wind energy will be curtailed. In this case, a unit increase in local electricity demand in the remote region will not increase emissions, resulting in zero marginal emissions in this region. In the meantime, local fossil fuel generators will be needed to meet the demand increase in the populated region, resulting in positive MEFs.

Different locational MEFs caused by transmission congestion will lead to different locational environmental values of DERs, measured by the potential mitigation of carbon emissions. DER technologies can reduce or shift load in certain hours and impact emissions based on the MEFs of the corresponding hours. For example, a battery can reduce electricity demand during discharging and increase electricity demand during charging. Based on the load shifts and MEFs during charging and discharging, the environmental value of a battery can be calculated as the net emissions reduction (avoided emissions minus created emissions), which will vary by location, given different locational MEFs.

Similarly, the environmental impacts of a load center may differ significantly depending on its location and the corresponding locational MEFs. By strategically placing load centers in regions with lower MEFs, carbon emissions can be minimized.

For simplicity and given data limitations, MEFs are measured with the assumption that all generators are used to meet local demand, regardless of local transmission conditions (e.g., Siler-Evans et al. 2012; Graff Zivin et al. 2014; Sexton et al. 2018; Holland et al. 2022). As a result, MEFs are often measured in broad regions with minimal energy imports and exports across regions (e.g., Siler-Evans et al. 2012; Graff Zivin et al. 2014) or in small subregions, assuming that local demand is met entirely with local generation (e.g., Sexton et al. 2018; Holland et al. 2022). In the first case, MEFs can only provide an average measure across multiple subregions and overlooks locational differences. In the second case, the MEFs do not capture generation resources located outside of the subregion. As transmission congestion can greatly affect the marginal emissions in a particular subregion, locational MEFs that take transmission conditions into consideration are needed to capture the locational environmental values of DER technologies or to evaluate other energy-shifting technologies and policies.

Our paper proposes a novel estimation method to estimate locational MEFs while considering transmission congestion. We use an endogenous regime-switching model to estimate the MEFs in congested and uncongested regimes. Then, we use the congestion-weighted MEFs to calculate the environmental values of DER technologies located in remote regions with high wind penetration versus more population-dense regions powered primarily by natural gas and coal. Our paper has two primary objectives. First, we aim to understand how transmission congestion could affect locational MEFs in different regions, especially with different levels of renewable penetration. Second, we aim to understand how locational environmental values of DERs could be impacted by transmission congestion, highlighting the importance of using locational MEFs to make policy decisions.

Our paper focuses on the territory of the Electric Reliability Council of Texas (ERCOT). This territory is particularly interesting because more than 80% of wind energy resources are located in one remote subregion (the West load zone) of ERCOT (Fell et al. 2021). Transmission congestion has become a major barrier to renewable integration in ERCOT (Deniel and Gomberg 2021). In 2020, 5% of all wind generation in ERCOT had to be curtailed due to insufficient network capacity, one the highest among regional transmission organizations (Deniel and Gomberg 2021).

We find that the traditional estimation method can greatly overestimate or underestimate the locational MEFs in regions with high wind penetration. In particular, when congestion is present, the average MEFs in the West load zone are 20% less than the ERCOT-wide MEFs estimate. In addition, the MEFs estimated using our method are 30% higher than the alternative MEFs estimation that assumes that the West zone is fully "independent". Applying our estimation to the DER technologies, we find that transmission congestion can greatly reduce the environmental values of solar in remote windy areas. However, congestion tends to increase the variance of the MEFs and may increase the environmental values of energy storage systems. For example, when the system is congested due to excessive wind energy, a storage system located in the remote region can store clean power that would otherwise be curtailed and discharge the stored power when the system is uncongested.

This paper sheds light on the ongoing rate design discussion regarding the incorporation of environmental attributes in the compensation of DERs, especially rooftop solar panels. The environmental value is a crucial component in the economic value of a DER and should be included in the "value of DER" tariffs where the DER compensation aligns with the underlying economic value of a DER (Minnesota Department of Commerce 2014).³ However, the environmental value is generally set at a predetermined rate, leading to a suboptimal level of DER compensation. Our paper highlights the importance of incorporating transmission congestion when considering the locational environmental values of DER.⁴

Our paper contributes to three strands of literature. The first relates to a large literature that relies on MEFs to evaluate policy and technology interventions, particularly energy storage and EVs (e.g., Graff Zivin et al. 2014; Siler-Evans et al. 2012; Holland et al. 2016). Previous studies that estimate MEFs have focused on coal and natural gas generators only, assuming non-emitting sources, such as wind, solar, and hydro, rarely contribute to marginal generation (Li et al. 2017). However, due to transmission constraints and improved dispatch-

^{3.} See Section 2.2.2 for a detailed discussion.

^{4.} Section 2.2.2 provides a detailed discussion on the ongoing rate design discussion regarding DER environmental values and compensations.

ability for renewables, renewables could affect MEFs substantially.⁵ In addition, MEFs from previous studies do not consider local constraints, and therefore overlook regional differences. To the best of our knowledge, our paper is the first to investigate how congestion affects MEFs in regions with different energy sources. This provides valuable insights for evaluating locational environmental values for DER technologies and EVs.

Second, our paper contributes to the literature that looks at the locational environmental values of DERs based on locational MEFs. Our paper is related to work by Callaway et al. (2018) and Sexton et al. (2018), who focus on the locational environmental values of rooftop solar across the US based on locational MEFs, and Fisher and Apt (2017), who use MEFs in different jurisdictions to estimate the environmental value of battery storage technologies. Our paper differs from theirs by considering the impact of transmission congestion. In contrast, they assume that locational MEFs are only affected by local generators.

Third, our paper contributes to the literature that identifies congestion as a barrier to renewable integration (e.g., Fell et al. 2021). In particular, our research sheds light on how transmission congestion will affect the environmental values of rooftop solar and batteries. Several papers have documented that transmission congestion will reduce the environmental values of renewables (e.g., Hitaj 2015; Fell et al. 2021). Our paper differs from theirs by first estimating the locational MEFs. This allows us to easily compare environmental outcomes with different counterfactual settings.

The paper proceeds as follows. Section 2.2 provides the conceptual framework of our paper. Section 2.3 provides the background information on Texas's electricity market. Section 2.4 details the data used in our analysis. Section 2.5 presents the methodology for estimating the MEFs. Section 2.6 presents the main results. Section 2.7 discusses the environmental values of DER. Section 2.8 concludes.

^{5.} For example, neglecting the effects of renewables can lead to a 30% overestimation of MEFs in the Midcontinent Independent System Operator (MISO) System in the US (Li et al. 2017).

2.2 Conceptual Framework

This section illustrates how transmission congestion could affect the locational environmental values of DER technologies using a simple numerical example.

2.2.1 Congestion and Locational MEFs

Consider two regions: a remote region (West) powered by wind farms and natural gas power generators and a population-dense region (East) powered by natural gas and coal-fired generators. The two regions are connected through a transmission line, which will be congested whenever the power flow (Q) between the two regions exceeds the transmission capacity, K. Electricity demand in each region is represented by D_i where $i = \{West, East\}$.

Figure 2.1 summarizes the assumed generation capacity, marginal cost, and marginal emissions rate for the power plants in each region. The wind farms in the West can provide 1,000 MWh of wind energy at zero marginal costs and zero emissions. The natural gas power plants in the West can provide another 1,000 MWh of electricity, with a marginal cost of \$50 per MWh and a marginal emissions rate of 0.5 tonnes per MWh. In the East, the coal-fired power plants can provide 2,500 MWh of electricity, with a marginal cost of \$20 per MWh and a marginal emissions rate of 1 tonne per MWh. The natural gas plants can provide another 1,500 MWh of power in the East, with a marginal cost of \$40 per MWh and a marginal emissions rate of 0.5 tonnes per MWh. We assume the maximum transmission capacity between the two regions is 500 MWh. The generators will be dispatched based on their marginal costs. Further, we assume that the demand for energy in the East and West regions falls within the intervals of $D_E \in [2,000,4,000]$ and $D_W \in [0,2,000]$, respectively.



Figure 2.1: Numerical Example Setting

In this simple example, we will show that the electricity demand in the West will determine whether the system is congested. Further, when the system is uncongested, the marginal emissions rates are the same across regions and will be determined by the marginal power plant⁶ in the East. When the system is congested, the marginal emissions rate in the West will be equal to or lower than that in the East, depending on the electricity demand in both regions.

Figure 2.2 presents the locational price and marginal emissions under different combinations of loads in the East and West. If $D_W < 500$, power will flow from the West to meet demand in the East via generation from the wind plant. The system will be congested (outflow congestion in the West) because the transmission capacity is lower than $Q = 1000 - D_W$, and the extra power $(500 - D_W)$ from wind will be curtailed. In this case, a marginal increase in the West load will not increase emissions but reduce wind curtailment. As a result, the MEF in the West will be zero. In the East, the marginal emissions rate will be either 1 tonnes/MWh (if $D_E \in [2000, 3000)$) or 0.5 tonnes/MWh (if $D_E \in [3000, 4000]$) depending on the marginal generation technology. It is worth noting that during congestion in the current example, a slight expansion of the transmission capacity will not change the marginal emissions rate in either region.

^{6.} The marginal power plant is the power generator used to meet the last MWh of electricity demand.



Figure 2.2: Numerical Example Illustration

If $500 \leq D_W \leq 1,000$, the wind power will flow from the West to the East without congestion. The marginal emissions rate will be the same across regions. Similarly, if $1000 < D_W \leq 1,500$, power will flow from the East to the West without congestion, leading to the same MEFs across regions. For both cases, the system MEF is determined by the system load $(D_W + D_E)$. As shown in Figure 2.2, the marginal emissions rate is determined by the marginal generation technology in the East. If $D_W + D_E \leq 3,500$, coal-fired plants will be the marginal plants whereas natural gas plants will be the marginal plants if $D_W + D_E > 3,500$.

Finally, when $D_W > 1,500$, the power will be restricted from the East to the West, causing inflow congestion to the West. For example, when the West load is 1,800 MWh, the marginal emissions rate in the West will be determined by its natural gas units. In our simple example, natural gas power plants in both regions have a marginal emissions rate of 0.5 tonnes/MWh. When the marginal fuel in the East is natural gas, the marginal emissions rate across the regions will be the same. However, if we relax the assumption for D_E and let $D_E < 2000$, coal will be used as the marginal fuel in the East. In this case, an inflow congestion in the West will still lead to lower West MEFs than the East.

Our simple example indicates that both inflow and outflow congestions can reduce the MEFs in clean regions during specific periods. In the outflow congestion case, renewable generation could be on the margin in the clean region, leading to curtailments and zero marginal emissions rate. In the inflow congestion case, the marginal emissions rate in the West could be lower when natural gas generation in the West is displacing coal generation in the East. The congestion will affect the merit order such that a less cost-effective power plant will be dispatched to meet electricity demand. However, a less cost-effective power plant is not necessarily more polluting (e.g. coal and natural gas).

Due to simplifying assumptions of generation resources and fixed renewable generation, the West load can solely determine the congestion status in our numerical example. In reality, the congestion status at a particular time will be determined by both regional electricity demands and renewable generations. In addition, it is worth noting that, other than congestion, factors such as changes in fuel prices may also affect the merit order of the electricity supply.

2.2.2 The Environmental Values of DERs

DER technologies, especially rooftop solar panels, are expanding rapidly worldwide, and it is critical to understand how these resources will impact emissions from the electricity sector (SEIA 2022). In particular, understanding the environmental value, as a vital component of the total economic values of DER technologies, plays an essential role in designing subsidization and compensation policies for such technologies.

To promote the adoption of rooftop solar panels, more than 40 US states have implemented net-metering policies that compensate for net generation from rooftop solar at the prevailing retail price of electricity (Brown and Sappington 2017b). However, several states have transitioned from this traditional net-metering policy towards an avoided-cost-based rate reflecting rooftop solar's time-varying economic values (NCCETC 2023). The main reason for this transition is to eliminate the cross-subsidization concerns in the traditional net-metering policies (NCCETC 2023).⁷ For example, the State of Minnesota implemented the value of solar (VOA) tariff in 2014 as an alternative to the net-metering policy, which seeks to quantify the real value of rooftop solar based on avoided costs (Minnesota Department of Commerce 2014). Similarly, the New York State Public Service Commission (NYSPSC) established the Value of Distributed Energy Resource (VDER) in 2017 as a new mechanism to compensate for electricity generation from DERs (Bowen et al. 2022).

In these new tariffs, the compensation rate for DERs is generally composed of three main components: avoided energy cost, avoided capacity cost, and avoided environmental cost. The avoided environmental costs of carbon are calculated by multiplying the social cost of the carbon (\$ per tonne) by its estimated marginal emissions rate (tonnes per MWh).⁸ Depending on the social costs of carbon and location, the avoided environmental costs can account for 10% to 20% of the current DER compensation rate and are expected to increase as the social cost of carbon rises (Hochul and Seggos 2023; Energy Sage 2023b). The marginal emissions rates used to calculate the avoided environmental costs are kept fixed by the authorities to avoid complications for the DER customers (Bowen et al. 2022). For example, the NYSPSC uses the average MEFs as the marginal emissions rate even though the MEFs are estimated for each hour of the day and each month of the year (Bowen et al. 2022). Further, the avoided environmental cost is locked for 25 years, meaning a customer will be given the same environmental credit for every kWh he supplied to the grid for the entire

^{7.} It is worth noting that a relatively "simple" solution is to price DERs based on the locational marginal prices, coupled with a carbon pricing mechanism. However, in practice, customers are not subject to these price signals.

^{8.} Other pollutants, such as NOx, are calculated using the same method but account for less than 3% of the avoided environmental costs (Minnesota Department of Commerce 2014). Therefore, our paper focus on the environmental value of DERs resulting from carbon mitigation.

lifecycle of the solar panel, regardless of when and where the power is generated (Minnesota Department of Commerce 2014; Bowen et al. 2022).

As shown in section 2.2.1, congestion can affect the marginal emissions rate based on location and time, leading to different environmental values of DERs in different locations. A solar project can reduce more emissions if it is located in a region where the marginal emissions rate during solar generating hours is high. Similarly, battery storage has a higher potential to reduce emissions if it is located in a region where the marginal emissions rate during battery charging and discharging spreads more extensively. In our numerical example, a solar panel can reduce more emissions if it is located in the East. However, a battery system can reduce more emissions if it is located in the West due to the higher MEF difference with and without congestion. Therefore, it is essential to consider DER's locational MEFs and locational environmental values when designing DER compensation policies.

Apart from the direct environmental impact of MEFs, a DER technology can alleviate congestion, indirectly affecting their environmental outcomes. For example, a battery can alleviate congestion by discharging power when the system is congested. Developing a dispatching model based on merit order in both regions is necessary to estimate these indirect effects. However, this is out of the scope of the current analysis. Our study will focus on the direct environmental impact of the MEFs.

2.3 Texas's Electricity Market

The wholesale electricity market in Texas is operated by The Electric Reliability Council of Texas (ERCOT), which covers 75% of Texas's land area and serves about 90% of Texas's electric load (ERCOT 2022; 2023a).

ERCOT optimizes energy dispatch using a security-constrained economic dispatch

(SCED) model, which provides the least-cost energy dispatch subject to the grid's transmission constraint. The real-time SCED model calculates the nodal locational marginal prices (LMPs) every 5 minutes for thousands of specific locations in the electricity grid based on the network's real-time transmission constraints. The LMPs in ERCOT comprise of real-time wholesale electricity prices and congestion costs. The LMPs reflect the value of electricity at different locations in the grid and are designed to motivate efficient location of new generation and transmission investments. Unlike many other jurisdictions in the US, as of 2023, the marginal transmission system losses (line losses) have not been incorporated into the LMP mechanisms in ERCOT (Potomac Economics 2024). Therefore, when the system is uncongested, the LMP in each node will converge to a uniform price. When congestion arises, the LMPs will increase in regions where power in-flow is constrained and decreases in regions where power out-flow is constrained (Brown et al. 2020).

Texas is the leading state in the US in terms of electricity generation, accounting for 12.3% of all electricity production in the US (EIA 2023). Natural gas power plants are the backbone of the generation resources in Texas, accounting for 42% of all electricity generation in 2021. Renewable energy resources, especially wind, are essential in ERCOT's generation mix, with more than 20% of electricity generation from wind and another 4% from solar. The share of electricity generation from coal has dropped significantly in the last decade, from 40% in 2011 to 19% in 2021 (EIA 2023).

Texas leads the US in wind generation, but its current electricity infrastructure cannot transmit or store all electricity generated from renewable energy resources, resulting in frequent curtailment of renewables (Potomac Economics 2023b). ERCOT divides its territory into four major load zones: West, South, North, and Houston (see Figure 2.3). Load zones are designed so that electricity transmission faces fewer limitations within zones than across zones (Fell et al. 2021). The wind energy resources are mainly located in the West and South zones of ERCOT, away from the population centers in the North and Houston areas. As a result, part of renewable generations will have to be curtailed when facing transmission constraints between distant wind farms and large population centers. During our sample period, about 12 TWh of wind power have been curtailed, accounting for 3% of all wind generation. However, with more wind projects being developed in recent years, wind curtailment reached 5% in 2021, and the trend is expected to continue without substantial investment for transmission upgrades. Most solar projects are also located in the West zone, and solar energy accounts for 15% of all electricity generation in the West. With rapidly expanding solar projects in Texas, solar curtailment also increases rapidly. In 2021, 10% of all solar energy had been curtailed (Potomac Economics 2023b).



Figure 2.3: Generation and Transmission Infrastructures in ERCOT Load Zones Source: EIA Energy Infrastructure and Resources Maps

As the West and South zones have the highest share of wind generation, more than 90% of renewable curtailment occurs in these areas due to transmission congestion between or within the two zones (BTU Analytics 2023). The curtailment is more evenly distributed across all wind farms in the West zone, as the transmission bottleneck occurs mainly across zones between the West and the South. In contrast, in the South zone, six wind projects account for more than half of the curtailments, which implies that transmission congestion within the South zone plays a significant role in renewable curtailment (BTU Analytics 2023).

2.4 Data

To perform our analysis, we collect hourly data on load, generation, emissions, weather conditions, and LMPs in ERCOT between 2017 and 2022.⁹ This period is particularly interesting due to the rapid expansion of wind energy resources and frequent wind curtailments due to system congestion. During the sample period, ERCOT added approximately 2,000 miles of new transmission lines to the grid system, representing a 4% expansion of the electric grid system in 2017 (ERCOT 2023b).¹⁰

To construct a measure for congestion status in each load zone in ERCOT, we collect the 15-minute zonal-level LMPs from ERCOT's real-time market for each load zone. The zonal-level LMPs are calculated by taking the load-weighted average of all nodal LMPs within a given load zone. The LMPs will converge to a uniform price when congestion does not occur. Therefore, the spread of different zonal LMPs can reflect the shadow price of constraints and serve as an indicator of congestion status across load zones. Section 2.5.1 provides more details on how we use LMPs to measure congestion.

To calculate the MEFs, we first collect emissions data from the Environmental Protection Agency (EPA)'s Continuous Emission Monitoring System (CEMS). CEMS provides hourly emissions data for each fossil-fuel generator with a capacity of 25 MW or higher in the US (EPA 2023a). To assign carbon emissions from a generator to a specific load zone in ERCOT, we use the EIA-860 survey provided by the EPA to pair each generator in Texas from the CEMS to a specific county. The EIA-860 survey collects detailed information for each generator in the US with a capacity of 1 MW or higher (EIA 2023). Then, we use the load zone map provided by ERCOT to assign each generator to a load zone based on its

^{9.} ERCOT does not publish data on renewable curtailments, so we cannot construct congestion measures based on curtailments.

^{10.} For reference, the CREZ (Competitive Renewable Energy Zones) project, conducted from 2005 to 2013, added 3,600 miles of transmission lines (ERCOT 2014). The purpose was to connect renewable-rich regions to population centers (ERCOT 2014).

location.

The CEMS only contains generation data for fossil fuel generators. To control for renewable generation, we collect ERCOT-wide hourly generation data for wind and solar resources from ERCOT's annual Fuel Mix Report (ERCOT 2023a). To obtain the zonal-level renewable generation estimates, we collect the most recent hourly zonal-level renewable generation data from ERCOT's Intermittent Renewable Resources data.¹¹ We then regress the zonallevel renewable generation data on ERCOT-wide generation, and estimate the zonal-level renewable generation based on the regression model. To control for shifts in electricity demand, we collect hourly load data from ERCOT at the ERCOT and zonal levels (ERCOT 2024).

Finally, to use as potential instruments for congestion status, we collect weather station data in Texas using National Centers for Environmental Information's Integrated Surface Dataset (ISD). The ISD contains hourly weather data for over 35,000 stations worldwide and 197 weather stations located in Texas (NOAA 2023). The dataset covers a wide range of climate data, including temperatures, wind speed, and wind directions. A comprehensive discussion on constructing instruments for congestion status using weather station data is provided in Section 2.5.2.

To evaluate the environmental values for solar panels, we collect solar irradiation data from the National Renewable Energy Laboratory (NREL)'s Typical Meteorological Year version 3 (TMY3) data sets (NREL 2014). The TMY3 dataset provides hourly solar irradiation data for 1,020 locations throughout the US. The irradiation data from this dataset allows users to compare the performances of solar panels of different configurations and locations. For comparability, we use the normalized solar panel output based on the average solar

^{11.} We are unable to obtain historical data from ERCOT, as it only retains a 7-day record for load-zone-specific wind and solar generation data. As a result, we only acquired zonal-level renewable generation data for February 2024.

irradiation in Texas to calculate the solar output from a rooftop solar.

Table 2.1 summarizes the hourly carbon emissions, hourly electricity load, and hourly LMPs for each load zone and hourly renewable generation in ERCOT. On average, the West zone accounts for 10% of the load in ERCOT but only emits 3% of carbon emissions due to the high penetration of wind and solar. The lowest hourly emissions in the West zone is 0, indicating that renewable energy generation 100% covers the West load. However, the average LMPs in the West are higher than other load zones due to very high LMPs in times with inflow congestion and low renewable outputs.

2.5 Methodology

This section presents the methodology used in this study. In section 2.5.1, we present two measures for the congestion status: an ERCOT-wide congestion measure and a load-zone-specific congestion measure. In section 2.5.2, we present two empirical strategies to examine the effect of congestion on MEFs. We first look at how ERCOT-wide congestion affects the ERCOT-wide MEFs. Then, we use an endogenous regime-switching model to investigate how load-zone-specific congestion will affect the locational MEFs of the corresponding load zone.

2.5.1 Congestion

We measure the congestion status of ERCOT based on its zonal-level LMPs in the real-time market, which is calculated by taking the average of all nodal-level LMPs in a load zone. The LMPs in ERCOT are comprised of wholesale electricity price and congestion cost. When congestion arises, the LMPs will increase in regions where power in-flow is constrained and decrease in regions where power out-flow is constrained (Brown et al. 2020). Therefore,

| | Units | Mean | Std. Dev. | Min | Max |
|-------------------------|---|--------------|-------------|--------------|--------------|
| Hourly ERCOT CO_2 | Tonnes | 22,475.2 | 7,126.4 | 7,435.4 | 61,761.4 |
| Hourly West CO_2 | Tonnes | 672.2 | 331.9 | 0 | $2,\!153.4$ |
| Hourly South CO_2 | Tonnes | 5,160.1 | 1,952.1 | 931.3 | 15,754.0 |
| Hourly North CO_2 | Tonnes | $10,\!974.4$ | 3,829.8 | 2,065.5 | 29,566.4 |
| Hourly Houston CO_2 | Tonnes | $5,\!668.5$ | 1,545.8 | 2,732.9 | $14,\!482.0$ |
| ERCOT Load | MWh | 43,221.1 | 9,739.0 | $25,\!566.1$ | 80,037.8 |
| West Load | MWh | 4,483.2 | 827.3 | 2,372.6 | 7,922.7 |
| South Load | MWh | $10,\!468.3$ | $2,\!661.5$ | $5,\!964.0$ | $20,\!127.7$ |
| North Load | MWh | $17,\!977.3$ | 4,537.7 | $10,\!212.7$ | 35,221.3 |
| Houston Load | MWh | 12,349.2 | 2,710.6 | $7,\!422.6$ | 22,014.0 |
| West LMP | \$/MWh | 41.37 | 259.40 | -31.65 | 9,299.49 |
| South LMP | MWh | 38.50 | 256.22 | -38.73 | 9,235.13 |
| North LMP | MWh | 35.80 | 256.75 | -58.52 | 9,304.38 |
| Houston LMP | MWh | 37.69 | 257.53 | -31.65 | 9,226.05 |
| Average Wind Direction | Degree | 186.5 | 89.6 | 10 | 360 |
| Average Wind Speed | $\rm km/h$ | 37.3 | 15.2 | 6.3 | 108.1 |
| Average Temperature | $^{\circ}\mathrm{C}/^{\circ}\mathrm{F}$ | 19.7/67.5 | 8.6/47.4 | 0.4/32.7 | 39.5/103.1 |
| Hourly Solar Generation | MWh | 725.6 | 1,371.7 | 0 | 10,248.7 |
| Hourly Wind Generation | MWh | 8,782.3 | 4,846.6 | 0 | 26,806.6 |

 Table 2.1: Summary Statistics

we can use the zonal-level LMP spread across regions to measure the congestion status in a system (Fell et al. 2021).¹² We first measure the overall congestion status in the entire ERCOT territory. Then, we measure the congestion status for each of ERCOT's load zones.

We create a binary indicator $Congest_t$ that equals one when congestion arises at time t, equals zero if otherwise. For the ERCOT-wide congestion status, we define $Congest_t = 1$ if the LMP difference between any two load zones exceeds some cutoff value. We use \$5 as our baseline cutoff value, which strongly indicates the system is congested. Alternative cutoff values of \$1 and \$10 are examined in the robustness checks.

Using our baseline specification, we find that from 2017 to 2022, the system in ERCOT was congested 38% of the time.¹³ Figure 2.4 presents the average percentage of congestion occurrence by the hour of the day. Congestion frequently occurs at all hours of the day but is more likely to occur during high-demand hours around 4:00 PM. The congestion status varies across years due to both renewable penetration and transmission expansion. Particularly, increased penetration of renewables has led to higher levels of congestion in ERCOT. Appendix F presents the evolution of congestion status over the years.

^{12.} A wider zonal-level Locational Marginal Price (LMP) spread can usually indicate a more severe congestion status between two load zones. For example, when a load zone faces congestion at just one of its nodes, the zonal-level LMP spread tends to be low. In contrast, when congestion occurs at the bottleneck between two load zones, all nodes within the congested zone will have either lower or higher LMPs than the uncongested zone.

^{13.} Under alternative cutoff values of \$1 and \$10, the ERCOT system will experience congestion for 67% and 17% of the time, respectively. In comparison, ERCOT is congested 42% of the time as measured in Fell et al. (2021). According to Potomac Economics (2023a), from 2018 to 2020, there is at least one binding constraint for more than 70% of the time. As we focus on more severe congestion, the average congestion time in our study is lower.



Figure 2.4: Congestion Occurrence by Hour

By examining the data, when the system is congested, a particular load zone tends to have significantly higher or lower LMP than the other zones. We identify the most congested zone and decompose congestion by load zone and power flow types. In particular, a zone is considered "congested" if its LMP and the mean of the LMPs from the other zones have a spread larger than \$5. It is worth noting that two zones could be congested simultaneously under this definition. As mentioned above, if the LMP in the congested zone is lower than the uncongested region, there is an outflow congestion; if it is higher, there is an inflow congestion in the congested zone. The results using the \$5 benchmark are shown in Figure 2.5. During our sample period, the West zone accounts for over 60% of all hourly congestion with both inflow and outflow constraints. In comparison, according to Potomac Economics (2023a), the West zone accounts for around 50% of the congestion in 2019 and 2020, driven by high renewable output. It is not surprising since the West zone is sparsely populated but accounts for around 80% of ERCOT's wind generation.



Figure 2.5: Congestion Decomposition by Load Zones

It is worth noting that our congestion measure is based on zonal-level LMPs. The spread of LMPs for each node within the zone is required to differentiate between congestion within and across zones. Unfortunately, nodal-level LMPs from ERCOT are only accessible through an official data request, which we have not pursued due to the associated costs. Therefore, we defer this aspect to future research. As a result, our congestion measure may underestimate the congestion status in a load zone with both inflow and outflow constraints, as the high and low nodal-level LMPs may cancel each other out. As mentioned in section 2.3, congestion occurs frequently within the South zone but less frequently within the West zone. Therefore, our measure for West congestion is relatively accurate.

2.5.2 Empirical Strategy

We use several empirical strategies to investigate how transmission congestion affects marginal emissions in ERCOT. First, we examine how ERCOT-wide transmission congestion will affect the average ERCOT-wide marginal emissions. During periods of congestion, the regional MEFs may increase or decrease depending on the changes in the marginal generators. Accordingly, the average ERCOT-wide MEFs may rise or fall, depending on the relative changes in regional MEFs and the respective demand levels for each subregion.¹⁴ Since ERCOT-wide MEFs do not account for the location and source of the unit increase in electricity demand, this increase will represent the probability of where the demand might originate. Consequently, the average ERCOT-wide MEFs can represent the locational MEFs.

We are interested in ERCOT-wide MEFs because they provide a general picture of how transmission congestion affects marginal and overall carbon emissions. If ERCOT-wide MEFs are higher during periods of congestion compared to uncongested periods, then, on average, meeting a unit increase in electricity demand in ERCOT during congested hours will result in more pollution.

Next, we investigate how transmission congestion of a particular load zone affects each zone's locational MEFs using an endogenous regime-switching model. During uncongested periods, the locational MEFs should be equal to the ERCOT-wide MEFs. However, when congestion arises, the locational MEFs are determined by the marginal generator of the specific load zone. Consequently, the locational MEFs tend to be lower in regions with higher renewable penetration when renewable energy could appear on the margin during congested hours. In contrast, if dirtier sources of electricity generation must be called upon to generate electricity when congestion arises, the locational MEFs will be higher during congested hours.

^{14.} For example, suppose ERCOT has two subregions, West and East. The energy demand in the East and West are 2,000 MWh and 1,000 MWh, respectively. During a congestion, the West MEF is 0 kg/kWh and the East MEF is 0.8 kg/kWh. The average ERCOT-wide MEFs will be the load-weighted average MEFs of the two subregions, which is given by 0*1/3+0.8*2/3=0.53 kg/kWh. Suppose the ERCOT-wide MEF is 0.5 kg/kWh when the system is uncongested. Congestion leads to an average increase of 0.03 kg/kWh in the MEFs.

ERCOT-Wide MEFs

We start by describing the primary empirical strategy we use to estimate the ERCOT-wide MEFs. Adapted from Fell et al. (2021), we estimate the following model:

$$E_{hdmy} = \beta_0 + \beta_1 D_{hdmy} + \beta_2 Congest_{hdmy} + \beta_3 D_{hdmy} \cdot Congest_{hdmy} + \sum_j \theta_j X_{hdmy} + \gamma_h + \eta_m + \delta_d + \tau_y + \varepsilon_{hdmy}$$

$$(2.1)$$

where E_{hdmy} is the total emissions from all fossil fuel power generators located in the ERCOT territory for hour h, day d, month m, and year y. D_{hdmy} is the hourly demand in ERCOT.

Congest_{hdmy} is a dummy variable indicating the congestion status, as described in Section 2.5.1. $\sum_{j} \theta_{j} X_{hdmy}$ represents a set of control variables, specifically including the average regional (zonal) temperatures for each load zone, and the total wind and solar generation in ERCOT, with θ_{j} being the corresponding coefficient for each variable. We include regional temperatures and total renewable generation because they may impact both congestion and emissions, leading to endogeneity concerns. For instance, increased wind generation might raise the probability of congestion while simultaneously lowering overall system emissions. Similarly, changes in regional temperatures may impact regional demand, indirectly influencing ERCOT-wide emissions and congestion status. Consequently, omitting these variables could introduce omitted variable bias.

Finally, γ_h , η_m , and δ_d are dummy variables that stand for the hour of the day, the month of the year, and the day of the week, respectively. We include these time variables to control for unobserved source of variation that may correlate with the electricity demand. To control for long-term factors such as increase in generation and transmission capacities, we include the year dummy variable τ_y . To deal with heteroskedasticity and autocorrelation, we use Newey-West standard errors in this estimation with 24 lags. If congestion is not considered, β_2 and β_3 are set to zero and β_1 captures the increase in emissions from a 1 MWh increase in ERCOT-wide electricity demand (also known as the average ERCOT-wide MEF). Our coefficient of interest is β_3 , which captures the changes in MEF caused by congestion and $\beta_1 + \beta_3$ captures the average MEF in ERCOT when congestion occurs.

The estimated coefficients in Equation (2.1) could be biased if congestion is endogenous, even after controlling for regional temperatures and total renewable generation. For example, endogeneity may arise due to unobserved plant outages that affect both congestion and emissions. Suppose a coal-fired generator is closed for maintenance; this can lead to regional congestion and reduce overall emissions in the jurisdiction. Consequently, this may systematically bias the estimated effects of congestion on emissions downward.

To address this potential endogeneity problem, we employ an instrumental variable approach. Weather conditions such as wind direction, wind speed, and air temperature have been utilized as instruments for congestion in previous studies (Fell et al. 2021; Woerman 2018). Air temperature and wind conditions can influence the transmission capacity of a transmission line because the capacity of a transmission line is constrained by its thermal limit. More specifically, the transmission line is limited by the maximum temperature the conductor can safely reach without risking damage or failure. Cooler air temperatures, higher wind speeds, and a more direct wind direction enhance the cooling of the conductor, thereby allowing the line to carry more current without exceeding its thermal limit.¹⁵ Consequently, during windy and low-temperature days, the transmission line can operate at a higher capacity due to the increased cooling effect.

However, a potential concern in using wind conditions and air temperatures as instruments

^{15.} Both wind speed and wind direction can affect the capacity of a transmission line due to cooling effect (Gentle et al. 2012). In particular, Gentle et al. (2012) find that the wind can increase the transmission capacity by 10% to 40% in Idaho most of the time. Phillips (2014) find that the parallel wind only achieves 40% of the cooling effect compared to the perpendicular wind.

is that they may still affect total emissions through channels other than their impact on congestion, such as electricity demand and renewable generation. To address this concern, we first control for regional temperatures and renewable generation as in Equation (2.1). After controlling for wind generation, the impact of wind speed and wind directions on emissions through wind generation can be teased out, making wind conditions valid instruments for congestion status. Similarly, after controlling for electricity demand, air temperatures should only affect emissions through their impact on transmission congestion. These statements hold true as long as the instruments for congestion are uncorrelated with the error term after controlling for demand and renewable generation (Wooldridge 2010).¹⁶

In addition, to ensure that air temperatures and wind conditions affect emissions solely through their impact on transmission congestion, and to ensure that these instrumental variables are not weak, we can choose air temperatures and wind conditions from certain locations as instruments. Specifically, the selected air temperatures and wind conditions should be correlated with congestion status but should not influence emissions through channels other than congestion after controlling for regional temperatures and total renewable generation.

It is reasonable to assume that some unobservable confounding factors, such as local demand, will only be affected by local weather conditions. In other words, only weather conditions near large load centers (e.g., metropolitan areas) or wind and solar farms may influence emissions through effects on local demand or renewable generation. However, given the complexity of the power grid, the likelihood of transmission congestion may change due

^{16.} It is important to consider that wind speed, temperature, and wind direction may also influence plant outages. For example, extreme weather conditions, including high winds and low temperatures, may contribute to operational issues or force outages at power plants. However, these factors are generally considered less likely to directly impact the specific relationship between congestion and emissions compared to their effects on transmission capacity. Despite this, weather conditions remain a valid choice for instrumental variables because they provide exogenous variation in congestion. The use of these weather-related instruments is supported by their demonstrated ability to affect transmission capacity without directly influencing emissions through other channels under normal weather conditions.

to weather conditions at various locations near the transmission system. Therefore, it is possible to identify weather conditions from locations far away from demand centers and wind farms that are correlated with the congestion status and use them as potential instruments. Therefore, after teasing out the effects of renewable generation and demand on emissions, changes in wind speed, wind direction, and air temperatures at these weather stations can only affect emissions by changing the transmission capacity of the transmission system. Finding exact weather conditions that meet these criteria can be challenging without first obtaining geographic data on locational demand and renewable generation. However, we can use a large set of weather conditions data from across Texas as instruments to avoid, or at least alleviate, the potential effects of local weather conditions on local emissions (Woerman 2018).

Following Belloni et al. (2012), we first employ an IV-LASSO (least absolute shrinkage and selection operator) method to choose a set of informative instruments from a large set of variables. These variables include wind conditions and air temperature data collected from more than 200 weather stations in Texas. Furthermore, we include a set of interactions between the weather conditions data from the weather stations with a set of control variables at the ERCOT level and zonal level, including the ERCOT-wide load, the renewable generation at the ERCOT level, and the average regional temperatures. This IV-LASSO procedure aims to identify a subset of weather conditions that are more likely to affect the likelihood of congestion. Figure 2.16 in Appendix E displays a map of all selected weather stations in Texas.

Next, we proceed with a standard two-stage IV estimation using the selected set of instruments. In particular, our excluded instruments include wind speed, wind direction, and air temperatures information from selected weather stations in the IV-LASSO procedure, as well as the selected interactions between weather conditions and the control variables. In the first stage, we utilize the selected variables to predict congestion and the interaction term
between congestion and demand. In the second stage, we employ the fitted values from the first stage as instruments for congestion and the interaction term. We include renewable generation and zonal temperatures as control variables in both stages to ensure that the instruments are valid.

Because we need to control for renewable energy generation to obtain valid IV estimators, the MEFs we obtained from Equation (2.1) will not include the case where renewable energy resources are on the margin.¹⁷ Although wind and solar will not be on the margin systemwide in ERCOT because the system demand is always higher than renewable generation in our sample period, solar and wind can be on the margin in certain areas, as indicated by the overall renewable curtailments. Unfortunately, measuring how this will impact ERCOT-wide MEFs is difficult without obtaining nodal-level data.¹⁸ As a result, when renewable is on the margin, the system-wide MEFs can be lower than we estimated in Equation (2.1).¹⁹

To obtain hourly MEFs, we modify Equation (2.1) by adding hourly dummies to the demand-congestion interactions based on Graff Zivin et al. (2014). Formally, we estimate the following model:

^{17.} Because wind and solar will not be on the margin without congestion, the average MEF when renewable energy resources are on the margin can be captured by the congested MEF ($\beta_1 + \beta_3$) plus the coefficient of wind or solar. Since wind and solar are clean energy resources, the average MEF when wind and solar are on the margin is expected to be zero.

^{18.} Appendix A provides a detailed discussion on this matter and how to estimate the average system-wide MEF using nodal-level LMPs.

^{19.} As a simple illustration, suppose the ERCOT-wide MEF we estimated is 0.6 kg/kWh when the system is congested. This implies that when ERCOT-wide demand increases by 1 kWh, emissions increase by 0.6 kg when a non-renewable generator is on the margin to meet the demand increase. Suppose that the West is congested, and wind is on the margin in the West. The 1 kWh demand increase could partially originate in the West. On average, the likelihood that the demand increase originated in the West can be approximated by the size of the demand in the West. Suppose that there is a 50% chance that the demand increase originated from the West whenever congestion occurs; the renewable-adjusted ERCOT-wide MEF should be 0.3 kg/kWh. This exercise further highlights the importance of regional MEFs.

$$E_{hdmy} = \beta_0 + \sum_{h=0}^{23} \beta_{h1} D_{hdmy} + \beta_2 Congest_{hdmy}$$

$$+ \sum_{h=0}^{23} \beta_{h3} D_{hdmy} \cdot Congest_{hdmy} + \sum_j \theta_j X_{hdmy} + \eta_m + \delta_d + \varepsilon_{hdmy}$$

$$(2.2)$$

Equations (2.1) and (2.2) explore how congestion will affect MEFs in ERCOT. However, congestion can affect regional MEFs differently depending on regional energy resources. In order to understand how transmission congestion affects the locational environmental value of DER technologies, we need to measure the marginal emissions rate at each location and in each hour when congestion arises.

Locational MEFs

Estimating locational MEFs is different from estimating ERCOT-wide MEFs. When the system is uncongested, both locational MEFs and ERCOT-wide MEFs can be estimated based on ERCOT-wide load and emissions. This is because any generation units located in ERCOT can respond to changes in local demand. However, once a region is congested, only local generators can be used to meet the marginal demand. Effectively, the locational MEFs are determined in two regimes: an uncongested regime where MEFs are based on local load and emissions and a congested regime where MEFs are based on local load and emissions. As a result, the endogenous regime switching model is an appropriate solution to estimating locational MEFs.

Regime-switching models have been widely used in the analysis of transmission congestion in electricity markets (Sapio 2015). However, it has not yet been used to answer how congestion will affect marginal emissions. There are two main advantages of using an endogenous regime-switching model in the analysis of congestion. First, compared to traditional econometric models that include congestion indicators as dummy variables, the endogenous regime switching model exhibits better forecasting performances and consistency (Fabra and Reguant 2014; Gianfreda and Grossi 2012). Second, the endogenous regime-switching model uses the instrumental variable approach to describe the switching process, which helps to deal with the endogenous nature of congestion.

In an endogenous regime-switching model, the data generating process is switching between two states, and the switching process is determined by a criterion function (Chang et al. 2017). When the state variable is endogenous, the model is estimated using the instrumental variable approach as described by Maddala (1983). Formally, the criterion function can be given as follows:

$$C_t = \begin{cases} 1 & \text{if } Z_t \eta + u_t > 0 \\ 0 & \text{if } Z_t \eta + u_t \le 0 \end{cases}$$

where $C_t = 1$ when there is congestion (ERCOT-wide or zonal-level), and $C_t = 0$ otherwise. Z_t is a vector of potential instrumental variables used for the congestion indicator. We use the same IV-LASSO procedure to select instrumental variables for each load zone in ERCOT as in the ERCOT-wide case.²⁰

 η is a vector of parameters and u_t is the Gaussian error term. Formally, we estimate the following equations in order to capture the hourly MEFs in each region:

Uncongested:
$$E_t = \beta_1 + \sum_{h=1}^{24} \beta_{2h} H_h D_t + \sum_j \theta_j X_t + \varepsilon_t$$
 if $C_t = 0$ (2.3)

Congested:
$$E_{it} = \beta_3 + \sum_{h=1}^{24} \beta_{4ih} H_h D_{it} + \sum_j \theta_j X_{it} + \varepsilon_{it}$$
 if $C_t = 1$ (2.4)

^{20.} Similar to the previous case, we select the most relevant variables from a large set of variables such that the variables can best "predict" zonal congestion. The variables include weather conditions (local wind speed, local wind direction, and local temperature) for over 200 weather stations in Texas, ERCOT-wide and zonal renewable generation, and ERCOT-wide and zonal load.

where E_t is the hourly emissions from all fossil fuel power generators located in the ERCOT territory. E_{it} is the hourly emissions from all fossil fuel generators from load zone *i*. D_t measures the hourly demand in ERCOT. D_{it} measures the hourly demand in load zone *i*. H_h are dummy variables for hour *h* of the day. X_t is a set of control variables for the month of the year, day or the week, system or zonal-level renewable generation, and system or zonal-level temperatures. The hourly MEFs when the system is uncongested are captured by β_{2h} , whereas the hourly MEFs when the load zone *i* is congested are captured by β_{4ih} .

The endogenous regime switching model described above is estimated using the maximum likelihood method (Lokshin and Sajaia 2004). The Huber-White sandwich estimator is used to obtain heteroskedasticity-consistent standard errors.

Similarly to the ERCOT-wide case, to ensure that the variables are valid instruments, we control for temperatures and renewable generation at each load zone. Therefore, wind conditions should only affect emissions through their impact on transmission congestion. Additionally, by using weather condition data from over 200 weather stations across Texas, we can select weather stations in specific locations as instruments. In particular, because only local temperatures will affect local demand and local emissions, we can use temperatures from locations across Texas as instruments to avoid local effects. Thus, after controlling for local demand and local renewable generation, the instrumental variables will only affect emissions through their impact on congestion.

The above estimating equations reflect the assumption that when a load zone is congested, it becomes an independent grid island. The actual congested area could be any subset of the ERCOT territory, which unfortunately cannot be observed without nodal-level data. As described in Section 2.5.1, we use two methods to measure the congestion status: a system congestion status and a load-zone-specific congestion status. In this case, using the loadzone-specific congestion measure is more appropriate as it more accurately reflects the case when a load zone becomes independent.

Because we have included renewable generation in the estimation of Equations (2.3) and (2.4), the locational MEFs will only capture cases where renewable energy resources are not on the margin. To capture the case when renewable energy resources are on the margin in a load zone, we calculate the average time that a zone has a negative zonal-level LMP. A negative zonal-level LMP indicates that most nodal-level LMPs in the load zone are negative, and renewable energy is on the margin for these nodes. Similarly to the ERCOT-wide case, there are times when wind and solar energy are on the margin in a limited area (covering a few nodes) in a load zone. To properly account for these cases, nodal-level data is required. A discussion on how to use nodal-level LMPs to address these cases is provided in Appendix D. Due to data limitations in our study, we leave this for future work.

2.6 Results

This section presents the locational MEFs identified in this paper. We first present how transmission congestion will affect the ERCOT-wide MEFs. We then focus on three load zones in ERCOT (West, North, and South) to show how congestion will affect the locational marginal emissions.²¹

2.6.1 ERCOT-Wide Marginal Emissions

We first investigate how congestion will affect the MEFs in ERCOT in general. Using our benchmark congestion measure described in section 2.5.1, we estimate Equation (2.1) and present the results in Table 2.2. The coefficient on ERCOT load (β_1) measures the average

^{21.} We exclude the Houston zone for two reasons. First, it is not directly adjacent to the West zone, which is the primary zone we are interested in. Second, the area is relatively small compared to other load zones, and the resulting congestion measure is more likely to be affected by local weather conditions, threatening the validity of our instrumental strategy.

MEFs when the system is uncongested. The average MEFs during congested hours can be obtained by summing β_1 and the interaction term between congestion and load (β_3). Column (1) presents the most common way of estimating MEFs, without considering the effects of congestion. Column (2) presents a simple specification with congestion, which only includes the time controls. Column (3) adds renewable generations as covariates. Column (4) adds linear and polynomial controls for Texas's average heating and cooling degree days.

As shown in Table 2.2 column (1), when congestion is not considered, the average ERCOTwide MEFs is 0.485 kg/kWh. As a comparison, Holland et al. (2022) estimated the average MEFs in Texas in 2019 as 0.454 kg/kWh. The coefficient on load in column (2) is similar to that of column (1), but once the system is congested, the average MEFs will increase by 0.029 kg/kWh. However, as wind generation can affect both total carbon emissions and congestion status, the coefficients in column (2) can be biased. Columns (3) and (4) provide more reliable coefficients by including additional covariates. As shown in the columns (3) and (4), the positive interaction between congestion and load indicates that the marginal emissions are slightly higher in congested hours than in uncongested hours. The MEF measure in column (4) is higher than that in column (3) due to the inclusion of temperature control, as the overall effects of temperature on emissions is negative.²² Using column (4)as our preferred specification, we find that congestion can increase ERCOT-wide MEFs by around 3.1%. This indicates that when the system is congested, more polluting units will be dispatched following the changes in the merit order. In particular, after controlling for both solar and wind generation, changes in MEFs reflect the tendency for more polluting fossil fuel generators to replace relatively cleaner fossil fuel generators during congested hours. The congestion variable is negative and significant, implying that although the MEFs during congested hours are slightly higher, the average emissions factors are similar during both

^{22.} This negative correlation between carbon emissions and temperature, after controlling for demand, congestion, and renewable generation, likely captures the non-linear relationship between temperature (demand) and emissions.

congested and uncongested hours.²³

The results using IV-LASSO are presented in Table 2.3. The IV-LASSO procedure leads to a larger effect of congestion on marginal emissions. In the IV-LASSO setting, the congestion will increase ERCOT-wide MEFs by 14.3% based on column (3), which is our preferred model specification. There are two possible reasons why the IV estimates are larger than the OLS estimates. First, the congestion status measure could be endogenous, leading to biased OLS estimates. This could be due to inadequate controls for wind in the OLS setting, caused by different locational effects of wind.²⁴ Without information on whether the wind has been curtailed, it is difficult to distinguish whether wind or natural gas is on the margin. When the wind speed in a certain region is positively correlated with the probability that the wind is causing the congestion, it will cause a negative bias in the OLS estimator. Using wind speed and temperature at different locations as instrumental variables, we correct the potential bias in the OLS setting.

Second, the higher IV estimates could happen due to the local average treatment effect differing from the overall average effect. This can lead to higher estimates if the variations in congested hours captured by the instruments are prone to hours with higher marginal emissions. For example, this could be the case if higher air temperatures increase the likelihood of congestion, and if higher air temperatures correlate with hours of higher marginal emissions because of renewable curtailment.

Next, we investigate the MEFs for each hour of the day by estimating Equation (2.2)

^{23.} Average emissions factors (AMFs) are calculated by dividing total emissions by total load. Based on the average ERCOT load of 43,000 MWh, the difference in total emissions between congested and uncongested hours, based on column (3) from Table 2.2, is 0.013 * 43,000 - 509.6 = 49.4, or 0.001 kg/kWh.

^{24.} For example, higher wind speeds near wind farms can increase wind generation, thereby reducing ERCOT-wide carbon emissions by displacing fossil fuel generators. However, higher wind generation can also cause more severe transmission congestion when the power flow exceeds transmission capacity. Additionally, higher wind speeds near transmission lines can increase the capacity of those lines, lowering the probability of congestion. In our case, wind speed in certain areas (e.g., west) can be positively correlated with increased MEF during congestion because when the wind, instead of natural gas, is on the margin, congestion would induce more emissions.

| | (1) | (2) | (3) | (4) |
|--------------------------|------------|------------|------------|-----------|
| | Carbon | Carbon | Carbon | Carbon |
| | Emissions | Emissions | Emissions | Emissions |
| | (kg) | (kg) | (kg) | (kg) |
| Congested×Load (kWh) | | 0.029*** | 0.013*** | 0.017*** |
| | | (0.003) | (0.002) | (0.002) |
| ERCOT Load (kWh) | 0.485*** | 0.478*** | 0.497*** | 0.540*** |
| | (0.002) | (0.002) | (0.001) | (0.002) |
| Congestion | | -3522.2*** | -509.6*** | -786.0*** |
| | | (139.2) | (73.5) | (63.7) |
| Solar Generation (kWh) | | | -0.479*** | -0.589*** |
| | | | (0.004) | (0.004) |
| Wind Generation (kWh) | | | -0.597*** | -0.581*** |
| | | | (0.002) | (0.001) |
| Constant | -2062.5*** | -580.7 | 3617.0*** | 2148.7*** |
| | (80.5) | (111.5) | (65.4) | (68.6) |
| Add Time Controls | Yes | Yes | Yes | Yes |
| Add Temperature Controls | No | No | No | Yes |
| Observations | 52,577 | $52,\!577$ | $52,\!577$ | 52,577 |

 Table 2.2: Effect of Congestion on ERCOT-Wide Marginal Emissions - OLS Estimates

Notes: This table presents the OLS estimation results of Equation (2.1). The dependent variable is ERCOT-wide carbon emissions. Columns (1) through (4) include different control variables. Time controls include dummy variables for the month of the year, the hour of the day, and the day of the week. Temperature controls include one- to fourth-degree polynomials of heating and cooling-degree days. Std. Dev. in parentheses* p < 0.05, ** p < 0.01, *** p < 0.001

| | (1) | (2) | (3) |
|-------------------------------------|-------------|------------|------------|
| | Carbon | Carbon | Carbon |
| | Emissions | Emissions | Emissions |
| | (kg) | (kg) | (kg) |
| Congested×Load (kWh) | 0.204*** | 0.033*** | 0.078*** |
| | (0.015) | (0.005) | (0.005) |
| ERCOT Load (kWh) | 0.421*** | 0.513*** | 0.548*** |
| | (0.001) | (0.003) | (0.004) |
| Congested | -18123.2*** | -1158.6*** | -4856.7*** |
| | (655.4) | (230.3) | (242.5) |
| Solar Generation (kWh) | | -0.489*** | -0.512*** |
| | | (0.003) | (0.005) |
| Wind Generation (kWh) | | -0.527*** | -0.452*** |
| | | (0.003) | (0.003) |
| Constant | 7246.6* | 3847.7*** | 3568.5*** |
| | (450.0) | (136.1) | (145.5) |
| Hausman Test (p-value) | 0.0000 | 0.0000 | 0.0000 |
| Kleibergen-Paap rk Wald F statistic | 46.60 | 52.99 | 31.98 |
| Add Time Controls | Yes | Yes | Yes |
| Add Temperature Controls | No | No | Yes |
| Observations | 41,457 | 41,457 | 41,457 |

 Table 2.3: Effect of Congestion on ERCOT-Wide Marginal Emissions - IV Estimates

Notes: This table presents the second stage of the 2SLS estimation results of Equation (2.1). The dependent variable is ERCOT-wide carbon emissions. Columns (1) through (4) include different control variables. Time controls include dummy variables for the month of the year, the hour of the day, and the day of the week. Temperature controls include one-to fourth-degree polynomials of heating and cooling-degree days. The instruments include weather conditions (wind speed, wind direction, and temperature) from weather stations across Texas and are selected using a LASSO regression. The Stock-Yogo weak identification cut-off at 10% is 10.90. The Kleibergen- Paap rk Wald F statistics suggest that the instruments are not weak. The Hausman test results reject the null that the OLS estimations are efficient. Std. dev. in parentheses* p < 0.05, ** p < 0.01, *** p < 0.001

using our IV-LASSO specification in column (3). The results are shown in Figure 2.6. The MEFs during congested hours and uncongested hours with 95% confidence intervals are presented by blue and red colours, respectively. The "System-Wide MEFs" shown in the green line presented MEFs estimations without considering congestion.²⁵ Compared to Tables 2.2 and 2.3, the hourly MEFs estimated using Equation (2.2) shows a 9.4% difference between congested and uncongested hours. The results suggest that, on average, more polluting sources will be dispatched after congestion. The results highlight the importance of including transmission congestion in estimating MEFs.



Figure 2.6: Hourly ERCOT-Wide MEFs during Uncongested and Congested Hours

^{25.} Because we have controlled for renewable generations in the estimation of MEFs, this typical "System-Wide MEFs" estimated do not represent the "typical" MEFs reported by other studies. The MEFs in other studies are used to "predict" environmental impacts for load changes that occur at a particular time of the day and therefore do not include control variables. In our study, we must control for renewable generations in order to have valid instruments for congestion. As a result, we include renewable generations in estimating all MEFs to keep them comparable.

2.6.2 Locational Marginal Emissions

In this section, we use the endogenous regime-switching model to estimate the locational marginal emissions in three load zones (North, South, West) in ERCOT. As described in Section 2.2, the location of DER technology can affect carbon emissions through two channels: a direct channel based on locational marginal emissions and an indirect channel based on the shadow environmental value of congestion.²⁶ Suppose marginal changes in load have minimal impact on congestion status, the locational environmental value of a DER technology will be determined by the locational marginal emissions are estimated in this section.

We first examine how congestion affects the MEFs in the North zone. Figure 2.7 presents the average hourly MEFs of the North zone with 95% confidence intervals during congested and uncongested hours. The green line shows the weighted-average congested MEFs accounting for the hours when wind is on the margin. In particular, the MEFs are weighted by the percentage of hours when renewable energy resources are on the margin.²⁷ The average MEFs during uncongested and congested hours are 0.579 kg/kWh and 0.543 kg/kWh, respectively. On average, congestion can reduce the MEFs in the North zone by 6.2% as the natural gas generators in the North is relatively cleaner than the ERCOT average. Wind is only on the margin less than 2% of the time, and can reduce the average North MEFs by 0.08 kg/kWh.

^{26.} The shadow environmental value of a transmission constraint represents the marginal change in total emissions resulting from relieving the transmission constraint by one MW. A less constrained electric grid can reduce renewable curtailments and often leads to fewer carbon emissions. Since DER technologies can be used to alleviate transmission congestion, they can indirectly affect carbon emissions.

^{27.} For example, during a given hour, if wind power is on the margin with a 5% probability when the load zone is congested, the weighted-average MEFs for that hour will be 0.95 times the estimated congested MEFs.



Figure 2.7: MEFs in the NORTH Zone During Uncongested and Congested Hours

As shown in Figure 2.8, the South zone MEFs differ from the North zone as the fossil fuel generators in the South that are more likely to be called upon for generation tend to have higher emissions intensities. In particular, coal generators are more likely to be on the margin in the South, as indicated by the relatively higher MEFs in the mornings. The average MEFs during uncongested and congested hours are 0.599 kg/kWh and 0.736 kg/kWh in the South zone, respectively. This means congestion increases average South MEFs by 22.6%. Similar to the North, wind energy only has a small effect on average MEFs, which reduces average South MEFs by 0.15 kg/kWh.



Figure 2.8: MEFs in the South Zone During Uncongested and Congested Hours

The renewable-rich West zone is frequently congested. As shown in Figure 2.9, congestion significantly impacts the MEFs in the West zone. The average MEFs during uncongested hours is 0.588 kg/kWh compared to 0.393 kg/kWh during congested hours. There are two reasons for the low MEFs in the West. First, wind energy resources are deployed on the margin during congestion. In our sample period, negative and zero LMPs (when the wind energy is on the margin) in the West occur in 7.0% of all hours and 15.1% of all congested hours. Second, cleaner natural gas units in the West may substitute dirtier technology in the other regions. The results suggest that congestion can significantly impact DER technologies without considering the impact of congestion will lead to substantial biases.



Figure 2.9: MEFs in the West Zone During Uncongested and Congested Hours

2.7 Locational Environmental Values

After estimating the locational MEFs, we can calculate the corresponding locational environmental values of solar and storage. The locational MEFs provided by Section 2.5.2 are estimated under two separate regimes: congested and uncongested. We calculate the associated environmental values of the DERs using the following formula:

$$\Delta CO_2 = \sum_m \sum_h \sum_c \Delta Load_{mh} Congest_{mhc} MEF_{hc}$$
(2.5)

where m, h, and c stand for the month of the year, hour of the day, and congestion status, respectively. In particular, we consider three congestion status in our study: uncongested, congested with positive zonal LMPs, and congested and with zero or negative zonal LMPs. As mentioned in the previous section, a zero or negative zonal LMP indicates that renewable energy is on the margin and the MEF is zero. $\Delta Load_{mh}$ is the load changes resulting from solar or battery for a given month-hour. $Congest_{mh}$ is the percentage of times of a congestion status for a given month-hour. MEF_{hc} is the corresponding MEF of hour h under congestion status c. In other words, the environmental values of DERs represent the changes in CO₂ emissions from both regimes (congested and uncongested), weighted by the likelihood of each regime for a given month-hour. It is important to note that by disregarding the monthly variations in load, the above formula can be simplified as:

$$\Delta CO_2 = \sum_h \Delta Load_h \sum_m \sum_c Congest_{mhc} MEF_{hc}$$
(2.6)

where $\sum_{m} \sum_{c} Congest_{mhc} MEF_{hc}$ represents the congestion-weighted MEFs for each monthhour. Comparing the congestion-weighted MEFs with typical MEFs provides a convenient way for assessing the impact of congestion on MEFs.

In addition to the congestion-weighted MEFs, the DER technologies' technological features and operational patterns can also affect their environmental outcomes. Without loss of generality, we calculate the annual carbon mitigation for a typical 5 kWh solar system and a 13.5 kWh battery. The capacity reflects the average solar and storage installation capacity for small residential and commercial customers (Energy Sage 2023a). We assume that the solar panel has an energy efficiency of 15%, which means that a solar panel can convert 15% of solar power into electricity (Energy Sage 2023a). Based on the monthly average solar irradiation in three major Texas cities (Houston, San Antonio and Dallas), a 5 kWh solar system can generate around 30 kWh in a typical day. We assume that the battery has a round-trip efficiency of 80%, which means that 80% of the power stored in the batteries can be retrieved for discharge.

We assume that the battery is operated to maximize profits by performing energy arbitrage. For simplicity, the battery operations are based on a typical residential time-of-use rate used in Texas, with a peak period between 1 PM and 7 PM on weekdays (Xcel Energy 2023). Under such a retail rate structure, the battery will be fully charged before 1 PM and discharged as soon as 1 PM to minimize energy losses. Solar will reduce carbon emissions based on how much power it generates in each hour of the day and what is the corresponding MEFs. A battery will increase emissions in hours during charging, and decrease emissions in hours during discharging, based on the MEFs during charging and discharging. We use the Distributed Energy Resources Customer Adoption Model (DER-CAM) to simulate the annual carbon emissions decrease (or increase) from using solar and battery. Detailed information on DER-CAM is available in Appendix B from the first chapter.

Based on the locational marginal emissions estimated in Section 2.6, we calculated the congestion-weighted MEFs for the three load zones in ERCOT. The results are presented in Figure 2.10. The average congestion-weighted MEFs for the North, South, and West zones are 0.561, 0.659, and 0.500 kg/kWh, respectively. In comparison, the average System-Wide MEF is 0.586 kg/kWh, which is estimated without considering congestion. These differences can be decomposed into two factors.

First, fossil fuel generators exhibit different emissions intensities in different regions, and can have different probabilities of being utilized to supply marginal demand. For instance, the higher-than-average MEFs in the South zone can be partly attributed to the higher percentages of coal-fired generations in the South, which increases the probability of coalfired plants being used on the margin during the early mornings when demand is low.²⁸ In addition, natural gas generators in the West and North zones are relatively cleaner compared to those in the South.

Second, in zones with high renewable penetration, the probability of renewable energy resources (mainly wind energy in the West zone) being utilized marginally are higher. For

^{28.} Appendix C presents the emissions intensities for all fossil fuel generators in each load zone.

instance, during early mornings and late nights with stronger wind and lower demand, wind appears on the margin for around 10% of all hours, explaining the lower MEFs in the West zone during these hours. The West zone's MEFs are 15.6% lower than the typical MEFs, with 11.4% attributable to renewable energy resources and 4.2% to relatively cleaner natural gas generators. In contrast, fossil fuel generators are responsible for the different MEFs in the north and South zones, as renewable energy resources are used marginally for less than 1% and 2% of the time, respectively.



Figure 2.10: Congestion-Weighted MEFs for Each Load Zone

How do the estimated MEFs translate to the environmental values of DER technologies? Table 2.4 reports the annual carbon mitigation (kg of CO_2) of solar and batteries located in different regions of ERCOT as compared to the typical case. The results are calculated based on Equation (2.5).

As shown in Table 2.4, the environmental outcomes of a DER technology in the North

| | Solar | Battery Storage |
|---------|--------|-----------------|
| Typical | -170.7 | 25.3 |
| West | -147.0 | 22.6 |
| South | -195.7 | 28.2 |
| North | -165.0 | 25.2 |

 Table 2.4:
 Annual Carbon Mitigation of DER Technologies

zone do not differ substantially from using the typical MEFs. However, for a DER technology located in the renewable-rich West zone, including congestion in the estimations of MEFs can have a substantial impact. Installing a solar panel in the West zone will only achieve 86.4% of the annual CO₂ mitigation compared to the typical case. This is because the MEFs during solar generating hours are significantly lower in the West, since the West is frequently congested and the MEFs are low during congested hours. Excess solar energy will be curtailed or will increase wind curtailments in some congested hours where wind energy is used on the margin. In contrast, solar located in the South can mitigate 14.1% more carbon emissions compared to the typical case, as solar energy is used to replace dirtier fossil fuel generators in the South.

Unlike solar, the environmental values of batteries are determined by the MEFs during charging and discharging. With a carefully designed time-varying electricity rate (e.g., timeof-use rate), the battery can charge with cleaner energy and use the stored energy to displace a dirtier generation source later. Therefore, the difference in MEFs between charging and discharging or the variance of MEFs determines the potential of carbon mitigation. In the West zone, a battery can charge when the West zone is congested and wind energy is on the margin. As a result, deploying a battery in the West can often yield better environmental outcomes than in other regions. However, in our settings, deploying batteries in the West will still increase carbon emissions due to energy losses, but the increase in carbon emissions is 10. 7% less than in the typical case.

As discussed in Section 2.2, installing DER technologies can help alleviate regional congestion and indirectly impact the environment. For example, installing a battery system in the west could reduce the inflow congestion to the West zone, whereas installing a battery system elsewhere could reduce the outflow congestion in the west. However, due to limitations in our methodology, we leave this indirect environmental impact for future research.

2.8 Conclusion

Transmission congestion has become an important barrier to the growth and integration of renewable energy resources, especially wind. In jurisdictions such as Texas, where wind energy resources are located in remote areas, wind energy resources are frequently curtailed due to insufficient transmission capacity. Frequent congestion in renewable-rich regions can have a significant impact on locational MEFs, which determines the environmental values of the DER technologies.

In our paper, we analytically highlight the importance of including transmission congestion in the estimation of locational MEFs, especially for regions with high penetration of wind energy resources. In particular, we provide a novel way of estimating MEFs considering congestion using an endogenous regime switching model. When the system is uncongested, we assume that all the generation resources located in the ERCOT territory can respond to load changes. However, when a particular load zone is congested, we assume that only local generation resources will be able to respond to load changes. Compared to previous estimations using strict assumptions, our MEFs estimations can lead to more reliable estimates for environmental values of DER technologies.

Based on our MEFs, we find that installing solar in a renewable-rich regions that are frequently congested tend to have less carbon mitigation as compared to other regions. For example, a solar panel located in the West zone in ERCOT will only bring about 70% of carbon mitigation compared to the other load zones. In comparison, installing battery storage in renewable-rich region can often reduce storage-induced carbon emissions as MEFs in these regions often have greater variances. With an appropriate retail tariff designs, a battery can charge using clean energy generated from wind (which might be curtailed) when the region is congested, and discharges to replace a dirtier marginal technology when the system is uncongested.

Our results have important policy implications. Most compensation policies for DER technologies do not take into account locational effects. A solar or battery will receive the same compensation even though the underlying environmental values may vary significantly (Sexton et al., 2021). Our study highlights the importance of using location-specific subsidies based on the locational environmental values.

Our study has several main limitations. First, our study assumes assumes that DER technologies have a negligible effect on congestion. However, DER technologies can be used to alleviate congestion and could have an indirect impact on environment. Without understanding how DER technologies affect congestion, and without understanding how DER technologies affect the underlying merit order based on a dispatch model, our study do not take into consideration this indirect effect.

Second, the environmental values of DER technologies are calculated based on average MEFs. In reality, the MEFs can vary significantly in real-time. For example, MEFs can drop to zero when wind energy is used in the margin. Therefore, the variance of MEFs could be more substantial and we might underestimate the role of battery storage in carbon mitigation. We leave this to future studies when real-time marginal emissions data becomes available.

Third, due to the limitation of the regime-switching model in our study, we do not differ-

entiate in-flow congestion with out-flow congestion. Our MEFs during congested hours is defined as a weighted-average measure for the two types of congestion. As different congestions can have different implications for the marginal emissions, future work should differentiate different types of congestion by defining multiple regimes.

References

- Belloni, A., D. Chen, V. Chernozhukov, and C. Hansen. 2012. "Sparse Models and Methods for Optimal Instruments With an Application to Eminent Domain." *Econometrica* 80 (6): 2369–2429. ISSN: 0012-9682, accessed June 10, 2024. https://doi.org/ 10.3982/ECTA9626. http://doi.wiley.com/10.3982/ECTA9626.
- Bowen, Thomas, Sam Koebrich, Kevin McCabe, and Ben Sigrin. 2022. "The locational value of distributed energy resources: A parcel-level evaluation of solar and wind potential in New York state." *Energy Policy* 166 (July): 112744. ISSN: 03014215, accessed June 10, 2024. https://doi.org/10.1016/j.enpol.2021.112744. https://linkinghub.elsevier.com/retrieve/pii/S0301421521006108.
- Brown, David P., and David E. M. Sappington. 2017a. "Optimal policies to promote efficient distributed generation of electricity." *Journal of Regulatory Economics* 52, no. 2 (October): 159–188. ISSN: 0922-680X, 1573-0468, accessed June 10, 2024. https://doi.org/10.1007/s11149-017-9335-9. http://link.springer.com/10.1007/s11149-017-9335-9.
- Brown, David P., and David E.M. Sappington. 2017b. "Designing Compensation for Distributed Solar Generation: Is Net Metering Ever Optimal?" *The Energy Journal* 38, no. 3 (May): 1–32. ISSN: 0195-6574, 1944-9089, accessed June 2, 2024. https://doi.org/ 10.5547/01956574.38.3.dbro. https://journals.sagepub.com/doi/10.5547/01956574.38. 3.dbro.
- Brown, David P., Jay Zarnikau, and Chi-Keung Woo. 2020. "Does locational marginal pricing impact generation investment location decisions? An analysis of Texas's wholesale electricity market." *Journal of Regulatory Economics* 58, no. 2 (December): 99–140. ISSN: 0922-680X, 1573-0468, accessed June 10, 2024. https://doi.org/10.1007/s11149-020-09413-0. https://link.springer.com/10.1007/s11149-020-09413-0.
- **BTU Analytics.** 2023. "Wind and Solar Curtailments in ERCOT." Accessed April 30, 2023. https://btuanalytics.com/power-and-renewables/wind-and-solar-curtailments-in-ercot/.
- CAISO. 2023a. 2022 Annual Report On Market Issues and Performance.
- Callaway, Duncan S., Meredith Fowlie, and Gavin McCormick. 2018. "Location, Location, Location: The Variable Value of Renewable Energy and Demand-Side Efficiency Resources." Journal of the Association of Environmental and Resource Economists 5, no. 1 (January): 39–75. ISSN: 2333-5955, 2333-5963, accessed June 2, 2024. https://doi.org/10.1086/694179. https://www.journals.uchicago.edu/doi/10.1086/694179.
- Chang, Yoosoon, Yongok Choi, and Joon Y. Park. 2017. "A new approach to model regime switching." Journal of Econometrics 196, no. 1 (January): 127–143. ISSN: 03044076, accessed June 10, 2024. https://doi.org/10.1016/j.jeconom.2016.09.005. https://linkinghub.elsevier.com/retrieve/pii/S0304407616301671.

- Deniel, Joe, and Sam Gomberg. 2021. Why Does Wind Energy Get Wasted? The Myth of Wind Energy Oversupply and the True Root Causes of Wind Curtailment. Cambridge, MA (United States): Union of Concerned Scientists. https://www.ourenergypolicy.org/ resources/why-does-wind-energy-get-wasted-the-myth-of-wind-energy-oversupplyand-the-true-root-causes-of-wind-curtailment/.
- **EIA.** 2023. *EIA-860 Suevey*. Accessed May 31, 2023. https://www.eia.gov/electricity/data/eia860/.
- Energy Sage. 2023a. "Buying solar panels in 2023: The complete guide." Accessed August 31, 2023. https://www.energysage.com/solar/.
- Energy Sage. 2023b. "How much do storage systems cost in Texas in 2023?" Accessed May 19, 2023. https://www.energysage.com/local-data/energy-storage-cost/tx/.
- **EPA.** 2023a. Continuous Emissions Monitoring System. Accessed May 5, 2023. https://www.epa.gov/emc/emc-continuous-emission-monitoring-systems.
- **EPA.** 2024. Inventory of U.S. Greenhouse Gas Emissions and Sinks 1990-2022. Washington, D.C. (United States): United States Environmental Protection Agency.
- **ERCOT.** 2014. The Competitive Renewable Energy Zones Process.
- ERCOT. 2022. ERCOT Fuel Mix Report. Accessed May 5, 2024. https://www.ercot.com/ gridinfo/generation.
- ERCOT. 2023a. Fuel Mix Report. Accessed May 5, 2023. https://www.ercot.com/gridinfo/generation.
- ERCOT. 2023b. Report on Existing and Potential Electric System Constraints and Needs.
- ERCOT. 2024. Hourly Load Data Archives. Accessed May 31, 2024.
- Fabra, Natalia, and Mar Reguant. 2014. "Pass-Through of Emissions Costs in Electricity Markets." American Economic Review 104, no. 9 (September 1, 2014): 2872–2899. ISSN: 0002-8282, accessed June 2, 2024. https://doi.org/10.1257/aer.104.9.2872. https: //pubs.aeaweb.org/doi/10.1257/aer.104.9.2872.
- Fell, Harrison, Daniel T. Kaffine, and Kevin Novan. 2021. "Emissions, Transmission, and the Environmental Value of Renewable Energy." American Economic Journal: Economic Policy 13, no. 2 (May 1, 2021): 241–272. ISSN: 1945-7731, 1945-774X, accessed June 10, 2024. https://doi.org/10.1257/pol.20190258. https://pubs.aeaweb.org/doi/10. 1257/pol.20190258.
- Fisher, Michael J., and Jay Apt. 2017. "Emissions and Economics of Behind-the-Meter Electricity Storage." *Environmental Science & Technology* 51, no. 3 (February 7, 2017): 1094–1101. ISSN: 0013-936X, 1520-5851, accessed June 5, 2024. https://doi.org/10. 1021/acs.est.6b03536. https://pubs.acs.org/doi/10.1021/acs.est.6b03536.

- Gentle, Jake, Kurt S Myers, Thomas Baldwin, Isaac West, Kenyon Hart, Bruce Savage, Mike Ellis, and Phil Anderson. 2012. "Concurrent Wind Cooling in Power Transmission Lines."
- Gianfreda, Angelica, and Luigi Grossi. 2012. "Forecasting Italian electricity zonal prices with exogenous variables." *Energy Economics* 34, no. 6 (November): 2228–2239. ISSN: 01409883, accessed June 10, 2024. https://doi.org/10.1016/j.eneco.2012.06.024. https://linkinghub.elsevier.com/retrieve/pii/S0140988312001338.
- Graff Zivin, Joshua S., Matthew J. Kotchen, and Erin T. Mansur. 2014. "Spatial and temporal heterogeneity of marginal emissions: Implications for electric cars and other electricity-shifting policies." *Journal of Economic Behavior & Organization* 107 (November): 248–268. ISSN: 01672681, accessed June 5, 2024. https://doi.org/10.1016/ j.jebo.2014.03.010. https://linkinghub.elsevier.com/retrieve/pii/S0167268114000808.
- Hitaj, Claudia. 2015. "Location matters: The impact of renewable power on transmission congestion and emissions." *Energy Policy* 86 (November): 1–16. ISSN: 03014215, accessed June 10, 2024. https://doi.org/10.1016/j.enpol.2015.06.019. https://linkinghub.elsevier. com/retrieve/pii/S0301421515002372.
- Hochul, Kathy, and Basil Seggos. 2023. Establishing a Value of Carbon. New York, NY (United States): New York State Department of Environmental Conservation.
- Holland, Stephen P., Matthew J. Kotchen, Erin T. Mansur, and Andrew J. Yates. 2022. "Why marginal CO₂ emissions are not decreasing for US electricity: Estimates and implications for climate policy." *Proceedings of the National Academy of Sciences* 119, no. 8 (February 22, 2022): e2116632119. ISSN: 0027-8424, 1091-6490, accessed June 10, 2024. https://doi.org/10.1073/pnas.2116632119. https://pnas.org/doi/full/10.1073/ pnas.2116632119.
- Holland, Stephen P., Erin T. Mansur, Nicholas Z. Muller, and Andrew J. Yates. 2016. "Are There Environmental Benefits from Driving Electric Vehicles? The Importance of Local Factors." *American Economic Review* 106, no. 12 (December 1, 2016): 3700–3729. ISSN: 0002-8282, accessed June 10, 2024. https://doi.org/10.1257/aer. 20150897. https://pubs.aeaweb.org/doi/10.1257/aer.20150897.
- Joskow, Paul L. 2020. "Transmission Capacity Expansion Is Needed to Decarbonize the Electricity Sector Efficiently." *Joule* 4, no. 1 (January): 1–3. ISSN: 25424351, accessed June 10, 2024. https://doi.org/10.1016/j.joule.2019.10.011. https://linkinghub.elsevier. com/retrieve/pii/S2542435119305276.

- Li, Mo, Timothy M. Smith, Yi Yang, and Elizabeth J. Wilson. 2017. "Marginal Emission Factors Considering Renewables: A Case Study of the U.S. Midcontinent Independent System Operator (MISO) System." *Environmental Science & Technology* 51, no. 19 (October 3, 2017): 11215–11223. ISSN: 0013-936X, 1520-5851, accessed June 5, 2024. https://doi.org/10.1021/acs.est.7b00034. https://pubs.acs.org/doi/10.1021/acs. est.7b00034.
- Lokshin, Michael, and Zurab Sajaia. 2004. "Maximum Likelihood Estimation of Endogenous Switching Regression Models." The Stata Journal: Promoting communications on statistics and Stata 4, no. 3 (August): 282–289. ISSN: 1536-867X, 1536-8734, accessed June 10, 2024. https://doi.org/10.1177/1536867X0400400306. http://journals.sagepub.com/doi/10.1177/1536867X0400400306.
- Maddala, Gangadharrao S. 1983. Limited-dependent and qualitative variables in econometrics. Cambridge University Press.
- Minnesota Department of Commerce. 2014. Minnesota Value of Solar: Methodology. Minnesota Department of Commerce, Division of Energy Resources. https://mn.gov/ commerce/energy/images/MN-VOS-Methodology-FINAL.pdf.
- NCCETC. 2023. 50 States of Solar: 2023 Quarterly Report. Raleigh, NC (United States): North Carolina Clean Energy Technology Center. https://nccleantech.ncsu.edu/2023/ 04/19/the-50-states-of-solar-states-move-toward-time-varying-or-avoided-costcompensation-for-distributed-generation-in-q1-2023/.
- NOAA. 2023. Environmental Information's Integrated Surface Dataset (ISD). Accessed June 30, 2023. https://www.ncei.noaa.gov/access/metadata/landing-page/bin/iso?id=gov.noaa.ncdc:C00532.
- NREL. 2014. Commercial and Residential Hourly Load Profiles for all TMY3 Locations in the United States. Accessed April 30, 2023. https://catalog.data.gov/dataset/ commercial-and-residential-hourly-load-profiles-for-all-tmy3-locations-in-the-unitedstate-bbc75.
- Phillips, Tyler Bennett. 2014. "Dynamic Rating of Transmission Lines for Improved Wind Energy Integration in Complex Terrain." PhD diss., Boise State University.
- **Potomac Economics.** 2023a. 2021 State of the Market Report for the ERCOT Electricity Markets.
- **Potomac Economics.** 2023b. 2022 State of the Market Report for the ERCOT Electricity Markets.
- **Potomac Economics.** 2024. 2023 State of the Market Report for the ERCOT Electricity Markets.
- Rhodes, Joshua, Thomas Deetjen, and Caitlin Smith. 2021. Impacts of Large, Flexible Data Center Operations on the Future of ERCOT. Lancium Balancing Energy.

- Sapio, Alessandro. 2015. "The effects of renewables in space and time: A regime switching model of the Italian power price." *Energy Policy* 85 (October): 487–499. ISSN: 03014215, accessed June 10, 2024. https://doi.org/10.1016/j.enpol.2015.07.025. https:// linkinghub.elsevier.com/retrieve/pii/S0301421515300367.
- **SEIA.** 2022. Solar Market Insight Report 2021 Year in Review. Washington, D.C. (United States): Solar Energy Industries Association.
- Sexton, Steven, A. Justin Kirkpatrick, Robert Harris, and Nicholas Muller. 2018. Heterogeneous Environmental and Grid Benefits from Rooftop Solar and the Costs of Inefficient Siting Decisions. w25241. Cambridge, MA: National Bureau of Economic Research, November. Accessed June 10, 2024. https://doi.org/10.3386/w25241. http: //www.nber.org/papers/w25241.pdf.
- Siler-Evans, Kyle, Inês Lima Azevedo, and M. Granger Morgan. 2012. "Marginal Emissions Factors for the U.S. Electricity System." *Environmental Science & Technology* 46, no. 9 (May 1, 2012): 4742–4748. ISSN: 0013-936X, 1520-5851, accessed June 6, 2024. https://doi.org/10.1021/es300145v. https://pubs.acs.org/doi/10.1021/es300145v.
- Stoll, Christian, Lena Klaaßen, Ulrich Gallersdörfer, and Alexander Neumüller. 2023. *Climate Impacts of Bitcoin Mining in the U.S.* MIT Center for Energy and Environmental Policy Research.
- Woerman, Matt. 2018. "Market Size and Market Power: Evidence from the Texas Electricity Market." *Energy Institute Working Paper.*
- Wooldridge, Jeffrey M. 2010. Econometric analysis of cross section and panel data. MIT press.
- Xcel Energy. 2023. "Time of Use Rates." Accessed June 1, 2023. https://tx.my.xcelenergy. com/s/business/rate-plans/time-of-use.

2.9 Appendices

Appendix A: Generation Mix

Figure 2.11 and 2.12 show the share of generation capacities and total generations for different technologies for each load zone in 2022. The West Zone is rich in wind energy resources, whereas the coal and natural gas generators form the backbone of the generation fleet in the North and South.



Figure 2.11: Generation Capacities (MW) by Technology



Figure 2.12: Electricity Generations (MWh) by Technology

Appendix B: Emission Intensities for Fossil Fuel Generators by Load Zone

Figures 2.13-2.15 plot the emission intensities for fossil fuel generators in the West, North, and South load zones, respectively. The emissions intensity for a plant is calculated by dividing the total annual emissions (in kg) of the plant by the total electricity generation (in kWh) in 2022. All data are from the EIA-923 form (EIA, 2023).

Compared to the North and South, the natural gas plants in the West are relatively cleaner. For an average West demand of less than 5,000 MWh, around 2,000 MWh of natural gas generation capacities have emissions intensities lower than 0.5 kg/kWh. The North and South have similar generation mixes. However, given that the South load (10,694 MW) is much lower than the North (16,123 MW), cheaper coal-fired generators are called upon more often in the South than in the North. Additionally, the average emissions intensity is slightly higher in the South, which means that, on average, a South plant is more polluting than a plant in the North.

It is worth noting that the generators are called upon based on their marginal cost (in a competitive market) rather than emissions factors. Without calculating the marginal costs for each generator unit and ranking the generators accordingly, it is difficult to assume which generators are more likely to be deployed on the margin. Due to the scope of our work, we leave this to the future.



Figure 2.13: Emission Intensities for West Generators



Figure 2.14: Emission Intensities for North Generators



Figure 2.15: Emission Intensities for South Generators

Appendix C: Regional Average MEFs (Congested vs. Uncongested)

Table 2.5 reports the regional MEFs using the regime-switching model without hourly interactions. The regional load and ERCOT load presents the MEFs during congested and uncongested hours, respectively. The results are similar to the main results described in section 2.6.

| | (1) | (2) | (3) |
|--------------------------|-----------------|-----------------|----------------|
| | North Emissions | South Emissions | West Emissions |
| | (kg) | (kg) | (kg) |
| Congested | | | |
| Regional Load (kWh) | 0.561*** | 0.699*** | 0.445*** |
| | (0.003) | (0.003) | (0.007) |
| Solar Generation (kWh) | -0.138*** | -0.041*** | -0.013*** |
| | (0.004) | (0.004) | (0.002) |
| Wind Generation (kWh) | -3.478 | -1.881*** | -0.170*** |
| | (0.026) | (0.003) | (0.001) |
| Unongested | | | |
| ERCOT Load (kWh) | 0.603*** | 0.599*** | 0.619*** |
| | (0.002) | (0.001) | (0.002) |
| Solar Generation (kWh) | -0.438*** | -0.463*** | -0.448*** |
| | (0.005) | (0.003) | (0.003) |
| Wind Generation (kWh) | -0.468*** | -0.391*** | -0.431*** |
| | (0.002) | (0.006) | (0.006) |
| Add Time Controls | Yes | Yes | Yes |
| Add Temperature Controls | Yes | Yes | Yes |
| Observations | 52,577 | 52,577 | 41,457 |

 Table 2.5: Regional Average MEFs (Congested vs. Uncongested)

Notes: This table presents the average regional MEFs estimated by the regime switching model. Columns (1) through (3) measures the regional MEFs for the North, South, and West load zone, respectively. Time controls include dummy variables for the month of the year, the hour of the day, and the day of the week. Temperature controls include one to fourth-degree polynomials of heating and cooling-degree days. Std. dev. in parentheses* p < 0.05, ** p < 0.01, *** p < 0.001

Appendix D: Estimate MEFs using Nodal LMPs

MEF measures the average increase in total emissions when the total electricity demand is increased by 1 MW. Ideally, MEFs should be measured in real time for each node in an electricity grid. However, due to data limits, MEFs are often measured in an aggregate manner over times and locations. For example, an ERCOT-wide MEF will capture the average increase in emissions if ERCOT-wide demand is increased by 1 MW. Likewise, a locational (regional) MEF measure will capture the average increase in emissions if demand increases by 1 MW in a particular region, such as a load zone.

However, when congestion occurs, the estimation for MEFs becomes more complicated because the increase in emissions resulting from a 1 MW increase in demand will differ based on where that demand increase comes from. For example, if the demand increase happens in a wind-congested area, it should have a zero MEF. If the demand increase happens in an uncongested area, the MEF will be determined by the emissions intensity of the marginal units, most likely a natural gas plant. As a result, the average MEF in the West should be a weighted average of the two MEFs, weighted by the likelihood of where the next demand increase will happen. With access to nodal-level LMPs, the average MEFs can be represented as the weighted average MEFs for each LMP node, weighted by the corresponding electricity demand for each node.

Suppose in a perfect world where only two technologies can be on the margin: natural gas units, all with the same emission intensities, and renewable energy resources (wind) with zero emission intensity. At any time t, the regional MEF can be represented as:

$$MEF_t = \frac{d_{t,ng}}{D_t} MEF_{ng} + \frac{d_{t,wd}}{D_t} MEF_{wd}$$
(2.7)

which reflects the average increase in emissions resulting from a 1 MW increase in demand. The weight $\left(\frac{d_{t,ng}}{D_t}\right)$ measures the percentage of the current load with natural gas units on the margin. For example, if 100% of the current load has natural gas units on the margin, then the current regional MEF should be equal to MEF_{ng} , which is the MEF we estimated using Equation (2.1) Note that wind could still generate electricity when this happens, as long as they are not used on the margin.

Based on Equation (2.7), the average regional MEF across time is:

$$MEF = \frac{1}{t} \sum_{t} \frac{d_{t,ng}}{D_t} MEF_{ng}$$
(2.8)

Note that the second term from Equation (2.7) is zero because the emissions intensity of wind is zero. Basically, Equation (2.8) states that the average regional MEF should be equal to the emission intensity of the marginal natural gas units, weighted by the likelihood that a natural gas unit is on the margin at any time t, averaged over time.

If we have nodal-level data, such as demand, renewable generation, and emissions in each node. We can estimate the MEF for each node when natural gas is on the margin (which indicates nodal-level LMP>0). The nodal-level MEF can then be calculated as:

$$MEF_{i} = \frac{t_{i,ng}}{t}MEF_{i,ng} + \frac{t_{i,wd}}{t}MEF_{i,wd}$$
(2.9)

which is the weighted average MEF based on the percentage of time when a natural gas unit is on the margin. To calculate the regional MEF, we need to consider the probability for a demand increase to occur in a node, which is implied by the demand size of that node. Formally, the MEF measures in a region can be represented as the following:

$$MEF = \sum_{i} \frac{d_i}{D} MEF_i = \sum_{i} \frac{d_i}{D} \frac{t_{i,ng}}{t} MEF_{i,ng}$$
(2.10)

Suppose $MEF_{i,ng}$ is the same across nodes, the regional MEF can be represented as:

$$MEF = MEF_{ng} \sum_{i} \frac{d_i}{D} \frac{t_{i,ng}}{t}$$
(2.11)

which provides an estimate for the average MEFs in different regions. As the environmental values of a DER technology is often based on some average MEF estimates, Equation (2.11)

provides a simple way of calculating this average MEF, without accessing real-time emissions data.

Appendix E: Weather Stations Selected by the IV-LASSO Procedure

Figure 2.16 plots the locations of selected weather stations identified using the IV-LASSO procedure. If at least one variable (wind speed, wind direction, or air temperature) from a weather station is selected by IV-LASSO to predict ERCOT-wide congestion, it is shown on the map. As illustrated in Figure 2.16, the weather stations are distributed across ERCOT, highlighting the complexity of the ERCOT transmission system.



Figure 2.16: Selected Weather Stations

Appendix F: Percentage of Congested Hours by Year

Figure 2.17 plots the average time of congestion in ERCOT by year. Both transmission expansion and additional renewable generation affect congestion in any given year. In 2017,



Figure 2.17: Percentage of Congested Hours by Year

the system was congested about 26% of the time. Due to increased penetration of solar and wind, the percentage of congested hours increased to around 40% between 2018 and 2021. By 2022, the percentage of congested hours was increased to more than 50%. This figure highlights the importance of considering congestion in estimating MEFs.
Chapter 3

Competing Perspectives of Maximum Demand Charges

3.1 Introduction

Economic theory suggests that the price of a good is efficient when it reflects the marginal cost of the good. However, for "natural monopolies," setting prices at marginal cost generally does not generate sufficient revenues to cover the total costs of production. As such, it is generally accepted that the regulation of these natural monopolies should aim at a second-best outcome: minimizing dead-weight loss while ensuring financial solvency (Joskow 2007). However, in reality, marginal prices in some natural monopoly industries are well above the marginal costs, leading to outcomes worse than the second-best scenario (Biggar 2009). One of the prominent examples is the electricity market, where some utilities set prices much higher than marginal costs, leading to substantial dead-weight losses (Borenstein and Bushnell 2022). This is because utilities tend to use volumetric (\$ per kWh) charges to recover the fixed and variable costs for generating and delivering electricity, which distorts the marginal cost signal for electricity consumption (Wood et al. 2016).

Finding an efficient and equitable way to recover the large fixed costs faced by electric utilities is a long-standing question in the electricity sector (Borenstein 2016). In particular, the costs of maintaining a reliable grid are expected to increase in the future. The increase stems from the rising demand for expanding electric network infrastructure, driven by the increased electrification from electric vehicles and electrified heat pumps (Elmallah et al. 2022). Consequently, there is an increased urgency for more efficient tariff designs to mitigate the need for expensive network expansion. In particular, maximum demand charges (MDCs) that charge customers based on their maximum demand (in kW or kVA) over a certain period have gained significant attention from regulators and utilities. As such, whether MDCs can provide efficient and fair fixed cost recovery and whether MDCs should play a larger role in the recent evolution of tariff designs have become topics of considerable debate.

MDCs have been used for commercial and industrial customers for several decades, serving as a tool to manage maximum demand and ensure that customers who place higher demands on the grid pay their fair share of costs (Berg and Savvides 1983; Neufeld 1987). By charging MDCs, utilities are less reliant on volumetric charges to recover the fixed costs of network infrastructure, enabling them to have more efficient marginal prices. In addition, it has been argued that MDCs provide a price signal to customers to reduce the maximum demand, which can ultimately lead to a lower system maximum demand (at various levels of the grid network) and lower investment costs for additional capacities for the generation, transmission, and distribution of the electric grid (Wood et al. 2016).

Traditionally, residential customers have rarely been subjected to MDCs (Hledik 2014). However, it has been argued that MDCs could play a crucial role in accommodating increased electrification and mitigating the need for future expansion of distribution networks, similar to their use in managing system maximum demand for commercial and industrial customers (Elmallah et al. 2022).

For MDC to effectively reduce grid investment costs, customers must be sensitive to the

price signal, and the timing of the price signal should coincide with the system maximum demand (Brown and Sappington 2018). In particular, since most MDCs target customerspecific maximum demand rather than system maximum demand, their effectiveness in reducing system maximum demand is frequently questioned (Brown and Sappington 2018). In addition, only a few studies have empirically examined the demand responses from MDCs, and these studies are often subject to the unique characteristics of MDC designs (e.g., Caves et al. 1984; Taylor and Schwarz 1986; 1990; Stokke et al. 2010). As a result, more evidence is needed to assess the effectiveness of MDCs in terms of maximum demand reductions.

Concern for fairness is another aspect of the ongoing debate about MDCs. As argued by Wood et al. (2016) and Hledik (2014), MDCs can better align electricity bill charges with the cost structure of the utilities so that customers who impose higher costs on the grid will pay more. However, Borenstein (2016) argues that MDCs based on a customer's individual maximum demand are not a good approximation for grid costs because a customer's individual maximum demand does not determine the overall need for grid capacity.

The concern for fairness is particularly important in the case of residential MDCs. With the rapid expansion of rooftop solar panels, many utilities have implemented residential MDCs to address the cross-subsidy concern from non-solar to solar customers, and many more are exploring the option (Faruqui et al. 2020).¹

Although MDCs have become increasingly popular among some utilities, the overall effectiveness of such tariffs is still under debate. Motivated by this recent debate on MDCs, we conduct a comprehensive literature review on whether MDCs can achieve efficient and fair cost recovery for utilities. Although the recent debate primarily centers on residential MDCs, it is essential to consider that residential MDCs share common objectives and features with existing MDCs for commercial and industrial customers. Therefore, incorporating evidence

^{1.} Section 3.2.2 provides a detailed analysis of this issue.

from all types of customers is crucial to providing a comprehensive understanding.

A central issue in this debate is whether MDCs can effectively motivate "peak shaving," thereby reducing maximum demand and alleviating the need for expensive grid expansions. To address this, we first review the empirical evidence related to MDC demand responses for all types of customers. Despite the growing body of research on residential MDCs, there remains a significant gap in the literature concerning the response of large commercial and industrial customers to MDCs in recent years. These customers remain the primary targets for MDCs and can provide valuable empirical insights for retail tariff designs concerning MDCs for all types of customers. To fill this gap, we conduct a detailed case study utilizing data from large commercial and industrial customers to shed light on customer responses to MDCs.

Our case study utilizes a unique meter data set for 126 large commercial and industrial customers in Lethbridge, Alberta.² We focus on large customers in Alberta because, unlike other jurisdictions, large commercial and industrial customers³ account for more than 70% of the electricity demand in Alberta, and play a critical role in determining the system peak (MSA 2014; AESO 2023). The MDCs in our case study are charged daily based on the highest electricity demand in a 12-month rolling period, providing customers with a strong price signal to reduce maximum demand. Compared to residential customers, the large customers in our dataset consume significantly more energy and can achieve substantial gains from responding to price signals, making them ideal candidates for demand response.

To perform the case study, we leverage an observed 8% decrease in the MDC rate level in Lethbridge from 2017 to 2018 to investigate the customers' demand response. To understand how customers respond to the changed price signal, we use machine learning techniques to

^{2.} Lethbridge has a 100% coverage of smart meters and mandatory MDCs for all but small residential customers.

^{3.} Consumers are considered large if their annual electricity consumption exceeded 250 MWh. For more information, see MSA (2014).

predict counterfactual demand for each customer, assuming that the MDC rate does not change. In particular, we use a series of climate, socioeconomic, and time variables to train customer-specific prediction models using data from 2017. We then use the "trained models" to predict customers' daily maximum demand in 2018, effectively holding the MDC level unchanged. The changes in maximum demand will be captured by changes in prediction errors between 2017 and 2018 during the period when annual maximum demand is likely to occur.

Contrary to previous studies focusing on "peak reduction", we investigate whether customers would increase maximum demand when faced with a lower MDC rate. For mediumlarge customers, we observed a 4% increase in electricity demand (as indicated by prediction errors) from 2017 to 2018, during the days that the annual maximum demand is most likely to occur. This implies a statistically significant but moderate change in daily maximum demand from 2017 to 2018. For large customers, we find no statistically significant evidence of a response.

These findings could be attributed to a potential asymmetry in demand responses to increases versus decreases in MDC rates. When faced with increasing MDC rate, customers might make significant investments in energy conservation or demand-shifting technologies, which could become sunk costs. On the other hand, responses to decreased MDC rates might be limited to behavioral adjustments, which tend to have a smaller impact.⁴ This potential asymmetry highlights the complexity of customer behavior in response to MDC rate adjustments and suggests a need for more nuanced tariff designs. Such designs could benefit from accounting for both the long-term investments driven by MDCs and the more immediate behavioral changes associated with changed MDC rates.

^{4.} For instance, with a lower MDC rate, customers might adjust their maximum demand through changes in cooling or heating needs, which represents a behavioral change. Conversely, investments in more energyefficient devices or technologies in response to higher MDC rates involve sunk costs and represent a long-term commitment that may not fluctuate with short-term MDC rate changes.

Our paper contributes to the literature in three ways. First, we provide a comprehensive literature review on the role and implications of MDCs, incorporating perspectives from both proponents and critics. Second, our discussion of MDCs contributes to the ongoing debate on retail tariff designs, emphasizing the evidence needed to assess whether MDCs can facilitate fair and efficient cost recovery for utilities. Third, our empirical case study adds to the existing literature on demand responses from MDCs. Specifically, our study addresses a gap in recent empirical research by focusing on large commercial and industrial customers, which complements existing studies that primarily target smaller customers.

This chapter proceeds as follows. Section 3.2 reviews the related literature on the debate of MDCs. Section 3.3 provides a review of the literature on empirical evidence for MDC responses. Section 3.4 details the background and data of our study. The empirical methodology of the case study is presented in Section 3.5. Section 3.6 presents the main results of our case study. Section 3.7 concludes.

3.2 Review of studies of MDC

The electric utility is a classic example of a "natural monopoly" as it faces large fixed costs: setting the electricity rate at the short-run marginal cost (SRMC) will not generate enough revenue to cover its total costs (Baumol and Willig 1981; Joskow and Schmalensee 1983). Utilities have employed several options to raise additional revenues, such as increasing the average volumetric charge (\$ per kWh), implementing MDCs, increasing fixed charges, etc. For large customers, MDC is one of the main ways for utility cost recovery and usually represents 30% to 70% of a customer's monthly electricity bill (NREL 2022).

For smaller customers, MDCs have rarely been implemented partially due to meter technology barriers, but the rapid distribution of smart meters in recent decades has largely eliminated this barrier (Hledik 2014). As of 2015, at least 16 utilities in the US have implemented MDCs for residential customers (Hledik and Faruqui 2016). By the end of 2019, the number of active residential MDC plans increased to 68, with at least ten mandatory plans (Faruqui et al. 2020). Residential MDCs in these utilities account for 10% to 82% of electricity bills for residential customers (Hledik 2014).

Whether MDCs can provide efficient and fair fixed-cost recovery has become a topic of considerable debate in the literature.⁵ This section will review the arguments of both proponents and opponents of MDCs, focusing on three distinct but related topics:

- Can MDCs reduce system peak demand?
- Can MDCs achieve cost-causation by aligning customer bill charges to costs they impose on the grid?
- What are the distributional effects of MDCs?

3.2.1 System Peak Reduction

The potential benefits of MDCs for motivating "peak reduction" and promoting a smooth consumption pattern have been recognized for decades (Henderson 1983). Berg and Savvides (1983) attributes some of the early interest in MDCs to their potential effects of deferring capacity investments in the electricity grid. For MDCs to reduce system peak demand, MDCs need to provide price signals that correlate with the system peak demand, and customers need to be price-responsive.

^{5.} The concept of "fairness" is subject to multiple interpretations. In this chapter, we use "cost-causation" to refer to the idea that customers who impose higher costs on the grid should pay more. We use "distributional effects" to reflect the idea that low-income customers should have lower electricity bills to improve energy affordability for low-income households. These ideas reflect the basic principles of who should pay for a public good among customers (Borenstein et al. 2021).

Until recently, most MDCs were designed to target a customer's individual maximum demand, with only a small portion of MDCs targeting the system maximum demand (Hledik 2014). This customer-specific MDC design can be traced back to the early stages of electrification and was appealing at that time due to the limitations of the metering technology⁶, the need for utilities to avoid grid bypass⁷, and the high correlation between individual maximum demand and system maximum demand during that period (Neufeld 1987). However, this design raises concern as customer-specific maximum demand may not correlate with the system maximum demand (Neufeld 1987; Hledik 2014; Brown and Sappington 2018). In response, system peak coincident MDCs have been implemented more frequently recently for residential MDCs (Faruqui et al. 2020). However, non-coincident MDCs are still the dominant form of MDCs in recent years, especially for large commercial and industrial customers (Passey et al. 2017). According to Passey et al. (2017), some utilities are hesitant to replace non-coincident MDCs with system peak coincident MDCs due to concerns that some customers may not be aware of the system peak or may not fully understand the term "coincident", potentially leading to unpleasant bill shock for these customers.

Existing studies have found that customer-specific MDCs are inefficient in reducing system maximum demand, even if customers are price-responsive. In particular, large industrial customers that are most likely to face MDCs tend to have the least coincident individual maximum demand (Neufeld 1987). For example, only 8% of customer-specific maximum demand reduction for industrial customers can be transformed into system peak reduction as the individual maximum demand does not usually coincide with the system maximum demand (Henderson 1983).⁸

^{6.} The early meters (around 1910) can only record aggregate consumption and maximum consumption, but cannot record when maximum consumption occurs (Neufeld 1987).

^{7.} A major concern for utilities at the early stages of electrification is that some large customers choose to generate electricity as an isolated plant and bypass the grid (Neufeld 1987). The costs of a customer to set up its own isolated plant are closely related to its maximum demand. Therefore, charging customers based on their individual maximum demand provides a way for utilities to exercise price discrimination (Neufeld 1987).

^{8.} See Appendix E for a detailed discussion on how customer-specific maximum demand in our sample

Whether customers respond to the MDC signals is crucial for peak reduction, even if the price signal is correlated with the system's maximum demand. Previous studies have found significant responses from both industrial, commercial, and residential customers toward MDCs. A more detailed review of these studies will be provided in Section 3.3.

In an environment with increasing penetration of solar panels and battery storage technologies, how MDCs can be used to reduce system maximum demand and achieve bill savings utilizing these new technologies has become increasingly important. A number of studies have investigated the role of solar and batteries in reducing maximum demand and MDCs for customers. Darghouth et al. (2017) investigated the effectiveness of the use of solar panels to reduce residential MDCs. They found that solar panels are generally inefficient in reducing MDCs for residential customers because their maximum demand period typically occurs in the evenings when solar cannot generate much power. Solar panels are more effective at reducing MDCs if the MDCs are based on the system maximum demand that correlates with the solar generating hours. However, with increasing penetration of rooftop solar, the system's maximum demand is likely to be shifted away from the solar generating hours in the near future.⁹

Compared to solar, batteries are more effective in reducing MDCs. McLaren et al. (2017) found that reducing MDCs is the main driver of battery installation, and the potential of bill savings brought about by batteries under MDCs is substantial. Boampong and Brown (2020) investigated the impact of MDCs on battery operation decisions and how they affect the utilities' avoided energy and capacity costs. They found that while batteries can reduce utilities' avoided costs under MDCs, MDCs based on the system's maximum demand can induce much higher cost reduction than non-coincident MDCs.

correlate with the system maximum demand in Alberta. Because more than 80% of the system demand is from large industrial and commercial customers in Alberta, the correlation between the customer-specific load and the system load can be higher in our sample.

^{9.} This has already happened in regions with high solar penetration, as represented by the classic "duck curve" in California and other jurisdictions.

Other retail tariff designs have been used to reduce the system's maximum demand. Examples include time-of-use (TOU) pricing, where electricity rates change according to peak/off-peak periods; real-time pricing (RTP), where electricity rates change in real time based on wholesale electricity prices; and critical peak pricing (CPP), where electricity prices are higher during a small number of critical days or hours (Faruqui et al. 2009). When smart meter technologies that can record energy consumption in real time are not available, TOU rates have often been discussed as an alternative to MDCs (e.g., Seeto 1997; Woo et al. 2002). A potential benefit of MDCs over TOU is that they provide incentives to reduce needle peaks, where the electricity demand surges in a short period (Seeto 1997). With better meters available, it has been shown that RTP and CPP bundled with demand charges have been shown to be more effective than TOU in terms of peak reduction (Faruqui and George 2005; Faruqui et al. 2009). For example, in several pilot studies conducted among residential customers, CPP induces an average of 13-20% of system peak reduction, RTP induces an average system peak reduction of 10-14%, whereas TOU only achieves 3-6% of the system peak reduction (Faruqui et al. 2009; Faruqui and Sergici 2010; Dutta and Mitra 2017). However, these studies have not isolated the unique impact of MDCs from CPP and RTP on peak demand reduction.

In recent years, the surge in residential electrification, driven by the adoption of electric vehicles (EVs) and electric heat pumps, has drawn significant attention to the need to upgrade local distribution networks (Elmallah et al. 2022). Several studies have looked at the role of MDCs in reducing the maximum demand associated with the distribution networks (reducing local distribution peaks). For example, Abdelmotteleb et al. (2018) proposes to use a peak-coincident MDC to recover part of the distribution charges during critical peak hours of the distribution network. Turk et al. (n.d.) proposes a subscription charges based on a customer's ex-ante maximum demand subscription to recover part of the distribution charges. These studies aimed to provide incentives for customers to reduce maximum demand with respect to

local distribution network, therefore, reducing the need to upgrade the distribution network.

Recognizing this potential benefit of MDCs in reducing peak demand, the effectiveness of different MDC designs is an important empirical question. For example, how would different types of customers (residential, commercial, and industrial) respond to system peak coincident MDCs compared to non-coincident MDCs? Is there any evidence to support the claims that customers would not fully comprehend the term "coincident" when designing MDCs? How efficient are system peak coincident MDCs compared to non-coincident MDCs at reducing system total avoided costs? How efficient are MDCs based on the monthly maximum demand compared to MDCs based on annual maximum demand? Due to the heterogeneity and complexity of MDC designs, more studies and evidence are needed to quantify the economic outcomes associated with each MDC design.

How the MDC rate should be determined is another important question to consider when designing efficient MDCs. Currently, there is no general rule on how MDC rates are determined. In some jurisdictions, MDCs are used to recover the full or a portion of fixed distribution and transmission charges (Brown et al. 2015). However, for the efficient pricing of maximum demand reduction, the system peak coincident MDC rate should reflect the total avoided costs of system maximum demand. More broadly, the non-coincident MDC rate for a customer should reflect the system's long-run marginal costs for serving this customer.¹⁰ Quantifying the inefficiencies associated with different MDC rate levels in different jurisdictions can be another important empirical question to answer in the future.¹¹

The most recent empirical studies on MDCs focus on residential customers (e.g., Stokke et al. 2010; Bartusch et al. 2011; Bartusch and Alvehag 2014; Öhrlund et al. 2019). However,

^{10.} It is worth noting that the primary efficiency goal of MDC in some jurisdictions is to motivate the efficient use of the distribution network (Passey et al. 2017). For this purpose, the MDC rate should be set at the long-run marginal cost (LRMC) of the distribution network (Brown et al. 2015). More details will be provided in Section 3.2.2.

^{11.} For example, one aspect to consider is how the price signal provided by the MDC could distort the price signals provided by other time-varying tariff designs such as RTP.

large commercial and industrial customers continue to be the main targets of MDCs and offer great potential for demand response. The last empirical study on MDCs for these large customers was conducted 40 years ago. Therefore, new studies are needed to account for advancements in empirical methodologies, potential changes in customer preferences and behaviors, and the evolution of underlying technologies installed at end-user facilities. In addition, recent developments in metering technologies, especially high-frequency data from smart meters, have enabled more advanced methods to be used to investigate the demand response towards MDCs. Considering the importance of large customers in terms of demand response and their relatively outdated designs of customer-specific MDCs, future studies should evaluate the role of MDCs and the demand response of large consumers to MDCs.

3.2.2 Cost-Causation

Proponents argue that MDCs provide better alignment of electricity tariffs and utility network costs. This can lead to better price signals for efficient use of the electric network (Hledik 2014; Passey et al. 2017; Brown et al. 2015). The reason is that network costs are often closely related to maximum demand at various levels. For example, a customer's distribution service cost is determined in part by the maximum demand of this customer; a neighborhood's need for an electricity distribution substation is determined by the collective maximum demand of the neighborhood; and the need for additional generation capacity is determined by the maximum demand of the system (Hledik 2014). Therefore, MDCs can reasonably approximate the fixed costs a customer imposes on the grid.

However, as argued by Borenstein (2016), a customer's individual maximum demand can only capture a portion of the costs associated with the grid's need for generation, transmission, and distribution capacity because the individual maximum demand may not coincide with the system maximum demand. In addition, the customer-specific distribution service costs are largely made up of sunk investment costs determined up-front by the attributes of the distribution service instead of the maximum demand of the customer (Borenstein 2016).¹² Economic theory suggests that the distribution tariff should provide price signals based on its long-run marginal costs¹³ and recovering sunk costs with MDCs will distort this price signal (Brown et al. 2015). Therefore, it is more efficient to recover these sunk costs with fixed charges based on the customer's service capacity (Borenstein 2016). However, as argued by Passey et al. (2017), customers belonging to the same customer class (i.e., residential) tend to have distribution services with the same technical attribute. Allocating sunk costs equally to all customers within a customer class can lead to distributional concerns, as low-income customers would have to pay the same as high-income customers. More discussion on distributional effects is available in Section 3.2.3.

Empirical evidence on how cost-reflective the MDCs are is scarce in the literature. To the best of our knowledge, Passey et al. (2017) is the only study examining how a customer's MDCs reflect the long-run marginal cost of providing network service to this specific customer. Passey et al. (2017) use a customer's contribution towards the highest system demand in a year to proxy the long-run marginal cost of serving this customer. Based on a sample of over 2,000 Australian households and a local MDC tariff, they found that non-coincident MDCs have a low correlation coefficient (r = 0.56) with the long-run marginal cost. In comparison, a system peak-coincident MDC with a similar design can raise the correlation coefficient to 0.9.

While the study of Passey et al. (2017) provides insight on how MDCs correlate with long-term marginal costs, its long-run marginal cost measure only captures part of the costs that correlate with the system peak. Given the complexity of the energy grid, the long-run

^{12.} According to Hanser (2013), about 80% of the distribution service costs are sunk investment costs. However, due to increased electrification (e.g., electric vehicles, electric heat pump), this is expected to change in the near future.

^{13.} Long-run marginal costs refer to the investment expenses needed to meet a marginal increase in the maximum capacity of an electric network.

marginal cost will also be affected by the customer's location on the electric network (Brown et al. 2015). Future studies should investigate the locational cost-causation of the MDCs. In addition, the correlation analysis of Passey et al. (2017) does not capture how the price signals provided by the MDCs reflect the level of long-run marginal costs. Achieving economic efficiency requires setting the MDC rate at the long-run marginal cost (Brown et al. 2015). However, MDCs from the real world usually deviate from the long-run marginal costs,¹⁴ and future studies should quantify the inefficiencies associated with this price distortion.

3.2.3 Distributional Effects

The distributional effects of tariff designs are a crucial factor that can impede retail rate reform for residential customers (Joskow and Wolfram 2012). In the US, over 80% of utility requests to increase fixed charges are denied by regulators over concerns about the potential distributional effects on low-income customers (Trabish 2018). For residential customers, fixed charges only account for a small portion of their electricity bills, whereas volumetric charges typically account for over 90% (Hledik 2014). Under this tariff structure, customers who use less electricity will face substantially lower electricity bills than customers who use electricity extensively. If a utility uses fixed charges to recover fixed costs, the electricity bills of small customers will increase substantially (Hledik and Faruqui 2016). However, with MDCs, this redistribution effect between small and large customers is much smaller, as smaller customers with a lower maximum demand are not automatically worse off (Hledik 2014; Wood et al. 2016). As evidence suggests that low-income customers tend to be lowusage customers¹⁵, have a flatter consumption pattern¹⁶, and have higher demand response

^{14.} Utilities tend to recover network costs through volumetric charges, especially for residential customers, due to distributional effects concern (Brown et al. 2015). Even if residential customers face MDCs, the MDC rate tends to be low (Brown et al. 2015). For large industrial and commercial customers, the MDC rate can also deviate from the long-run marginal costs as MDCs can be used to recover some of the sunk costs. 15. For example, see Bird et al. (2015), Lazar (2016), Wood et al. (2016), and Zethmayr and Makhija

^{(2019). (2019).}

^{16.} For example, see Lazar (2016) and Zethmayr and Makhija (2019).

than high-income customers¹⁷, MDCs are often favored by utilities and regulators concerned with social fairness (Wood et al. 2016).

However, when better energy conservation and self-generating technologies become more accessible, smaller customers (who purchase fewer kWh from the utilities) may not imply low-income customers (Hledik 2014). In particular, energy efficiency devices and solar panels are more affordable for high-income customers, which reduces the need to purchase electricity directly from the utilities. In addition, some low-income customers can have a limited ability to shift load (Wood et al. 2016). To address these concerns, a few studies use both billing and income data to simulate the distributional effects of various tariff designs. In particular, Hledik (2014) simulated the effects of MDCs on customer electricity bills compared to a tariff design with high fixed charges. He finds that low-income customers are more likely to benefit than the average customer under MDCs. In particular, around 75% of low-income customers are better off under the MDCs compared with fixed charges. To quantify the distributional effects of MDCs compared to the current two-part tariff design (small fixed charges plus significant volumetric charges), Hledik and Greenstein (2016) collected a sample of over 2,000 residential customers in Vermont, US. They find that, unlike fixed charges, MDCs do not disproportionately affect low-income customers, and the regressive distributional effects of MDCs can be minimized by developing a careful transition strategy, including education, optional rate exemptions, or other social programs.

Reducing cross-subsidy concerns among customers, especially residential customers, is another significant benefit proposed by the proponents of MDCs (Hledik 2014). In particular, in an environment with rapid expansion of rooftop solar panels, there is a significant crosssubsidy concern between solar and non-solar customers (Brown and Sappington 2018). This is because most utilities rely on volumetric (per kWh) charges to recover fixed costs, so the per kWh volumetric rate is often higher than the avoided energy costs for the utilities.

^{17.} For example, see Wolak (2010), Gyamfi et al. (2013), and Burger et al. (2020).

As solar customers can self-generate part of the electricity and do not purchase from the utility, the lost revenues outweigh the avoided costs, forcing utilities to raise the volumetric rate for all customers. In addition, solar customers are often allowed to export excess power generated from solar panels back to the grid and be compensated at the prevailing volumetric rate. This further pressured utilities' financial solvency, forcing them to increase volumetric charges. To eliminate this cross-subsidy concern, the electricity tariff should be aligned with the cost structure of the utilities: the volumetric charge should reflect the avoided energy costs, and fixed charges should be recovered through alternative tariff designs, such as fixed charges or MDCs. Hledik (2014) argued that recovering fixed costs through fixed charges can raise distributional concerns, whereas MDCs help protect small customers and provide opportunities for customers to reduce electricity bills.

However, as advocated by Borenstein et al. (2021), an income-based fixed charge can address the distributional concern of higher fixed charges for low-income families. Income-based fixed charges would assign different levels of fixed charges based on households' ability to pay. By setting the volumetric rate at the SRMC and recovering fixed costs through incomebased fixed charges, utilities can achieve both efficiency and equity. Furthermore, a utility can design income-based fixed charges in a way that ensures that low-income households do not suffer, therefore ensuring a progressive distribution among households (Borenstein et al. 2021).

Current studies that look at the distributional effects of MDCs generally compare between MDCs and non-income-based fixed charges or between MDCs and volumetric charges.¹⁸ How income-based fixed charges should be designed and what the distributional effect it has compared to MDCs have not been empirically examined in the literature. Future studies are needed to answer these questions.

^{18.} For example, see Hledik (2014), Rubin (2015), Hledik and Greenstein (2016), and Lazar (2016).

3.3 Review of Empirical Studies on MDC Response

In theory, MDCs provide incentives for customers to reduce peak demand in order to cut electricity bills (Taylor and Schwarz 1986). In some utilities, MDCs represent up to 80% of a customer's monthly electricity bill (Hledik 2014). But to what level are customers willing to exploit the opportunity? As summarized in Table 3.1, only a handful of studies have attempted to answer this question.

Early empirical studies (e.g., Henderson 1983; Mountain and Hsiao 1986) tend to focus on large commercial and industrial consumers as they are the primary targets of MDCs. Henderson (1983) estimates the MDC elasticity of peak demand using load data at the utility level. Using cross-sectional data from 89 privately owned utilities, Henderson (1983) estimated that the MDC elasticity is -0.78 and -0.61 for industrial and commercial loads, respectively. Mountain and Hsiao (1986) and Veal (1986) conducted an analysis based on large industrial firms in Ontario, Canada. Mountain and Hsiao (1986) measure the substitution between monthly maximum demand and off-peak demand in response to changes in price structure (demand charge and energy charge). They find that about half of the firms in their sample show significant responses toward MDCs by reducing maximum demand. Veal (1986) uses the extreme value likelihood function to estimate how the maximum demand of industrial firms changes with different levels of MDC and find a small MDC response for 3 of 5 cement firms.

Caves et al. (1984) and Taylor and Schwarz (1986) are the first to measure demand response for residential customers. They conducted experiments by randomly assigning households to a time-of-use (TOU)¹⁹ plus demand charge group and a flat rate control group. They find positive evidence for peak reduction under the MDCs. In addition, both Caves

^{19.} Under a TOU rate, a customer will pay according to the time of day and day of the week he uses electricity.

| Studies | Regions | Tvne ^a | Period | Dat | a | Response |
|--|--------------------------|-------------------|---------------|-------------------|-----------|------------------|
| | 0 | - | | Structure | Level | J |
| Henderson (1983) | NS | C&I | 1970 | Cross-sectional | Utility | Significant |
| Caves et al. (1984) | Wisconsin (US) | R | 1977-1979 | Panel | Household | Significant |
| Mountain and Hsiao (1986) | Ontario (Canada) | Ι | 1970-1980 | Time-series | Firm | Some Significant |
| Veal (1986) | Ontario (Canada) | Ι | 1970-1977 | Time-series | Firm | Some Significant |
| Taylor and Schwarz (1986) | North Carolina (US) | R | 1978-1983 | Panel | Household | Significant |
| Taylor and Schwarz (1990) | North Carolina (US) | R | 1985-1988 | Panel | Household | Significant |
| Stokke et al. (2010) | Norway | R | 2006 | Panel | Household | Significant |
| Bartusch et al. (2011) | Sweden | R | 2005 - 2008 | Panel | Household | |
| Bartusch and Alvehag (2014) | Sweden | R | 2006-2012 | Panel | Household | |
| Öhrlund et al. (2019) | Sweden | C&R | 2015-2017 | Panel | Consumer | Significant |
| ^a C, I, and R stand for com | mercial, industrial, and | d resider | ntial custome | rs, respectively. | | |

| Resp | |
|-----------------------|--|
| Charge | |
| Demand | |
| of I | |
| sudies | |
| $\tilde{\mathbf{v}}$ | |
| mpirical | |
| Ē | |
| of | |
| Review | |
| •• | |
| | |
| 3 | |
| Table | |

et al. (1984) and Taylor and Schwarz (1986) found that customers had a higher tendency to shift maximum demand (highest demand in a month) to off-peak periods compared to peak periods.

Taylor and Schwarz (1990) conducted a follow-up study and estimated the long-run effects of TOU rates with demand charges. They find that MDCs reduce both maximum demand and demand in peak periods. Moreover, they found a surprising result: MDCs induce a greater reduction in demand in peak periods than in the maximum demand itself. This highlights the possibility that customers are not aware when their demand peaks, especially when smart meters are not accessible. In addition, they find that the customer response towards MDCs is increasing over time.

These studies with controlled experimental designs provide valuable insight into customer response towards MDCs, but they were conducted more than 30 years ago. Therefore, the results do not account for changes in consumer behavior caused by shifts in preferences, changes in business electric equipment or household appliances, and changes in economic conditions (Hledik 2014). For instance, customers from the early studies did not have smart meters, which means that customers did not know their current electricity usage or when their demand peaked. This could explain why customers tend to shift both the maximum and peak periods demand to off-peak periods.

Compared to previous studies (e.g., Caves et al. 1984; Taylor and Schwarz 1986; 1990), more recent studies often lack an experimental design and therefore must rely on econometric techniques to infer causality. Stokke et al. (2010) analyzed the demand response of 443 Norwegian households to a winter-only demand charge in 2006. They estimate a fixedeffects panel model based on hourly meter data. They use hourly dummy variables to capture the changes in hourly energy demand when MDCs are active, and control for factors like temperature, wind speed, and daylight hours. They find that there is an average 5% reduction in maximum demand due to MDCs, corresponding to a 0.7% reduction in total load.

Bartusch et al. (2011) and Bartusch and Alvehag (2014) use descriptive analysis to study the effects of demand charges on 500 households in Sweden. Based on a descriptive comparison of households before and after the introduction of the MDCs, they found that singlehouse consumers (residential consumers who live in a single house) had reduced maximum demand in response to the MDCs by 10% to 22% during the summer. However, during winter months, the maximum demand change exhibits much higher variations, from a 3% increase to a 22% decrease. In addition, they found that single-house consumers had shifted consumption from peak to off-peak hours by about 2.5% during the summer, but only 0.25% during the winter. However, due to its descriptive nature, it is difficult to establish a causal relationship between the introduction of MDCs and the reduction in electricity maximum demand.

Öhrlund et al. (2019) conducted the most sophisticated and recent study on MDC responses using billing data for around 50,000 small customers (but only 612 with MDCs). He used a propensity score method to create a sample of 1,110 customers with an MDC and a control group. Using a customer-specific two-level time series model, Öhrlund et al. (2019) found that the MDC group has lowered maximum demand by 7.4% compared with the control group since the MDCs were introduced. However, Öhrlund et al. (2019) did not find evidence that customers with stronger financial incentives and can benefit more from demand response (higher potential bill savings) will respond more to the MDCs. This finding suggests that customers could respond to price signals for non-financial reasons (Öhrlund et al. 2019). The MDC treatment in Öhrlund et al. (2019) is limited to a particular region in Sweden. While this feature allows Öhrlund et al. (2019) to define a control group using regions proximate to the treatment region, it may prove challenging to replicate this approach in other jurisdictions. It is worth noting that there are no studies to the best of my knowledge that investigate large commercial and industrial customers' responses to MDC since the late 1980s. Considering that large customers possess great potential for demand responses and continue to be the main recipients of MDCs, it is vital to provide an updated study on how they will respond to MDCs. Our study aims to fill this gap.

In addition, unlike previous studies, our study looks at how customers would respond to changes in the level of MDCs, rather than to the existence of MDC. Therefore, our study provides insight into how financial incentives of MDCs, rather than the perceptions of MDC itself, can affect customer decisions. Previous studies have claimed that the mere presence of the MDC can induce demand response due to non-financial reasons such as social responsibilities (Darby and McKenna 2012). In addition, there is limited evidence suggesting that customers will respond to the financial incentives provided by MDCs (Öhrlund et al. 2019). Our study aimed to provide direct evidence on how customers will respond to the financial incentives provided by the MDC.

Similarly to the study of Stokke et al. (2010), our study relies on pre-treatment vs. posttreatment comparisons to estimate customer demand response toward MDCs. However, our data and methodology differ from Stokke et al. (2010) in two ways. First, Stokke et al. (2010) examines the data for only one year, with the pre-treatment period occurring during summer and the post-treatment period during winter. The unique environmental conditions in Norway can lead to significantly different consumption patterns between seasons, which could be difficult to control using a fixed-effects model. In comparison, the pre-treatment vs. post-treatment comparison of our study occurs within the same season, and our model accounts for other factors, including temperature and economic conditions. Second, by using a machine learning method to create a counterfactual demand profile for each customer, our study can better account for heterogeneity across customers and can provide more robust results than a fixed effects model.

3.4 Background and Data of the Case Study

The previous section summarizes the primary motives of MDCs in three main aspects. As mentioned, a crucial motivation for MDCs is to incentivize maximum demand reductions, thereby reducing electric grid upgrade costs.²⁰ The remainder of the study will focus on this aspect by providing an empirical analysis that fills an important gap in the literature.

3.4.1 Case Study Objective

As summarized in Section 3.3, previous studies have generally found that MDCs can reduce maximum demand. However, recent studies have focused mainly on residential and small commercial customers. The main objective of our case study is to provide an updated examination of demand response from large commercial and industrial customers. Estimating demand response from large customers is interesting for three reasons. First, large commercial and industrial customers have been the primary targets of MDCs for decades, and MDCs continue to be an important feature in electricity charges faced by large customers (Hledik and Faruqui 2016). In particular, it is estimated that almost all medium and large customers are subject to MDCs or have the option to select electricity plans with an MDC feature (McLaren et al. 2017).

Second, commercial and industrial customers, compared to residential customers, present greater potentials for demand response in the electric grid. In general, large customers can derive greater benefits from responding to MDCs and tend to have better means to adjust their demand. For instance, large customers are more likely to have smart meters and may install battery technologies, employing battery operation algorithms to mitigate MDCs (McLaren et al. 2017). Therefore, large commercial and industrial customers could be more

^{20.} Although how customer-specific electricity demand correlates with the system demand is not the main objective of this paper, we provide a detailed summary in Appendix E.

price responsive toward MDCs than residential customers. Given that large customers can play an important role in terms of maximum demand reduction, it is crucial to estimate the demand response for these customers and design more effective MDCs accordingly.

Third, estimating demand response from large customers can offer insight into the maximum response to demand from smaller customers. For example, if large customers demonstrate limited demand response, it is reasonable to assume that residential customers will also exhibit limited demand response, considering that larger customers have greater incentives to respond.²¹ In addition, conducting pilot MDC studies for residential customers can be costly or infeasible in some jurisdictions. In such cases, estimating the demand response for large customers can be more feasible as the data is readily available. In summary, considering the significance of large customers in the context of MDCs and the scarcity of related studies in recent years, it is crucial to estimate the demand response for these customers. Our study aims to fill this gap.

Apart from the goal of estimating demand response toward MDCs from large customers, our study differs from previous studies in two unique ways. First, while previous studies have focused on the existence of MDCs, our case study examines changes in the level of MDCs. Notably, previous studies have emphasized the significance of the mere presence of MDCs in terms of demand response, attributing it to non-financial factors. For example, Öhrlund et al. (2019) observed that customers' demand response does not consistently grow in alignment with the magnitude of their financial incentive. Therefore, Öhrlund et al. (2019) concluded that demand response can be partially attributed to non-financial factors such as social responsibilities. Given that MDCs already exist in our study and assuming that rate changes solely affect the financial incentives of customers, our study can isolate the impact

^{21.} Large customers are likely to respond more to demand charges than smaller customers due to several factors. They may benefit from economies of scale when implementing energy management solutions like batteries, automation, and solar systems. In addition, large customers might have more financial incentives and resources to invest in such technologies, potentially enhancing their ability to manage demand charges.

of financial incentives from other factors. As such, our study aims to investigate whether demand response to MDCs will be influenced by the financial incentives provided by the MDC rate rather than simply by the presence of MDCs.

Second, our study looks at the implications of a reduction in MDC rates, a perspective not explored in the existing literature. Our study aims to understand whether customers will increase maximum demand when the MDC rate is reduced, while conventional studies have often focused on the opposite scenario. Our study offers valuable insight into customer behavior and the effectiveness of MDCs in influencing customer maximum demand. In particular, understanding how customers react to a decrease in MDC rates is essential for policymakers and utility companies in devising strategies for demand management and cost recovery.

To perform the case study, we utilize a MDC rate change for large customers located in Lethbridge, Canada. The remaining sections of the case study are organized as follows. Section 3.4.2 discusses the background information for the MDCs in Lethbridge. Section 3.4.3 outlines the data used in our case study. Section 3.4.4 summarizes the impact of the MDC rate change on customers' electricity bills. Section 3.5 presents the empirical methodology employed for the case study. Section 3.6 presents the main results. Section 3.7 concludes.

3.4.2 Distribution Tariffs in Lethbridge

Lethbridge is located in southern Alberta, Canada, with a total population of 87,572 in 2016 (Statistics Canada Government of Canada 2023). Agriculture is the primary industry in Lethbridge, and the city holds more than 120 agri-food processing companies (Choose Lethbridge 2020). Lethbridge has a moderate continental climate. A normal summer in Lethbridge has an average temperature between 8°C (46.4 °F) to 26°C (78.8 °F) whereas a normal winter has an average temperature between -11°C (12.2 °F) to 3°C (37.4 °F).

Retail electricity prices in Lethbridge are composed of three parts: energy charges, transmission charges, and distribution charges. MDCs are utilized in Lethbridge to recover the distribution and transmission charges associated with delivering electricity. In Alberta, there is a restructured wholesale market for energy rates, whereas the transmission and distribution charges are set under a regulatory process. In Lethbridge, distribution tariffs are approved by the Alberta Utilities Commission (AUC) and the City of Lethbridge under the cost-ofservice regulation (Alberta Utilities Commission 2021; City of Lethbridge 2023). The rates are designed to ensure that the revenues earned by the utility cover its operating costs and provide a reasonable rate of return to its investors.

The City of Lethbridge owns and operates a Municipal Electric Distribution Utility, which is responsible for providing access to the municipal distribution system and the provincial transmission grid. On the 1st day of January each year, the City of Lethbridge establishes a new Distribution Tariff Bylaw, which sets transmission and distribution charges for different rate classes.

MDCs are mandatory for all medium and large customers in Lethbridge, constituting most commercial and industrial customers, and some large residential customers. The same rate applies to the same rate class regardless of retailer affiliation (which only affects energy charges). Unlike many MDCs that are based on the monthly maximum demand, MDCs in Lethbridge are calculated based on the maximum demand over the past 12 months²² (City of Lethbridge 2023). Compared to previous studies (where MDCs are based on the monthly highest maximum demand), MDCs in Lethbridge provide a strong incentive for customers to reduce annual maximum demand rather than monthly maximum demand. The incentive can

^{22.} The MDCs in Lethbridge is called the "ratchet demand". Customers will pay their demand fees according to the highest metered demand on their services over the past year. If a customer's current monthly demand reading is lower than readings in the past 11 months, then the highest demand in those 11 months is used to calculate the current bill. If the current monthly demand reading is higher than readings in the past 11 months, then the new ratchet value moving forward in time.

exhibit significant seasonal effects based on the characteristics of a customer (e.g., winterpeak vs. summer-peak). On top of these charges, distribution utilities often use rate riders to adjust their revenues if the approved revenues do not match the actual operational costs.

3.4.3 Data

Our study makes use of meter data for 126 large commercial and industrial customers in Lethbridge, which represents the entire population of customers of rate codes 995 and 996 (43 995 customers and 83 996 customers, respectively).²³ The characteristics of the customers, such as industry and retailer affiliation²⁴, are not available from the dataset. This data set provides 15-minute energy usage (kWh) and energy demand (kVA) measures between May 2017 and March 2019 for each customer. On average, a 996 customer uses 222 MWh of electricity per month, about the monthly consumption of a 50,000 sqft grocery store, a public high school, or a hotel (Energy Information Administration 2018). An average 995 customer uses 106 MWh of electricity per month, about the monthly consumption of a 25,000 sqft grocery store. More information on customer electricity consumption characteristics is provided in Section 3.3.3.

In addition to meter data, we also collect data on wholesale electricity rates, weather information, monthly unemployment rate, and future prices of oil and natural gas to control for factors that affect consumption decisions. In Alberta, most industrial and large commercial customers act as self-retailers who purchase power directly from the wholesale electricity market (Alberta Utilities Commission 2021).²⁵ The electricity rates they pay will

^{23.} This rate class contains customers with a maximum energy demand of 150 kVA or higher. Like kW (kilowatts), kVA is used to measure instantaneous energy demand but without considering the energy conversion efficiency of the system. kVA is a measure of apparent power, which is the total amount of power used in a system. kW is a measure of actual power, or the amount of power that is converted for actual use. In a 100% efficient system, kW = kVA. In general, 1 kVA \approx 0.8 kW and 1 kWh is using 1 kW of power continuously for an hour. A detailed description of the rate classes is available in Appendix A.

^{24.} This includes self-retailer, who can purchase electricity directly from the wholesale market.

^{25.} Like many other restructured electricity markets, Alberta's wholesale electricity price (also known as

be the wholesale electricity price plus an energy market trading charge (fixed per MWh). We therefore use the monthly average wholesale electricity price to proxy the energy rate they face. However, as the energy demand of large customers can endogenously affect the wholesale electricity price, we use natural gas future price as an instrumental variable for the wholesale rate. Natural gas future price is an ideal instrument because it can affect the wholesale electricity price by changing the energy supply curve and will affect consumer demand only through its impact on wholesale prices.

Climate data such as temperature, wind speed, and other weather conditions are reported by weather stations on an hourly basis and are available from the Government of Canada. Local economic conditions, such as the monthly unemployment rate, are from Statistics Canada (Statistics Canada 2020). In addition, we collect the future prices for crude oil and natural gas from the Government of Alberta, and future prices for canola and feeder beef from Investing.com, which extracts historical price data from two commodity market in the US: Chicago Mercantile Exchange (CME) and Chicago Board of Trade (CBOT).²⁶ We gather this information because it can impact a customer's electricity demand decisions. Temperatures significantly influence the need of electricity for heating and cooling (Faruqui and Aydin 2017). Economic conditions, as measured by labor market statistics and commodity prices, can affect the level of business activities and the associated demand for electricity in Lethbridge.

the pool price) is determined through a bidding process. The generators submit bids to the system operator specifying several price-quantity offer blocks. The system operator will aggregate these offer blocks to create a supply curve based on the descending order of offer prices. The intersection of the electricity demand and the supply curve determines the wholesale price.

^{26.} We select these commodities because these prices can have a significant impact on Lethbridge's economy. As stated in Serecon (2014), agriculture is the primary industry in Lethbridge, and beef feeding made a contribution of \$621.5 million to the local economy (55% of the agriculture industry). Canola is one of the major crops in Lethbridge and brought \$55 million to the economy. The energy sector is the backbone of Alberta's economy. In 2016, Alberta exported \$51.0 billion in energy resources and about 86.1% is crude oil (Alberta Government 2018).

3.4.4 Effect of MDC Rate on Customer Electricity Bills

How customers will respond to an MDC rate change will be determined by how the MDC rate would affect a customer's electricity bills. To understand the magnitude of the MDC rate change, we compare the monthly electricity bills with different MDC rate levels while holding electricity demand constant. Because most customers in our sample incur maximum demand during the summer, we compare the MDC rate levels in 2017 and 2018, the two summer seasons in our sample period. For 996 customers, the MDC rate changed from \$0.4253 per kVA per day in 2017 to \$0.3941 per kVA per day in 2018, reflecting a 7.34% decrease from 2017 to 2018. For 995 customers, the MDC rate was reduced by 8.58% from 2017 to 2018, changing from \$0.4848 per kVA per day to \$0.4432 per kVA per day.

Table 3.2 presents the summary statistics for the electricity consumption of 995 and 996 customers used in our analysis. Both 995 and 996 consumers had lower kWh consumption in 2018 than in 2017. The kWh consumption decreased by 3.0% and 0.4% for 995 and 996 customers, respectively. The daily maximum demand increased by 1.7% for 995 customers and decreased by 2.2% for 996 customers. The annual maximum demand was essentially unchanged between 2017 and 2018.

Figure 3.1 shows the monthly electricity bills for an average 995 and 996 customer under different MDC rate levels during the summer months.²⁷ We use the kWh and kVA consumption in 2018 as the baseline and calculate the corresponding monthly electricity bills under MDC rate levels in 2017 and 2018. As shown in Figure 3.1, if the 2017 MDC rate were used in 2018, holding consumption constant, the MDCs would increase by \$468.7 and \$645.1 for 995 and 996 customers, respectively. In percentage terms, these changes represent 7.9% and 7.3% of the customer's MDCs and 2.8% and 3.4% of the total electricity bills, respectively.

^{27.} The volumetric charge includes all per kWh charges, including the monthly average wholesale rate, variable distribution and transmission charges, and rate riders. For simplicity, MDCs in a given year are calculated based on the maximum demand within that year.



Figure 3.1: MDC Rates and Corresponding Monthly Electricity Bills

| | 995 Customers | | 996 Customers | |
|--------------------------|---------------|-------|---------------|-------|
| | 2017 | 2018 | 2017 | 2018 |
| Daily kWh Mean | $3,\!525$ | 3,419 | 8,009 | 7,974 |
| Daily kWh Std. Dev. | 1,889 | 1,771 | 9,221 | 9,562 |
| Daily Max kVA Mean | 283 | 288 | 510 | 499 |
| Daily Max kVA Std. Dev. | 179 | 220 | 544 | 547 |
| Annual Max kVA Mean | 460 | 458 | 656 | 657 |
| Annual Max kVA Std. Dev. | 274 | 294 | 638 | 646 |

Table 3.2: Summary Statistics of Electricity Demand

Figure 3.2 shows the histograms of the percentage bill changes in 2017 if the 2018 MDC rate was applied. Most 995 customers faced a percentage bill change between 3% and 5% if the 2018 MDC rate was used. In comparison, most 996 customers will face a bill change between 2% and 3.5%. The long tails in these figures suggest that some customers face more substantial incentives from the MDC rate change than others. For example, while an average 995 customer can only save 4% of the total electricity bill if the 2018 rate was applied in 2017, one customer could save up to 7%. These numbers suggest that the MDC rate change had a modest impact on 995 and 996 customers between 2017 and 2018.

3.5 Methodology

Our case study aims to estimate the demand response resulting from changes in the MDC rate for large commercial and industrial customers. In particular, we investigate how customers adjust maximum demand when facing an MDC rate decrease. However, unlike energy charges that change monthly, the MDC rate only changes once in our sample period, which makes it difficult to estimate a complete demand curve or related elasticity. As a result, our study



Figure 3.2: Distribution of Percentage Bill Changes Resulting from the MDC Rate Change

focus on comparing customer maximum demand before and after the MDC rate change, controlling for other factors affecting customer demand, such as temperature.

In particular, based on a set of predictor variables, we use machine learning techniques to predict counterfactual electricity demand in 2018 based on electricity demand in 2017, holding the MDC rate unchanged. We then compare the prediction errors before and after the MDC rate change. If customers increase maximum demand after the rate change in 2018, the prediction error will be higher in 2018 than in 2017. In addition, since MDCs are based on maximum demand in a 12-month rolling period, they should only affect electricity demand in a period when maximum demand is likely to occur. As most customers have annual maximum demand during summer times, we divide customers into a summer-peak group and a non-summer-peak group based on a summer-peak ratio²⁸ using data from 2017 (prior to the MDC change) and investigate how maximum demand changes for each group during the summer months.²⁹ It is anticipated that the MDCs will only affect maximum demand for summer-peak customers during the summer.

As argued by Burlig et al. (2020), the machine-learning analysis can outperform traditional fixed-effects models in two critical aspects. First, the estimates of the fixed effects model are more sensitive to control variables than the estimates from the machine learning approach. Second, the machine learning approach can avoid certain biases that are common to a fixed effect model. It has been shown that fixed effects models are often inadequate to control for heterogeneity across the subjects (Burlig et al. 2020). This may threaten the validity of the fixed-effects model in our analysis as electricity customers in our sample can exhibit very different behavior patterns that the fixed effects cannot adequately control. Therefore, we use the machine learning approach as the benchmark in our analysis.

^{28.} See Section 3.5.1 for more details.

^{29.} We are unable to perform a formal difference-in-difference analysis because there are no clear boundaries or distinctions between the groups and because the treatment (MDC rate change) is applied to both groups.

In the remainder of this section, we first discuss our classification of summer-peak and non-summer-peak customers. In Section 3.5.2, we provide detailed insight into the machine learning approach employed to generate the counterfactual demand profiles. In Section 3.5.3, we explain how to estimate the demand response based on a before-and-after comparison of prediction errors.

3.5.1 Summer-Peak vs Non-Summer-Peak Group

Most customers in our sample reach annual maximum demand during the summer months. These customers may respond to high temperatures more strongly than customers who have an annual maximum demand in winter. As a result, for a summer-peak customer, the 12month rolling maximum demand tends to occur during high-temperature days.

A natural first measure to determine whether a customer is a summer-peak customer can simply be the average daily temperature for this customer's highest demand day. If this day coincides with a sufficiently hot day, it is likely that the customer will experience high demand on another hot day. The threshold for what constitutes "sufficiently hot" can be determined by the average daily temperature during the highest temperature day. Therefore, a straightforward initial measure is the ratio between the temperature of the highest demand day and the highest temperature day. When a customer's highest demand day coincides with the highest temperature day, this ratio will be 1.

However, it is possible that a customer can reach maximum demand due to temperatureunrelated demand shocks. For this reason, instead of focusing on a single day, the measure could be based on the highest t days. Formally, a summer-peak customer is defined based on the temperature (measured by the cooling degree days, or CDDs) during the customer's t highest-demand days as a fraction of the temperature during the t hottest days. CDDs are often used to measure the cooling requirements of a given day, and is defined as $\max\{0^{\circ}C, \operatorname{temp}^{\circ}C - 18^{\circ}C\}$ in Canada, where $\operatorname{temp}^{\circ}C$ is the average temperature of the given day. Formally, a customer is summer-peak if the ratio defined below is equal to or larger than $0.5.^{30}$

Summer-Peak Ratio =
$$\frac{\sum_{i=1}^{t} \text{CDD}_{i}}{\sum_{j=1}^{t} \text{CDD}_{j}}$$
 (3.1)

where i = 1, 2, ... is a descending rank based on the daily maximum electricity demand, and j = 1, 2, ... is a descending rank based on the daily cooling degree days. When a customer's t highest demand days coincide with the t hottest days in the summer, he will have a summerpeak ratio of 1. If all of the t highest demand days happen during the winter (with 0 CDD), the customer will have a ratio of 0. In the default setting, we let t equal 10. However, changing t to 20 does not change the classification results of the customers in our case study. Figure 3.3 presents the histogram of the summer-peak ratio for 995 and 996 customers, respectively.

Based on the benchmark definition, we find that 41.9% (18 out of 43) of 995 customers and 49.4% (41 out of 83) of 996 customers are summer-peak customers. A small percentage of customers are non-summer customers with a summer-peak ratio of 0. Specifically, 18.6% (8 out of 43) of 995 customers and 13.3% (11 out of 83) of 996 customers are clearly nonsummer-peak customers, whose 12-month rolling maximum demand generally occurs during winter months.

3.5.2 The Machine Learning Approach

We use the machine learning approach to create a prediction model for daily maximum demand for each customer in our sample. The goal of the prediction model is to create counterfactual maximum demand that describe what would have happened if the MDC rate change did not exist. We only use data in 2017 to train the prediction model, effectively

^{30.} We conduct robust tests for alternative cut-off values of 0.8 and 0.2, respectively. The main results are similar to the benchmark case.



Figure 3.3: Distribution of Summer-Peak Ratio for 995 and 996 Customers

holding MDC rate constant. This allows us to estimate the maximum demand response toward MDCs by comparing the out-of-sample prediction errors across years. In particular, if customers are increasing maximum demand in 2018 given lower MDC rate, the prediction model will underestimate the maximum demand in 2018. Compared to a classic fixed effect model, the machine learning approach allows us to specify prediction models for each customer and provide more robust predictions.³¹

Machine learning techniques are attracting increasing attention across various economic studies. As a significant part of the machine learning toolbox, regularized regression has been widely used for model selection, prediction, and causal inference. Unlike OLS regression, regularized regression imposes a regularization penalty for adding explanatory variables to balance between bias and variance (Ahrens et al. 2020). This feature makes regularized regression a generally better tool than OLS regression regarding out-of-sample prediction. One of the most popular regularized regression approaches is the Least Absolute Shrinkage and Selection Operator (LASSO), which penalizes the absolute size of the coefficient estimates.

LASSO regression was first introduced by Tibshirani (1996) in order to enhance prediction accuracy and perform subset selection when a a large number of predictors exist. It minimizes the residual sum of squares subject to a penalty on the absolute size of coefficient estimates:

$$\widehat{\beta}_{lasso}(\lambda) = argmin\frac{1}{n}\sum_{i=1}^{n}(y_i - x'_i\beta)^2 + \frac{\lambda}{n}\sum_{j=1}^{p}\psi_j |\beta_j|$$
(3.2)

Compared to the OLS estimator, the LASSO estimator contains a second part as a regularization penalty. The overall penalty level is controlled by λ (tuning parameter), and ψ_j is the predictor-specific penalty loading. There is usually no predictor-specific penalty loading,

^{31.} Appendix C provides the corresponding methodology and results for a fixed-effect approach.
and ψ_i will be set to one.³²

LASSO Prediction models have three advantages compared to traditional prediction methods based on the OLS estimator. First, LASSO prediction models have significantly better prediction accuracy than traditional prediction methods by balancing between bias and variance (Ahrens et al. 2020). Second, LASSO prediction models can better capture heterogeneity between customers by self-selecting the best prediction model for each customer (Burger et al. 2020). This is important in estimating the maximum demand of the customer, as different customers can exhibit different demand patterns. Third, LASSO prediction models can often produce more robust results than traditional prediction methods (Burger et al. 2020). In contrast, traditional prediction methods are very sensitive to different selections of prediction variables, leading to a complicated model selection process, especially when faced with a large set of potential prediction variables. Machine learning techniques have been used for load prediction (e.g., Fan et al. 2014), but our paper is the first to use such technologies to analyze the MDC responses.

To examine how maximum demand changes from 2017 to 2018 as the MDC rate changes, we forecast each customer's daily maximum demand during the summer months using machine learning techniques. The machine learning algorithms will help us select the best prediction model from a large set of potential explanatory variables. The prediction model has the following structure but will only include appropriate prediction variables based on the extended Bayesian information criteria (EBIC)³³:

$$D_{it} = \gamma_{i1}T_t + \gamma_{i2}T_t^2 + \gamma_{i3}T_t^3 + \gamma_{i4}T_t^4 + \beta X + \delta Z + \varepsilon_{it}$$

$$(3.3)$$

^{32.} Generally, setting ψ_j to one maintains simplicity and consistency for the LASSO estimator, making it easier to apply and interpret the regularization process in LASSO regression (Ahrens et al. 2020).

^{33.} Compared to Bayesian information criteria (BIC), EBIC constrain the number of unknown parameters, and limit the model space, which provides a more stringent way to select the appropriate λ in Equation (3.2), reducing the likelihood of over-fitting (Chen and Chen, 2008).

where D_{it} represents the daily maximum demand in kVA for the customer *i* in day *t*. We estimate kVA demand at the daily level as the MDCs are based upon the highest daily maximum demand in 12 months. Compared to estimating at the hourly level, a daily level estimation can enhance the prediction accuracy of the model and is commonly used for estimating maximum demand (Gibbons and Faraqui, 2014).

 T_t is the CDD in day t. Temperature is widely considered a key variable for forecasting electricity demand due to the high penetration of electrical cooling and heating. We include the first to fourth temperature polynomial to capture the nonlinearity effects of temperature on daily maximum demand. X stands for other control variables, such as wind speed, rainfall, snowfall, daylight hours, humidity, visibility, commodity prices for crude oil, natural gas, feeder beef and canola, and unemployment rate. We use the cluster-robust covariance estimator (sandwich estimator) and clustered standard errors on the customer level to correct autocorrelation and heteroskedasticity in the residuals.

The selected control variables can have important impacts on electricity demand. Wind speed can influence the electrical demand as higher wind speed will increase the heating loss for buildings. Longer daylight hours will decrease facilities' lighting and heating needs, reducing the electrical demand. The unemployment rate and commodity prices for crude oil, feeder beef, and canola are important economic indicators for Lethbridge, which can significantly impact industrial and commercial customers. Natural gas commodity prices are an important proxy for the electricity prices faced by large industrial and commercial customers, as natural gas generators are the backbone of Alberta's electricity generation fleet, and changes in wholesale electricity prices largely follow the changes in natural gas costs (MSA, 2018). Z stands for time dummy variables for the day of the week, holidays, and the month of the year.

The goal of the prediction model is to create counterfactual electricity demands that de-

scribe what would have happened if the MDC change did not exist. Therefore, we only use data from 2017 to "train" the model. To understand the in-sample and out-of-sample prediction accuracy, we randomly selected 70% of the summer days in 2017 to "train" the model and validate the model on the remaining 30% of days. Because we only have complete data in summer months for both 2017 and 2018, we focus on summer days and compare between summer-peak and non-summer-peak customers. We include non-summer-peak customers as a placebo test because they are not likely to incur maximum demand during this period and have no incentives to adjust daily maximum demand.

The general goodness of fit measure for the prediction models can be measured by the in-sample and out-of-sample Mean Absolute Error (MAE) and Mean Absolute Percentage Error (MAPE):

$$MAE = \frac{1}{n} \sum_{i=1}^{n} |\hat{y}_i - y_i|$$

$$MAPE = \frac{100\%}{n} \sum_{i=1}^{n} \left| \frac{\hat{y}_i - y_i}{y_i} \right|$$

where y_i is the daily maximum demand for each customer.

3.5.3 Before-After Comparison

The next step of our analysis is to compare the counterfactual consumption profiles with the observed meter data. However, since counterfactual consumption is subject to prediction errors, we cannot directly compare it with actual consumption. Instead, we focus on the prediction errors before and after the MDC change. If the MDC change causes a substantial behavioral change, the prediction errors after the change will differ because the prediction errors after the MDC change causes and the change in actual

demand behavior. If the model errors are consistent before and after the MDC change, the changes in prediction errors will imply a change in the daily maximum demand.

To assess whether changes in prediction errors are caused by the MDC rate change and not just model forecast errors, we conduct a placebo test by randomly assigning dates in 2017 into a "pre" and a "post" period. We then follow our standard machine learning procedure to obtain prediction errors for each period. Because the "structural break" between periods (pseudo "years") does not actually exist, any differences in prediction errors across periods can be attributed to model errors. If the model prediction is consistent across time after controlling for temperature and economic conditions, we anticipate that the differences in prediction errors across periods will be statistically insignificant and close to zero. The results of the placebo test are provided in Section 3.6.2.

For summer-peak customers, if the prediction errors are higher in 2018 than in 2017 during hot days, customers are increasing maximum demand. This could be because a lower MDC rate reduces the costs of having a higher maximum demand. In contrast, if the prediction errors in 2018 are lower than those in 2017, customers are reducing the maximum demand, further reducing the electricity bills they face by utilizing the MDC. For non-summer-peak customers, if the prediction model is well specified, we expect the difference in prediction errors during summer days to be close to zero and insignificant.

If a customer has perfect information regarding when maximum demand will occur, comparing the annual maximum demand across years, controlling for other related factors, can reveal the effects of the MDC rate on maximum demand. However, in reality, customers often do not know when maximum demand will occur, so they might adjust maximum demand whenever they feel a day may become the peak day of a year. For summer-peak customers, this usually happens when the days are sufficiently hot. The probability of hitting the maximum demand increases as the temperature increases. As a result, customers face stronger motivations to adjust peak demand as days get hotter, or as their daily maximum demand gets higher. For this reason, we divide each customer's daily maximum demand into five temperature quantiles for each year and compare the prediction errors in each quantile. Formally, we use the following fixed effects model with cluster-robust standard errors to compare the prediction errors across time:

$$error_{it} = \sum_{q}^{5} \gamma_{q=2}Q_{qt} + \sum_{q}^{5} \delta_{q=2}Q_{qt} * Year_{2018} + Year_{2018} + In-sample_{it} + \varepsilon_{it}$$
(3.4)

where $error_{it}$ is the prediction error in percentage terms calculated as the actual logged kVA minus predicted logged kVA. Q_{qt} is the factorial variable for the qth temperature quantile. $Year_{2018}$ is the dummy variable for the year 2018. The difference in prediction errors across years for different quantiles are captured by δ_q . If customers respond to the demand change, at least one of the δ_q should differ from zero. As summer customers are more likely to respond to the demand change when daily maximum demand is sufficiently high, δ_5 should be more likely to be significant and differ from zero. For non-summer customers, we expect all δ_q to be close to zero. As the in-sample prediction errors tend to differ from the out-of-sample errors, we use $In-sample_{it}$, which is an indicator for in-sample observation, to recenter the in-sample and out-of-sample errors toward zero.

Our empirical strategy can be summarized as follows. If customers do not respond to the MDC change, the prediction errors will only reflect the lack of prediction accuracy due to the prediction power of the model. If the prediction power before and after the MDC change are similar (e.g., no time-variant error caused by omitted variables), there will be no substantial changes in the prediction errors. If customers do respond to the MDC change by adjusting their maximum demand, the prediction errors after the MDC rate change will also reflect this behavioral change. In addition to the MDCs, other unobservable factors can also affect

maximum demand, and it is challenging to establish a causal relationship between MDC rate and customer behavior (Öhrlund et al. 2019). Our empirical strategy uses the narrow time window to identify the potential effect of MDCs, as MDCs will only affect maximum demand on certain high-demand days. In addition, we utilize the comparison between summerpeak and non-summer-peak customers because the MDC rate change should only affect the summer-peak customers. However, our strategy will not capture the effects of MDCs on energy conservation investments, as most energy conservation technology can reduce electricity demand on both high-demand and low-demand days.

3.6 Results

3.6.1 Model Performance

The model's prediction accuracy is measured by the in-sample and out-of-sample MAE and MAPE in 2017, which is presented in Table 3.3. We reestimate Equation (3.3) using the OLS estimator on the same training data to examine how the LASSO prediction model compared with the traditional OLS prediction model. As shown in Table 3.3, the LASSO estimator provides better out-of-sample prediction accuracy than the OLS estimator at the cost of lower in-sample MAPE, reflecting a potential overfitting problem of training data in the OLS regression. The average in-sample and out-of-sample MAPE using the LASSO prediction model is 5.4% and 6.6% for 995 customers. The model performs slightly better for 996 customers, with an in-sample and out-of-sample MAPE of 3.9% and 5.2%, respectively.

| 005 | 2017 In-sample | | 2017 Out-of-sample | | |
|--------------|------------------------|--------------------------|------------------------|------------------------------|--|
| 990 | MAE | MAPE | MAE | MAPE | |
| LASSO | 13.2 | 5.4% | 16.0 | 6.6% | |
| OLS | 11.8 | 4.7% | 18.0 | 7.4% | |
| 0.06 | 2017 In-sample | | | | |
| 006 | 2017 II | n-sample | 2017 Ou | ut-of-sample | |
| 996 | 2017 In MAE | n-sample MAPE | 2017 Ou MAE | it-of-sample MAPE | |
| 996 LASSO | 2017 In MAE 15.2 | n-sample MAPE 3.9% | 2017 Ou MAE 17.8 | nt-of-sample MAPE 5.2% | |

Table 3.3: Summary of MAE and MAPE

Given the complexity of predicting individual maximum demand, previous studies obtained a wide range of prediction MAPEs using various prediction models. Our prediction models perform similarly to these studies. For example, Ke et al. (2016) predicted the maximum demand of a campus building using polynomial regression, and the resulting MAPE is 3.37%. Fan et al. (2019) predicted the maximum demand for residential buildings using a general linear model and obtained a MAPE of 4.6%. Kim et al. (2022) estimate the daily maximum demand for residential customers using artificial neural networks and get an average MAPE of 11.7%. Taheri and Razban (2022) use a probabilistic regression model to forecast daily maximum demand for commercial and manufacturing buildings and obtain an average MAPE of 9.3%.

3.6.2 Before-After Maximum Demand Comparison

Table 3.4 presents the estimation results of Equation (3.4) for 995 customers. Columns (1) and (2) present the results for summer-peak and non-summer-peak customers, respectively. Columns (3) and (4) add temperature controls to the estimation equation, which aims to control for prediction errors that correlate with the daily temperature. The coefficients of

interest are the interactions between the quantiles and the year 2018, which captures the difference in daily maximum demand between 2017 and 2018 in percentage terms. For example, the first column suggests that, on average, a summer-peak customer's daily maximum demand increased by 3.1% in 2018, but it is not statistically significant.

As shown in Table 3.4, controlling for temperature, a summer-peak customer's daily maximum demand increased on average by 4% in 2018 in the 5th demand quantile. This implies that medium-sized summer-peak customers have higher maximum demand in 2018 (with a lower MDC rate) when the annual maximum demand is likely to occur. In comparison, as anticipated, the changes in maximum demand from 2017 to 2018 are insignificant for non-summer-peak customers. We conduct a Wald test to test for the equality of the two coefficients of interest from columns (2) and (4) within the framework of a seemingly unrelated regression. The null hypothesis that the two coefficients are equal is rejected at a 0.1% significance level. The results show that customers may increase maximum demand when faced with a lower MDC rate. However, due to data limitations, it is challenging to infer a causal relationship between MDC rate change and changes in maximum demand.

| | (1) | (2) | (3) | (4) |
|--------------------------|--------------|--------------|--------------|--------------|
| | Error | Error | Error | Error |
| 5th Quantile * 2018 | 0.031 | -0.006 | 0.040** | 0.007 |
| | (0.019) | (0.053) | (0.019) | (0.055) |
| 4th Quantile * 2018 | 0.020 | -0.056 | 0.023 | -0.043 |
| | (0.022) | (0.049) | (0.023) | (0.052) |
| 3rd Quantile * 2018 | 0.031 | -0.009 | 0.028 | -0.000 |
| | (0.018) | (0.039) | (0.020) | (0.039) |
| 2nd Quantile * 2018 | 0.027 | -0.010 | 0.033 | -0.001 |
| | (0.020) | (0.051) | (0.031) | (0.042) |
| 5th Quantile | 0.104*** | 0.251*** | 0.127^{**} | 0.249*** |
| | (0.034) | (0.039) | (0.047) | (0.040) |
| 4th Quantile | 0.061^{**} | 0.191*** | 0.076^{*} | 0.185*** |
| | (0.030) | (0.046) | (0.041) | (0.048) |
| 3rd Quantile | 0.037 | 0.114^{**} | 0.045 | 0.109** |
| | (0.024) | (0.041) | (0.030) | (0.043) |
| 2nd Quantile | 0.022^{*} | 0.086^{*} | 0.039 | 0.097^{**} |
| | (0.010) | (0.041) | (0.027) | (0.033) |
| Year (2018) | -0.072*** | 0.045 | -0.081*** | 0.037 |
| | (0.024) | (0.047) | (0.023) | (0.049) |
| Summer-Peak Customers | Yes | No | Yes | No |
| Add Temperature Controls | No | No | Yes | Yes |
| Observations | 1,136 | 1,471 | 1,136 | 1,471 |

 Table 3.4:
 Panel Fixed Effects Results - 995 Customers

Notes: This table presents the estimation results of Equation (3.4) for 995 customers. The dependent variable is prediction error. The 2nd, 3rd, 4th, and 5th quantiles represent different quantiles in daily maximum temperatures, with the 5th quantile indicating the hottest 20% days. Columns (1) and (2) present the estimation results for summer-peak and non-summer-peak customer groups, respectively. Columns (3) and (4) present the same estimation results but with additional control for temperature variables. Std. Dev. in parentheses* p < 0.05, ** p < 0.01, *** p < 0.001. 176

Table 3.5 presents the estimation results for 996 customers. Unlike 995 customers, all coefficients of interest are insignificant. An insignificant result can suggest two possibilities. First, the null hypothesis is true, and there is no real effect. Second, the null hypothesis is not true, but we do not have sufficient evidence to reject it due to sample limitations. In our case study, it is challenging to rule out the second possibility due to data limitations. More data is needed to support the evidence that large customers do not respond to MDC rate decreases.

MDCs can have a more substantial impact on larger customers, leading to stronger incentives to reduce maximum demand. This could potentially explain why larger customers do not respond to a decrease in the MDC rate. Faced with high MDC bills, larger customers have more substantial incentives to invest in energy conservation or load-shifting technologies, such as batteries. Therefore, it is likely that these customers have already responded to the MDCs before. The remaining electricity demand could be largely inflexible, leading to minimum responses.

These findings suggest a possible asymmetric response to changes in MDC rates. When MDC rates increase, customers may be inclined to make significant investments in energy conservation or demand-shifting technologies, which could become sunk costs and represent a long-term commitment to reducing maximum demand. Conversely, when MDC rates decrease, the response might be limited to short-term behavioral adjustments, such as modifying cooling or heating needs, which generally have a smaller impact. This potential asymmetry highlights the complexity of customer behavior in response to MDC rate changes and underscores the need for more nuanced tariff designs that account for these varying responses.

| | (1) | (2) | (3) | (4) |
|--------------------------|-----------|----------|-----------|---------------|
| | Error | Error | Error | Error |
| 5th Quantile * 2018 | -0.019 | -0.014 | -0.015 | -0.006 |
| | (0.014) | (0.019) | (0.014) | (0.019) |
| 4th Quantile * 2018 | -0.010 | -0.001 | -0.012 | 0.004 |
| | (0.014) | (0.021) | (0.013) | (0.021) |
| 3rd Quantile * 2018 | -0.013 | -0.013 | -0.022 | -0.010 |
| | (0.016) | (0.018) | (0.015) | (0.018) |
| 2nd Quantile * 2018 | -0.016 | -0.015 | -0.018 | -0.007 |
| | (0.018) | (0.018) | (0.020) | (0.019) |
| 5th Quantile | 0.096*** | 0.211*** | 0.146*** | 0.214*** |
| | (0.016) | (0.028) | (0.023) | (0.028) |
| 4th Quantile | 0.068*** | 0.152*** | 0.102*** | 0.155^{***} |
| | (0.013) | (0.023) | (0.018) | (0.023) |
| 3rd Quantile | 0.056*** | 0.109*** | 0.076*** | 0.111*** |
| | (0.015) | (0.021) | (0.018) | (0.021) |
| 2nd Quantile | 0.044*** | 0.097*** | 0.066*** | 0.099*** |
| | (0.013) | (0.017) | (0.015) | (0.016) |
| Year (2018) | -0.028*** | 0.005 | -0.081*** | -0.005 |
| | (0.018) | (0.016) | (0.023) | (0.023) |
| Summer-Peak Customers | Yes | No | Yes | No |
| Add Temperature Controls | No | No | Yes | Yes |
| Observations | 2,289 | 2,861 | 2,289 | 2,861 |

 Table 3.5:
 Panel Fixed Effects Results - 996 Customers

Notes: This table presents the estimation results of Equation (3.4) for 996 customers. The dependent variable is prediction error. The 2nd, 3rd, 4th, and 5th quantiles represent different quantiles in daily maximum temperatures, with the 5th quantile indicating the hottest 20% days. Columns (1) and (2) present the estimation results for summer-peak and non-summer-peak customer groups, respectively. Columns (3) and (4) present the same estimation results but with additional control for temperature variables. Std. Dev. in parentheses* p < 0.05, ** p < 0.01, *** p < 0.001. 178

Placebo Test

This section presents the results of the placebo test, where only the pre-treatment data are used to develop the prediction model. This allows us to isolate the effects of the MDC rate change and focus on the prediction model's inherent error. The main results of our paper will be robust if the model error is close to zero.

Table 3.6 presents the fixed effects model for the prediction errors of the placebo test. Columns (1) through (4) correspond to summer-peak 995 and 996 customers with and without temperature controls, respectively. As shown in Table 3.6, all coefficients capturing the prediction error differences between the "pseudo" pre- and post-treatment periods are statistically insignificant. These results suggest that the model error is unlikely to significantly affect the main findings in our paper.

| | (1) | (2) | (3) | (4) |
|--------------------------|----------|----------|----------|----------|
| | 995 | 995 | 996 | 996 |
| 5th Quantile * 2018 | 0.006 | -0.006 | -0.000 | -0.003 |
| | (0.013) | (0.125) | (0.003) | (0.004) |
| 4th Quantile * 2018 | 0.021 | 0.000 | 0.000 | -0.006** |
| | (0.014) | (0.014) | (0.003) | (0.003) |
| 3rd Quantile $*$ 2018 | 0.011 | -0.001 | -0.000 | -0.004 |
| | (0.012) | (0.018) | (0.004) | (0.003) |
| 2nd Quantile $*$ 2018 | -0.000 | -0.006 | -0.003 | -0.007 |
| | (0.013) | (0.011) | (0.004) | (0.003) |
| 5th Quantile | 0.081*** | 0.152*** | 0.071*** | 0.115*** |
| | (0.013) | (0.022) | (0.006) | (0.007) |
| 4th Quantile | 0.043*** | 0.096*** | 0.055*** | 0.083*** |
| | (0.011) | (0.018) | (0.006) | (0.005) |
| 3rd Quantile | 0.035*** | 0.065*** | 0.034*** | 0.049*** |
| | (0.014) | (0.016) | (0.004) | (0.004) |
| 2nd Quantile | 0.027*** | 0.042*** | 0.030*** | 0.044*** |
| | (0.009) | (0.010) | (0.004) | (0.004) |
| Placebo Year (2018) | -0.007 | -0.005 | 0.004 | 0.000 |
| | (0.010) | (0.010) | (0.151) | (0.002) |
| Summer-Peak Customers | Yes | Yes | Yes | Yes |
| Add Temperature Controls | No | Yes | No | Yes |
| Observations | 1,808 | 1,808 | 2,861 | 2,861 |

Table 3.6:PlaceboTest - 995996Customers

Notes: This table presents the estimation results of the placebo test. The dependent variable is prediction error. The 2nd, 3rd, 4th, and 5th quantiles represent different quantiles in daily maximum temperatures, with the 5th quantile indicating the hottest 20% days. Columns (1) and (2) present the estimation results for 995 customers with and without temperature controls, respectively. Columns (3) and (4) present the same estimation results for 996 customers. Std. Dev. in parentheses* p < 0.05, ** p < 0.01, *** p < 0.001.

3.7 Conclusion

Our paper reviews the literature on the current debate as to whether MDC can provide fair and efficient fixed-cost recovery for utilities. We discussed how future studies should be conducted to provide further evidence for the debate. In addition, as a key motivation for the MDCs, we investigated how customers respond to MDCs by reviewing the related studies and, finally, conducted a case study to study how medium and large customers respond to an MDC rate decrease.

In terms of customer responses, existing studies find that customers will reduce maximum demand when faced with MDCs. However, how customers adjust maximum demand when the MDC rate decreases has not been investigated. In theory, customers may increase maximum demand because it is less expensive to do so with a lower MDC level. The opposite can also be true because the decrease in MDC rates is temporary and customers may still be in a learning phase to reduce maximum demand. By doing a case study, we find limited evidence that large customers will increase or decrease maximum demand when faced with a lower MDC rate. We show that medium-sized customers exhibit a higher maximum demand when the MDC rate is lower. These findings may be explained by an asymmetric response to MDC rate changes, where increases in MDCs can motivate both investment and behavioral changes, while decreases in MDC rates typically do not lead to significant changes because investments have already been sunk and demand responses are limited to temporary behavioral changes. However, due to the data limitations of our study, future work is needed to reassess the causal relationship between the MDC rate level and customer maximum demand.

From an economic point of view, the volumetric charge should reflect the time-varying costs of generating electricity. Recovering fixed costs using volumetric charges will lead to inefficient consumption decisions and cross-subsidy concerns. Economic theory suggests that fixed costs should be recovered through fixed charges (Borenstein 2016). However, increasing fixed charges can have substantial distributional effects and face severe political barriers. An income-based fixed charge can alleviate some of the distributional concerns associated with a universal increase in fixed charges. However, designing efficient and fair income-based fixed charges would require household-level income data, which may be challenging for some jurisdictions.

As an alternative to higher fixed charges, MDCs are favored by utilities for several reasons. The most important reason is that it can reduce utilities' reliance on volumetric charges to recover fixed charges, reducing cross-subsidy concerns, and providing better cost recovery for utilities (Hledik 2014). At the same time, compared to a tariff with higher fixed costs, MDCs can reduce the distributional effects between smaller and larger customers, leading to fewer political obstructions (Wood et al. 2016). However, although there is some evidence that smaller customers tend to be low-income customers, the correlation can become weaker in the future, since energy conservation and DER technologies are more accessible for high-income customers. Therefore, it could be more economically efficient to directly target low-income customers through social programs (Wood et al. 2016).

The second reason is that MDCs can better align customer billing with the cost structure of utilities, leading to fairer cost recovery. However, around 80% of the costs related to investments in generation, distribution, and transmission capacity are sunk (Passey et al. 2017). From an economic point of view, sunk costs should be recovered using fixed charges as the price signal for the efficient use of the electric grid should be based on long-run marginal costs only (Brown et al. 2015; Borenstein 2016). However, in the future, with increasing electrification, the unsunk part of LRMC for distribution services is expected to increase, and MDCs could play a role in recovering this part of the costs.

The third reason is that MDCs can be used to reduce system maximum demand, leading

to lower system investment costs. Existing studies generally find that customers will reduce maximum demand when facing the MDCs. However, as most MDCs are targeting customers' individual maximum demand, whether MDCs can be used to curtail system peak demand will also be determined by whether a customer's individual peak demand coincides with the system peak demand. However, most studies find that non-coincide MDCs are ineffective at reducing system maximum demand compared to other tariff designs with the same goal, such as CPP.

In the short run, MDCs can be used to address some of the current challenges (e.g., crosssubsidies) faced by the utilities. However, more effective tariff designs could exist in the long run: use real-time pricing to recover volumetric charges, set transmission and distribution charges at their long-run marginal cost level, use fixed charges to recover other fixed charges, and use social programs to address distributional concerns.

Future studies are needed to assess the efficiency of MDCs in fulfilling their objectives compared to alternative tariff designs. First, further research is needed to assess the effectiveness of various MDC designs in reducing maximum demand for various customer classes. In particular, for residential consumers, these studies play a crucial role in understanding how MDCs can be designed to optimize the social benefits of DERs. Additionally, for larger consumers, these studies can offer insights into designing more efficient demand response programs.

Second, the evidence regarding whether MDCs can offer more cost-reflective recovery is scarce in the literature. In particular, to what extent can current MDC rates reflect the locational long-run marginal costs of the electric network system has not been explored.

Finally, in terms of distributional effects, it is crucial to explore how MDCs impact consumer electricity bills compared to alternative tariff designs (e.g., income-based fixed charges). Specifically, as new technologies such as batteries become more affordable for higher-income consumers and can significantly influence MDCs, it is essential to assess the distributional effects of MDCs in an environment with expanding penetration of DERs.

References

- Abdelmotteleb, Ibtihal, Tomás Gómez, José Pablo Chaves Ávila, and Javier Reneses. 2018. "Designing efficient distribution network charges in the context of active customers." *Applied Energy* 210 (January): 815–826. ISSN: 03062619, accessed April 11, 2024. https://doi.org/10.1016/j.apenergy.2017.08.103. https://linkinghub.elsevier. com/retrieve/pii/S0306261917311236.
- AESO. 2023. AESO 2022 Annual Market Statistics.
- Ahrens, Achim, Christian B. Hansen, and Mark E. Schaffer. 2020. "lassopack: Model selection and prediction with regularized regression in Stata." *The Stata Journal: Promoting communications on statistics and Stata* 20, no. 1 (March): 176–235. ISSN: 1536-867X, 1536-8734, accessed October 17, 2023. https://doi.org/10.1177/ 1536867X20909697. http://journals.sagepub.com/doi/10.1177/1536867X20909697.
- Alberta Government. 2018. *Highlights of the Alberta Economy 2017.* chrome-extension: //efaidnbmnnibpcajpcglclefindmkaj/https://open.alberta.ca/dataset/10989a51-f3c2-4dcb-ac0f-f07ad88f9b3b/resource/513eef5f-aa53-4cde-888d-8e52822b6db4/download/sp-eh-highlightsabeconomypresentation.pdf.
- Alberta Utilities Commission. 2021. Distribution System Inquiry Final Report. Alberta Utilities Commission.
- Bartusch, Cajsa, and Karin Alvehag. 2014. "Further exploring the potential of residential demand response programs in electricity distribution." Applied Energy 125 (July): 39–59. ISSN: 03062619, accessed September 15, 2023. https://doi.org/10. 1016/j.apenergy.2014.03.054. https://linkinghub.elsevier.com/retrieve/pii/ S0306261914002943.
- Bartusch, Cajsa, Fredrik Wallin, Monica Odlare, Iana Vassileva, and Lars Wester. 2011. "Introducing a demand-based electricity distribution tariff in the residential sector: Demand response and customer perception." *Energy Policy* 39, no. 9 (September): 5008– 5025. ISSN: 03014215, accessed September 15, 2023. https://doi.org/10.1016/j.enpol. 2011.06.013. https://linkinghub.elsevier.com/retrieve/pii/S030142151100468X.
- Baumol, William J., and Robert D. Willig. 1981. "Fixed Costs, Sunk Costs, Entry Barriers, and Sustainability of Monopoly." *The Quarterly Journal of Economics* 96, no. 3 (August): 405. ISSN: 00335533, accessed April 7, 2024. https://doi.org/10.2307/1882680. https://academic.oup.com/qje/article-lookup/doi/10.2307/1882680.
- Berg, Sandford, and Andreas Savvides. 1983. "The theory of maximum kW demand charges for electricity." *Energy Economics* 5, no. 4 (October): 258–266. ISSN: 01409883, accessed September 15, 2023. https://doi.org/10.1016/0140-9883(83)90030-0. https://linkinghub.elsevier.com/retrieve/pii/0140988383900300.

- Biggar, Darryl. 2009. "Is Protecting Sunk Investments by Consumers a Key Rationale for Natural Monopoly Regulation?" *Review of Network Economics* 8, no. 2 (January 1, 2009). ISSN: 1446-9022, 2194-5993, accessed October 30, 2023. https://doi.org/10. 2202/1446-9022.1173. https://www.degruyter.com/document/doi/10.2202/1446-9022.1173/html.
- Bird, Lori, Carolyn Davidson, Joyce McLaren, and John Miller. 2015. "Impact of Rate Design Alternatives on Residential Solar Customer Bills: Increased Fixed Charges, Minimum Bills and Demand-Based Rates." *Renewable Energy*.
- Boampong, Richard, and David P. Brown. 2020. "On the benefits of behind-the-meter rooftop solar and energy storage: The importance of retail rate design." *Energy Economics* 86 (February): 104682. ISSN: 01409883, accessed October 17, 2023. https://doi.org/10.1016/j.eneco.2020.104682. https://linkinghub.elsevier.com/retrieve/pii/S0140988320300219.
- Borenstein, Severin. 2016. "The economics of fixed cost recovery by utilities." The Electricity Journal 29, no. 7 (September): 5–12. ISSN: 10406190, accessed September 15, 2023. https://doi.org/10.1016/j.tej.2016.07.013. https://linkinghub.elsevier.com/ retrieve/pii/S1040619016301130.
- Borenstein, Severin, and James B. Bushnell. 2022. "Do Two Electricity Pricing Wrongs Make a Right? Cost Recovery, Externalities, and Efficiency." *American Economic Journal: Economic Policy* 14, no. 4 (November 1, 2022): 80–110. ISSN: 1945-7731, 1945-774X, accessed October 30, 2023. https://doi.org/10.1257/pol.20190758. https: //pubs.aeaweb.org/doi/10.1257/pol.20190758.
- Borenstein, Severin, Meredith Fowlie, and James Sallee. 2021. Designing Electricity Rates for An Equitable Energy Transition. Next 10. Accessed October 23, 2023.
- Brown, David P., and David E.M. Sappington. 2018. "On the role of maximum demand charges in the presence of distributed generation resources." *Energy Economics* 69 (January): 237–249. ISSN: 01409883, accessed September 15, 2023. https://doi.org/10.1016/ j.eneco.2017.11.023. https://linkinghub.elsevier.com/retrieve/pii/S0140988317304152.
- Brown, Toby, Ahmad Faruqui, and Léa Grausz. 2015. "Efficient tariff structures for distribution network services." *Economic Analysis and Policy* 48 (December): 139–149. ISSN: 03135926, accessed October 17, 2023. https://doi.org/10.1016/j.eap.2015.11.010. https://linkinghub.elsevier.com/retrieve/pii/S0313592615300552.
- Burger, Scott P., Christopher R. Knittel, Ignacio J. Perez-Arriaga, Ian Schneider, and Frederik Vom Scheidt. 2020. "The Efficiency and Distributional Effects of Alternative Residential Electricity Rate Designs." *The Energy Journal* 41, no. 1 (January): 199–240. ISSN: 0195-6574, 1944-9089, accessed April 5, 2024. https://doi.org/10.5547/ 01956574.41.1.sbur. https://journals.sagepub.com/doi/10.5547/01956574.41.1.sbur.

- Burlig, Fiona, Christopher Knittel, David Rapson, Mar Reguant, and Catherine Wolfram. 2020. "Machine Learning from Schools about Energy Efficiency." Journal of the Association of Environmental and Resource Economists 7 (6): 1181–1217.
- Caves, Douglas W., Laurits R. Christensen, and Joseph A. Herriges. 1984. "Modelling alternative residential peak-load electricity rate structures." *Journal of Econometrics* 24, no. 3 (March): 249–268. ISSN: 03044076, accessed September 25, 2023. https://doi.org/10.1016/0304-4076(84)90052-6. https://linkinghub.elsevier.com/retrieve/pii/0304407684900526.
- **Choose Lethbridge.** 2020. "Agriculture | Economic Development Lethbridge." Accessed October 17, 2023. https://chooselethbridge.ca/agriculture.
- City of Lethbridge. 2023. "Utility rates." City of Lethbridge. Accessed October 17, 2023. https://www.lethbridge.ca/utilities/utility-rates/.
- Darby, Sarah J., and Eoghan McKenna. 2012. "Social implications of residential demand response in cool temperate climates." *Energy Policy* 49 (October): 759–769. ISSN: 03014215, accessed June 18, 2024. https://doi.org/10.1016/j.enpol.2012.07.026. https://linkinghub.elsevier.com/retrieve/pii/S0301421512006076.
- Darghouth, Naim, Galen Barbose, Andrew Mills, Ryan Wiser, Pieter Gagnon, and Lori Bird. 2017. Exploring Demand Charge Savings from Residential Solar. LBNL-1007030, 1373280. NREL, July 31, 2017. Accessed October 17, 2023. https: //doi.org/10.2172/1373280. http://www.osti.gov/servlets/purl/1373280/.
- Dutta, Goutam, and Krishnendranath Mitra. 2017. "A literature review on dynamic pricing of electricity." Journal of the Operational Research Society 68, no. 10 (October): 1131–1145. ISSN: 0160-5682, 1476-9360, accessed October 17, 2023. https://doi.org/10. 1057/s41274-016-0149-4. https://www.tandfonline.com/doi/full/10.1057/s41274-016-0149-4.
- Elmallah, Salma, Anna M Brockway, and Duncan Callaway. 2022. "Can distribution grid infrastructure accommodate residential electrification and electric vehicle adoption in Northern California?" *Environmental Research: Infrastructure and Sustainability* 2, no. 4 (December 1, 2022): 045005. ISSN: 2634-4505, accessed April 12, 2024. https://doi.org/10.1088/2634-4505/ac949c. https://iopscience.iop.org/article/10.1088/2634-4505/ac949c.
- Energy Information Administration. 2018. "Use of energy in commercial buildings." Accessed October 17, 2023. https://www.eia.gov/energyexplained/use-of-energy/commercial-buildings.php.

- Fan, Cheng, Fu Xiao, and Shengwei Wang. 2014. "Development of prediction models for next-day building energy consumption and peak power demand using data mining techniques." Applied Energy 127 (August): 1–10. ISSN: 03062619, accessed October 17, 2023. https://doi.org/10.1016/j.apenergy.2014.04.016. https://linkinghub.elsevier. com/retrieve/pii/S0306261914003596.
- Faruqui, Ahmad, and Geronimo Aydin. 2017. "Moving Forward with Electricity Tariff Reform." Cato Institute, September 25, 2017. Accessed October 17, 2023. https://www. cato.org/regulation/fall-2017/moving-forward-electricity-tariff-reform.
- Faruqui, Ahmad, and Stephen George. 2005. "Quantifying Customer Response to Dynamic Pricing." *The Electricity Journal* 18, no. 4 (May): 53–63. ISSN: 10406190, accessed September 15, 2023. https://doi.org/10.1016/j.tej.2005.04.005. https://linkinghub. elsevier.com/retrieve/pii/S1040619005000412.
- Faruqui, Ahmad, Ryan Hledik, and Long Lam. 2020. Modernizing Distribution Rate Design. The Brattle Group.
- Faruqui, Ahmad, Ryan Hledik, and John Tsoukalis. 2009. "The Power of Dynamic Pricing." *The Electricity Journal* 22, no. 3 (April): 42–56. ISSN: 10406190, accessed October 16, 2023. https://doi.org/10.1016/j.tej.2009.02.011. https://linkinghub.elsevier.com/retrieve/pii/S1040619009000414.
- Faruqui, Ahmad, and Sanem Sergici. 2010. "Household response to dynamic pricing of electricity: a survey of 15 experiments." *Journal of Regulatory Economics* 38, no. 2 (October): 193–225. ISSN: 0922-680X, 1573-0468, accessed October 16, 2023. https: //doi.org/10.1007/s11149-010-9127-y. http://link.springer.com/10.1007/s11149-010-9127-y.
- Gyamfi, Samuel, Susan Krumdieck, and Tania Urmee. 2013. "Residential peak electricity demand response—Highlights of some behavioural issues." *Renewable and Sustainable Energy Reviews* 25 (September): 71–77. ISSN: 13640321, accessed June 18, 2024. https://doi.org/10.1016/j.rser.2013.04.006. https://linkinghub.elsevier.com/retrieve/pii/S1364032113002578.
- Henderson, J. Stephen. 1983. "The Economics of Electricity Demand Charges." The Energy Journal 4, no. 1 (September 1, 1983). ISSN: 01956574, accessed September 15, 2023. https://doi.org/10.5547/ISSN0195-6574-EJ-Vol4-NoSI-8. http://www.iaee.org/en/publications/ejarticle.aspx?id=932.
- Hledik, Ryan. 2014. "Rediscovering Residential Demand Charges." The Electricity Journal 27, no. 7 (August): 82–96. ISSN: 10406190, accessed September 15, 2023. https://d oi.org/10.1016/j.tej.2014.07.003. https://linkinghub.elsevier.com/retrieve/pii/ S104061901400150X.
- Hledik, Ryan, and Ahmad Faruqui. 2016. "Competing Perspectives on Demand Charges." *Public Utilities Fortnightly*, 20–25.

- Hledik, Ryan, and Gus Greenstein. 2016. "The distributional impacts of residential demand charges." *The Electricity Journal* 29, no. 6 (July): 33–41. ISSN: 10406190, accessed October 16, 2023. https://doi.org/10.1016/j.tej.2016.07.002. https://linkinghub.elsevier.com/retrieve/pii/S1040619016300963.
- Joskow, Paul L, and Catherine D Wolfram. 2012. "Dynamic Pricing of Electricity." Public Utilities Fortnightly 102 (3): 381–385. https://doi.org/10.1257/aer.102.3.381.
- Joskow, Paul L. 2007. "Regulation of Natural Monopoly." In Handbook of law and economics, 2:1227–1348.
- Joskow, Paul L., and Richard Schmalensee. 1983. Markets for Power: An Analysis of Electric Utility Deregulation. Cambridge, MA: Massachusetts Inst. of Technology Press.
- Lazar, Jim. 2016. "Use Great Caution in Design of Residential Demand Charges." Natural Gas & Electricity 32, no. 7 (February): 13–19. ISSN: 15457893, accessed September 15, 2023. https://doi.org/10.1002/gas.21884. https://onlinelibrary.wiley.com/doi/10.1002/ gas.21884.
- McLaren, Joyce, Seth Mullendore, Nicholas Laws, and Kate Anderson. 2017. "Identifying Potential Markets for Behind-the-Meter Battery Energy Storage: A Survey of U.S. Demand Charges."
- Mountain, Dean C., and Cheng Hsiao. 1986. "Peak and Off-Peak Industrial Demand for Electricity: The Hopkinson Rate in Ontario, Canada." *The Energy Journal* 7, no. 1 (January 1, 1986). ISSN: 01956574, accessed September 15, 2023. https://doi.org/10. 5547/ISSN0195-6574-EJ-Vol7-No1-10. http://www.iaee.org/en/publications/ejarticle. aspx?id=1755.
- **MSA.** 2014. Alberta Retail Markets for Electricity and Natural Gas: A description of basic structural features.
- Neufeld, John L. 1987. "Price Discrimination and the Adoption of the Electricity Demand Charge." The Journal of Economic History 47, no. 3 (September): 693–709. ISSN: 0022-0507, 1471-6372, accessed September 15, 2023. https://doi.org/10.1017/S002205070004 9068. https://www.cambridge.org/core/product/identifier/S0022050700049068/type/ journal_article.
- NREL. 2022. Maximum demand charge rates for commercial and industrial electricity tariffs in the United States. Golden, CO (US). https://doi.org/10.7799/1392982. https: //data.nrel.gov/submissions/74.
- Öhrlund, Isak, Mårten Schultzberg, and Cajsa Bartusch. 2019. "Identifying and estimating the effects of a mandatory billing demand charge." *Applied Energy* 237 (March): 885–895. ISSN: 03062619, accessed September 15, 2023. https://doi.org/ 10.1016/j.apenergy.2019.01.028. https://linkinghub.elsevier.com/retrieve/pii/ S0306261919300285.

- Passey, Robert, Navid Haghdadi, Anna Bruce, and Iain MacGill. 2017. "Designing more cost reflective electricity network tariffs with demand charges." *Energy Policy* 109 (October): 642–649. ISSN: 03014215, accessed September 15, 2023. https://doi. org/10.1016/j.enpol.2017.07.045. https://linkinghub.elsevier.com/retrieve/pii/ S0301421517304767.
- Rubin, Scott J. 2015. "Moving Toward Demand-Based Residential Rates." The Electricity Journal 28, no. 9 (November): 63–71. ISSN: 10406190, accessed September 15, 2023. https://doi.org/10.1016/j.tej.2015.09.021. https://linkinghub.elsevier.com/retrieve/ pii/S1040619015002043.
- Seeto, Dewey. 1997. "Time-of-use rates vs. Hopkinson tariffs redux: An analysis of the choice of rate structures in a regulated electricity distribution company." *Energy Economics* 19 (2): 169–185.
- Serecon. 2014. Economic Impact of Agriculture in Lethbridge County.
- Statistics Canada. 2020. Regional unemployment rates used by the Employment Insurance program, three-month moving average, seasonally adjusted. Accessed May 31, 2021. htt ps://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=1410035401.
- Statistics Canada Government of Canada. 2023. "Labour force characteristics by economic region, three-month moving average, unadjusted for seasonality, last 5 months, inactive." Last Modified: 2021-01-08. Accessed October 17, 2023. https://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=1410029301.
- Stokke, Andreas V., Gerard L. Doorman, and Torgeir Ericson. 2010. "An analysis of a demand charge electricity grid tariff in the residential sector." *Energy Efficiency* 3, no. 3 (September): 267–282. ISSN: 1570-646X, 1570-6478, accessed September 15, 2023. https://doi.org/10.1007/s12053-009-9071-9. http://link.springer.com/10.1007/s12053-009-9071-9.
- Taylor, Thomas N., and Peter M. Schwarz. 1986. "A Residential Demand Charge: Evidence from the Duke Power Time-of-Day Pricing Experiment." *The Energy Journal* 7, no. 2 (April 1, 1986). ISSN: 01956574, accessed September 15, 2023. https://doi.org/ 10.5547/ISSN0195-6574-EJ-Vol7-No2-10. http://www.iaee.org/en/publications/ ejarticle.aspx?id=1768.
- Taylor, Thomas N., and Peter M. Schwarz. 1990. "The Long-Run Effects of a Time-of-Use Demand Charge." The RAND Journal of Economics 21 (3): 431. ISSN: 07416261, accessed September 15, 2023. https://doi.org/10.2307/2555618. http://doi.wiley.com/ 10.2307/2555618.

- Tibshirani, Robert. 1996. "Regression Shrinkage and Selection Via the Lasso." Journal of the Royal Statistical Society: Series B (Methodological) 58, no. 1 (January): 267–288. ISSN: 0035-9246, 2517-6161, accessed October 17, 2023. https://doi.org/10.1111/j.2517-6161.1996.tb02080.x. https://rss.onlinelibrary.wiley.com/doi/10.1111/j.2517-6161.1996.tb02080.x.
- Trabish, Herman K. 2018. "Are regulators starting to rethink fixed charges?" Utility Dive. Accessed October 17, 2023. https://www.utilitydive.com/news/are-regulators-startingto-rethink-fixed-charges/530417/.
- Turk, Graham, Tim Schittekatte, Pablo Dueñas Martínez, Paul L Joskow, and Richard L Schmalensee. n.d. "Designing distribution network tariffs under increased residential end-user electrification: Can the US learn something from Europe?"
- Veal, Michael R. 1986. "Time-of use rates and peak period electricity consumption." Energy economics 8 (4): 257–262.
- Wolak, Frank A. 2010. "An Experimental Comparison of Critical Peak and Hourly Pricing: The PowerCentsDC Program."
- Woo, C.K., Brian Horii, and Ira Horowitz. 2002. "The Hopkinson tariff alternative to TOU rates in the Israel Electric Corporation." *Managerial and Decision Economics* 23, no. 1 (January): 9–19. ISSN: 0143-6570, 1099-1468, accessed September 15, 2023. https: //doi.org/10.1002/mde.1040. https://onlinelibrary.wiley.com/doi/10.1002/mde.1040.
- Wood, Lisa, Ross Hemphill, John Howat, Ralph Cavanagh, Severin Borenstein, Jeff Deason, Lisa Schwartz, and Lisa Schwartz. 2016. Recovery of Utility Fixed Costs: Utility, Consumer, Environmental and Economist Perspectives. LBNL-1005742, 1342757. Future Electric Utility Regulation, June 14, 2016. Accessed September 15, 2023. https://doi.org/10.2172/1342757. http://www.osti.gov/servlets/purl/1342757/.
- Zethmayr, Jeff, and Ramandeep Singh Makhija. 2019. "Six unique load shapes: A segmentation analysis of Illinois residential electricity consumers." *The Electricity Journal* 32, no. 9 (November): 106643. ISSN: 10406190, accessed June 18, 2024. https://doi.org/10.1016/j.tej.2019.106643. https://linkinghub.elsevier.com/retrieve/pii/S1040619019301800.

3.8 Appendices

Appendix A: Rate Class

The rate classes in Lethbridge are designed to align the distribution and transmission charges of a customer to the costs he incurred on the grid based on his service of distribution and kVA demand. Since 2013, the City of Lethbridge adopts the following rate classes as shown in Table 6 (City of Lethbridge Electric Utility, 2012 to 2019). 991 customers are residential customers. 992 customers are large residential and small commercial customers. 994 customers are medium-sized commercial customers. 995, 996 and 997 customers are large commercial or industrial customers.

| Rate Code | Rate Distribution Name | Phase Service | Demand Conditions |
|-----------|--|---------------|----------------------------|
| 991 | Standard Single Phase ^{34} | Single | less than 12 kVA $$ |
| 992 | Medium Single Phase | Single | 12 kVA or higher |
| 994 | General Three $Phase^{35}$ | Three | less than 150 kVA $$ |
| 995 | Medium Three Phase | Three | between 150 and 300 kVA $$ |
| 996 | Large Three Phase | Three | 300 kVA or higher |
| 997 | Primary | Primary | less than 6 mVA $$ |

 Table 3.7:
 Rate Classes in Lethbridge

Appendix B: Temperature and Demand

One of the key factors used to predict daily peak demand in our study is the daily temperature, measured by the daily cooling degree days (CDD). To what extent can daily

^{34.} Single Phase service uses two wire Alternating Current (AC) power circuit. It is the most common household power circuit that powers lights, TV, etc.

^{35.} Three Phase service uses three wire Alternating Current (AC) power circuit. Most commercial buildings use this power arrangement because it provides higher power density and flexibility than a single phase service.



Figure 3.4: Relationship Between Temperature and Demand for a Summer-Peak Customer

temperature determine daily peak demand is key in determining the performance of our prediction model. As an illustration, Figure 3.4 plots the relationship between daily maximum temperature and daily maximum demand for a typical summer-peak customer. This section investigates the relationship between temperature and demand by using a simple fixed effects model:

$$D_{it} = \gamma_1 CDD_t + \gamma_2 CDD_t^2 + \beta X + \varepsilon_{it}$$
(3.5)

where D_{it} is the daily maximum demand, CDD_t is the daily cooling degree days, and X includes dummy variables for weekends and holidays. The results are presented in Table 3.8. Columns (1) and (2) show the estimation results for summer-peak and non-summer-peak 995 customers, respectively. For summer-peak customers, a one-degree Celsius increase in CDD will increase daily peak demand by 21.4 kVA, or approximately 7.5% of the average daily peak demand. The relatively high within R-square (0.3371) for summer-peak 995 customers suggests that variations in peak demand can largely be explained by variations in CDD. For non-summer-peak 995 customers, a one-degree increase in CDD will increase average daily peak demand by 15.6 kVA. However, the relatively low R-square suggests that other factors, which might not easily be controlled, can significantly affect daily peak demand. Similarly, for 996 customers, an increase in CDD will significantly increase daily peak demand, but the effects are smaller in percentage terms compared to 995 customers. Moreover, the relatively low R-square suggests that variation in CDD alone can only capture a small portion of variations in the daily peak demand. These results suggest that our prediction model will be more accurate for summer-peak 995 consumers than for 996 consumers. Therefore, the main findings in our case study could be more robust for 995 consumers.

| | (1) 995 | (2) 995 | (3) 996 | (4) 996 |
|-----------------------|------------|------------|------------|------------|
| CDD | 21.42*** | 15.58*** | 24.70*** | 18.00** |
| | (3.08) | (3.55) | (-1.54) | (7.31) |
| CDD^2 | -1.44*** | -1.76*** | -1.54*** | -1.48** |
| | (0.39) | (0.37) | (0.20) | (0.58) |
| Weekend | -17.48*** | -127.38*** | -25.97* | -121.54*** |
| | (5.03) | (31.90) | (13.63) | (31.70) |
| Holiday | -9.04 | -102.87*** | -25.12 | -118.05*** |
| | (5.85) | (19.60) | (15.03) | (32.13) |
| Constant | 222.62*** | 314.90*** | 466.18*** | 547.00*** |
| | (2.65) | (5.12) | (3.43) | (7.94) |
| Summer-Peak Customers | Yes | No | Yes | No |
| Observations | 2,186 | 4,583 | 6,624 | 7,728 |
| R-square Within | 0.3371 | 0.0741 | 0.0802 | 0.0470 |
| R-square Between | 0.0350 | 0.0281 | 0.0000 | 0.0000 |
| R-square Overall | 0.1202 | 0.0436 | 0.0037 | 0.0092 |

Table 3.8: Daily Maximum Temperature vs. Daily Maximum Demand

Notes: This table presents the estimation results of Equation (3.5) for 995 and 996 customers. The dependent variable is the daily peak kVA. CDD is the daily cooling degree days, measured in degree Celsius. Std. Dev. in parentheses^{*} p < 0.05, ^{**} p < 0.01, ^{***} p < 0.001.

Appendix C: Fixed Effects Approach

This section reports the estimation results for the fixed-effects approach. Similar to the machine-learning approach described in Section 3.5.2, the fixed-effects approach utilizes the

fact that MDCs are based solely on the 12-month rolling maximum demand. Therefore, after controlling for a set of covariates such as temperature, weekends, holidays, and economic conditions, the differences in daily maximum demand across years should be insignificant for periods when the annual maximum demand is unlikely to occur. Formally, we estimated the following fixed-effects model:

$$D_{it} = \sum_{q}^{5} \gamma_{q} Q_{qt} + \sum_{q}^{5} \delta_{q} Q_{qt} * Year_{2018} + Year_{2018} + X_{it} + \varepsilon_{it}$$
(3.6)

where D_{it} is the the logged daily maximum demand measured in kVA. Q_{qt} is the factorial variable for the *q*th temperature quantile. $Year_{2018}$ is the dummy variable for the year 2018. X includes a set of covariates such as daily cooling degree days, time dummy variables for weekend and holidays, and local economic conditions (commodity prices and labor market statistics). The difference in daily maximum demand across years for different quantiles are captured by δ_q . If customers respond to the demand change, δ_5 should be significant and differ from zero. For non-summer customers, we expect all δ_q to be close to zero.

The estimation results are presented in Table 3.9. Consistent with our benchmark results obtained using the machine-learning approach, the average daily maximum demand in the 5th quantile is significant and is 5.7% higher in 2018 compared to 2017 for summer-peak 995 customers. This suggests that customers increase their daily maximum demand when facing a lower MDC rate. In contrast, for non-summer-peak customers, the daily maximum demand remains similar across years. In the machine-learning approach, for 996 customers, the differences in daily maximum demand across years are insignificant at the 5th quantile. However, the fixed effects approach shows that 996 customers exhibit a similar increase in demand when facing a lower MDC rate. Notably, as illustrated in Table 3.9, aside from the 5th quantile, the 4th quantile for summer-peak 995 customers and non-summer-peak 996 customers is also significant. This underscores the heightened sensitivity of the fixed effects model compared to the machine-learning approach, as discussed in (Burlig et al. 2020). For this reason, we take the machine-learning results as the benchmark results in our study.

| | (1) | (2) | (3) | (4) |
|-----------------------|-----------|----------|-----------|----------|
| | log kVA | log kVA | log kVA | log kVA |
| 5th Quantile * 2018 | 0.057** | -0.000 | 0.034* | -0.043 |
| | (0.021) | (0.033) | (0.020) | (0.030) |
| 4th Quantile * 2018 | 0.041* | -0.010 | 0.016 | -0.056* |
| | (0.021) | (0.033) | (0.019) | (0.030) |
| 3rd Quantile * 2018 | 0.020 | -0.011 | -0.004 | -0.037 |
| | (0.180) | (0.046) | (0.018) | (0.027) |
| 2nd Quantile * 2018 | 0.011 | -0.016 | -0.010 | -0.045 |
| | (0.159) | (0.057) | (0.022) | (0.039) |
| 5th Quantile | 0.490*** | 1.202*** | 0.424*** | 0.836*** |
| | (0.060) | (0.144) | (0.074) | (0.115) |
| 4th Quantile | 0.390*** | 1.038*** | 0.357*** | 0.752*** |
| | (0.053) | (0.139) | (0.068) | (0.116) |
| 3rd Quantile | 0.309*** | 0.919*** | 0.291*** | 0.658*** |
| | (0.047) | (0.138) | (0.060) | (0.108) |
| 2nd Quantile | 0.288*** | 0.875*** | 0.237*** | 0.598*** |
| | (0.032) | (0.121) | (0.053) | (0.088) |
| Year (2018) | -0.109*** | -0.001 | -0.085*** | 0.027 |
| | (0.023) | (0.037) | (0.021) | (0.032) |
| Rate Code | 995 | 995 | 996 | 996 |
| Summer-Peak Customers | Yes | No | Yes | No |
| Observations | 3,246 | 4,261 | 6,502 | 8,320 |

Table 3.9: Fixed Effects Approach Results

Notes: This table presents the estimation results of Equation (3.6) for 995 and 996 customers. The 2nd, 3rd, 4th, and 5th quantiles represent different quantiles in daily maximum temperatures, with the 5th quantile indicating the hottest 20% days. Std. Dev. in parentheses* p < 0.05, ** p < 0.01, *** p < 0.001.

Appendix D: Load Profiles

In our sample, differences in load profiles across customers are substantial, making it challenging to create a representative load profile. Figure 3.5 shows the load profile for a randomly selected customer during the summer months of 2017 and 2018.



Figure 3.5: Load Profile of a Random 995 Customer

Appendix E: System Maximum Demand and Customer-Specific Maximum Demand

This section explores the correlation between customer-specific maximum demand and Alberta's system maximum demand. First, we examine the correlations between customer-specific load and system load. For 995 customers, the correlation ranges from 0.07 to 0.71, with an average of 0.39. For 996 customers, the correlation ranges from -0.27 to 0.72, averaging at 0.28.

During our sample period, Alberta's system load peaked in February. Figures 3.6 and 3.7 present the relative frequency of when the annual system peak for a customer occurs in



Figure 3.6: 995 Customers Annual Peaks by Month

a month. As shown in Figure 3.6, the 995 customers' annual peaks can occur during both summer and winter. In contrast, the 996 customers tend to have their annual peaks in the summer.

Figure 3.8 presents the relative frequency of daily maximum demand occurring at each hour. As shown, the system maximum demand tends to peak at 4 PM and 5 PM. In contrast, as shown in Figures 3.9 and 3.10, both 995 and 996 customers tend to peak during the midday hours.

In summary, there appears to be a relatively low or moderate degree of correlation between the maximum demand of the system and the maximum demand specific to the customer. Therefore, MDCs are less efficient in reducing system maximum demand if they primarily target customer-specific maximum demand in Alberta.



Figure 3.7: 996 Customers Annual Peaks by Month



Figure 3.8: Daily System Peaks by Hour



Figure 3.9: 995 Customers Daily Peaks by Hour



Figure 3.10: 996 Customers Daily Peaks by Hour

Bibliography

- Abdelmotteleb, Ibtihal, Tomás Gómez, José Pablo Chaves Ávila, and Javier Reneses. 2018. "Designing efficient distribution network charges in the context of active customers." *Applied Energy* 210 (January): 815–826. ISSN: 03062619, accessed April 11, 2024. https://doi.org/10.1016/j.apenergy.2017.08.103. https://linkinghub.elsevier. com/retrieve/pii/S0306261917311236.
- AESO. 2023. AESO 2022 Annual Market Statistics.
- Ahrens, Achim, Christian B. Hansen, and Mark E. Schaffer. 2020. "lassopack: Model selection and prediction with regularized regression in Stata." *The Stata Journal: Promoting communications on statistics and Stata* 20, no. 1 (March): 176–235. ISSN: 1536-867X, 1536-8734, accessed October 17, 2023. https://doi.org/10.1177/ 1536867X20909697. http://journals.sagepub.com/doi/10.1177/1536867X20909697.
- Alberta Government. 2018. *Highlights of the Alberta Economy 2017.* chrome-extension: //efaidnbmnnibpcajpcglclefindmkaj/https://open.alberta.ca/dataset/10989a51-f3c2-4dcb-ac0f-f07ad88f9b3b/resource/513eef5f-aa53-4cde-888d-8e52822b6db4/download/sp-eh-highlightsabeconomypresentation.pdf.
- Alberta Utilities Commission. 2021. Distribution System Inquiry Final Report. Alberta Utilities Commission.
- Andruszkiewicz, Jerzy, Józef Lorenc, and Agnieszka Weychan. 2019. "Demand Price Elasticity of Residential Electricity Consumers with Zonal Tariff Settlement Based on Their Load Profiles." *Energies* 12, no. 22 (November 13, 2019): 4317. ISSN: 1996-1073, accessed June 2, 2024. https://doi.org/10.3390/en12224317. https://www.mdpi.com/ 1996-1073/12/22/4317.
- Antweiler, Werner, and Felix Muesgens. 2021. "On the long-term merit order effect of renewable energies." *Energy Economics* 99 (July): 105275. ISSN: 01409883, accessed June 2, 2024. https://doi.org/10.1016/j.eneco.2021.105275. https://linkinghub.elsevier. com/retrieve/pii/S0140988321001808.
- Arciniegas, Laura M., and Eric Hittinger. 2018. "Tradeoffs between revenue and emissions in energy storage operation." *Energy* 143 (January): 1–11. ISSN: 03605442, accessed June 2, 2024. https://doi.org/10.1016/j.energy.2017.10.123. https://linkinghub.elsevier.com/retrieve/pii/S0360544217318145.
- Astoria, Ross. 2017. "On the radicality of New York's Reforming the Energy Vision." *The Electricity Journal* 30, no. 5 (June): 54–58. ISSN: 10406190, accessed June 2, 2024. https://doi.org/10.1016/j.tej.2017.04.018. https://linkinghub.elsevier.com/retrieve/ pii/S1040619017301070.
- Bartusch, Cajsa, and Karin Alvehag. 2014. "Further exploring the potential of residential demand response programs in electricity distribution." Applied Energy 125 (July): 39–59. ISSN: 03062619, accessed September 15, 2023. https://doi.org/10.1016/j.apener gy.2014.03.054. https://linkinghub.elsevier.com/retrieve/pii/S0306261914002943.
- Bartusch, Cajsa, Fredrik Wallin, Monica Odlare, Iana Vassileva, and Lars Wester. 2011. "Introducing a demand-based electricity distribution tariff in the residential sector: Demand response and customer perception." *Energy Policy* 39, no. 9 (September): 5008– 5025. ISSN: 03014215, accessed September 15, 2023. https://doi.org/10.1016/j.enpol. 2011.06.013. https://linkinghub.elsevier.com/retrieve/pii/S030142151100468X.
- Baumol, William J., and Robert D. Willig. 1981. "Fixed Costs, Sunk Costs, Entry Barriers, and Sustainability of Monopoly." *The Quarterly Journal of Economics* 96, no. 3 (August): 405. ISSN: 00335533, accessed April 7, 2024. https://doi.org/10.2307/1882680. https://academic.oup.com/qje/article-lookup/doi/10.2307/1882680.
- Belloni, A., D. Chen, V. Chernozhukov, and C. Hansen. 2012. "Sparse Models and Methods for Optimal Instruments With an Application to Eminent Domain." *Econometrica* 80 (6): 2369–2429. ISSN: 0012-9682, accessed June 10, 2024. https://doi.org/ 10.3982/ECTA9626. http://doi.wiley.com/10.3982/ECTA9626.
- Beppler, Ross C., Daniel C. Matisoff, and Matthew E. Oliver. 2023. "Electricity consumption changes following solar adoption: Testing for a solar rebound." *Economic Inquiry* 61, no. 1 (January): 58–81. ISSN: 0095-2583, 1465-7295, accessed June 2, 2024. https://doi.org/10.1111/ecin.13031. https://onlinelibrary.wiley.com/doi/10.1111/ecin. 13031.
- Berg, Sandford, and Andreas Savvides. 1983. "The theory of maximum kW demand charges for electricity." *Energy Economics* 5, no. 4 (October): 258–266. ISSN: 01409883, accessed September 15, 2023. https://doi.org/10.1016/0140-9883(83)90030-0. https://linkinghub.elsevier.com/retrieve/pii/0140988383900300.

- Biggar, Darryl. 2009. "Is Protecting Sunk Investments by Consumers a Key Rationale for Natural Monopoly Regulation?" *Review of Network Economics* 8, no. 2 (January 1, 2009). ISSN: 1446-9022, 2194-5993, accessed October 30, 2023. https://doi.org/10. 2202/1446-9022.1173. https://www.degruyter.com/document/doi/10.2202/1446-9022.1173/html.
- Bilich, Andy, Elisheba Spiller, and James Fine. 2019. "Proactively planning and operating energy storage for decarbonization: Recommendations for policymakers." *Energy Policy* 132 (September): 876–880. ISSN: 03014215, accessed June 2, 2024. https://doi.org/10.1016/j.enpol.2019.06.033. https://linkinghub.elsevier.com/retrieve/pii/S0301421519304033.
- Bird, Lori, Carolyn Davidson, Joyce McLaren, and John Miller. 2015. "Impact of Rate Design Alternatives on Residential Solar Customer Bills: Increased Fixed Charges, Minimum Bills and Demand-Based Rates." *Renewable Energy*.
- Boampong, Richard, and David P. Brown. 2020. "On the benefits of behind-the-meter rooftop solar and energy storage: The importance of retail rate design." *Energy Economics* 86 (February): 104682. ISSN: 01409883, accessed October 17, 2023. https://doi.org/10.1016/j.eneco.2020.104682. https://linkinghub.elsevier.com/retrieve/pii/S0140988320300219.
- Borenstein, Severin. 2005. "The Long-Run Efficiency of Real-Time Electricity Pricing." The Energy Journal 26 (3): 93–116.
- Borenstein, Severin. 2016. "The economics of fixed cost recovery by utilities." The Electricity Journal 29, no. 7 (September): 5–12. ISSN: 10406190, accessed September 15, 2023. https://doi.org/10.1016/j.tej.2016.07.013. https://linkinghub.elsevier.com/retrieve/pii/S1040619016301130.
- Borenstein, Severin. 2017. "Private Net Benefits of Residential Solar PV: The Role of Electricity Tariffs, Tax Incentives, and Rebates." Journal of the Association of Environmental and Resource Economists 4 (S1): S85–S122. ISSN: 2333-5955, 2333-5963, accessed June 2, 2024. https://doi.org/10.1086/691978. https://www.journals.uchicago.edu/ doi/10.1086/691978.
- Borenstein, Severin, and James B. Bushnell. 2022. "Do Two Electricity Pricing Wrongs Make a Right? Cost Recovery, Externalities, and Efficiency." American Economic Journal: Economic Policy 14, no. 4 (November 1, 2022): 80–110. ISSN: 1945-7731, 1945-774X, accessed October 30, 2023. https://doi.org/10.1257/pol.20190758. https: //pubs.aeaweb.org/doi/10.1257/pol.20190758.
- Borenstein, Severin, Meredith Fowlie, and James Sallee. 2021. Designing Electricity Rates for An Equitable Energy Transition. Next 10. Accessed October 23, 2023.

- Bowen, Thomas, Sam Koebrich, Kevin McCabe, and Ben Sigrin. 2022. "The locational value of distributed energy resources: A parcel-level evaluation of solar and wind potential in New York state." *Energy Policy* 166 (July): 112744. ISSN: 03014215, accessed June 10, 2024. https://doi.org/10.1016/j.enpol.2021.112744. https://linkinghub.elsevier.com/retrieve/pii/S0301421521006108.
- Brown, David P., and David E. M. Sappington. 2017a. "Optimal policies to promote efficient distributed generation of electricity." *Journal of Regulatory Economics* 52, no. 2 (October): 159–188. ISSN: 0922-680X, 1573-0468, accessed June 10, 2024. https://doi.org/10.1007/s11149-017-9335-9. http://link.springer.com/10.1007/s11149-017-9335-9.
- Brown, David P., and David E.M. Sappington. 2017b. "Designing Compensation for Distributed Solar Generation: Is Net Metering Ever Optimal?" *The Energy Journal* 38, no. 3 (May): 1–32. ISSN: 0195-6574, 1944-9089, accessed June 2, 2024. https://doi.org/ 10.5547/01956574.38.3.dbro. https://journals.sagepub.com/doi/10.5547/01956574.38. 3.dbro.
- Brown, David P., and David E.M. Sappington. 2018. "On the role of maximum demand charges in the presence of distributed generation resources." *Energy Economics* 69 (January): 237–249. ISSN: 01409883, accessed September 15, 2023. https://doi.org/10.1016/ j.eneco.2017.11.023. https://linkinghub.elsevier.com/retrieve/pii/S0140988317304152.
- Brown, David P., Jay Zarnikau, and Chi-Keung Woo. 2020. "Does locational marginal pricing impact generation investment location decisions? An analysis of Texas's wholesale electricity market." *Journal of Regulatory Economics* 58, no. 2 (December): 99–140. ISSN: 0922-680X, 1573-0468, accessed June 10, 2024. https://doi.org/10.1007/s11149-020-09413-0. https://link.springer.com/10.1007/s11149-020-09413-0.
- Brown, Toby, Ahmad Faruqui, and Léa Grausz. 2015. "Efficient tariff structures for distribution network services." *Economic Analysis and Policy* 48 (December): 139–149. ISSN: 03135926, accessed October 17, 2023. https://doi.org/10.1016/j.eap.2015.11.010. https://linkinghub.elsevier.com/retrieve/pii/S0313592615300552.
- **BTU Analytics.** 2023. "Wind and Solar Curtailments in ERCOT." Accessed April 30, 2023. https://btuanalytics.com/power-and-renewables/wind-and-solar-curtailments-in-ercot/.
- Burger, Scott P., Christopher R. Knittel, Ignacio J. Perez-Arriaga, Ian Schneider, and Frederik Vom Scheidt. 2020. "The Efficiency and Distributional Effects of Alternative Residential Electricity Rate Designs." *The Energy Journal* 41, no. 1 (January): 199–240. ISSN: 0195-6574, 1944-9089, accessed April 5, 2024. https://doi.org/10.5547/ 01956574.41.1.sbur. https://journals.sagepub.com/doi/10.5547/01956574.41.1.sbur.

- Burke, Paul J., and Ashani Abayasekara. 2018. "The Price Elasticity of Electricity Demand in the United States: A Three-Dimensional Analysis." *The Energy Journal* 39, no. 2 (March): 123–146. ISSN: 0195-6574, 1944-9089, accessed June 2, 2024. https://doi.org/10.5547/01956574.39.2.pbur. https://journals.sagepub.com/doi/10.5547/01956574.39.2.pbur.
- Burlig, Fiona, Christopher Knittel, David Rapson, Mar Reguant, and Catherine Wolfram. 2020. "Machine Learning from Schools about Energy Efficiency." Journal of the Association of Environmental and Resource Economists 7 (6): 1181–1217.
- CAISO. 2023a. 2022 Annual Report On Market Issues and Performance.
- CAISO. 2023b. Wind and Solar Summary. Accessed May 5, 2023. http://oasis.caiso.com/ mrioasis/logon.do.
- Callaway, Duncan S., Meredith Fowlie, and Gavin McCormick. 2018. "Location, Location, Location: The Variable Value of Renewable Energy and Demand-Side Efficiency Resources." Journal of the Association of Environmental and Resource Economists 5, no. 1 (January): 39–75. ISSN: 2333-5955, 2333-5963, accessed June 2, 2024. https://doi.org/10.1086/694179. https://www.journals.uchicago.edu/doi/10.1086/694179.
- **CARB.** 2022. Cap-and-Trade Program Data. Accessed May 31, 2023. https://ww2.arb.ca. gov/our-work/programs/cap-and-trade-program/cap-and-trade-program-data.
- Cardoso, G., M. Stadler, S. Mashayekh, and E. Hartvigsson. 2017. "The impact of ancillary services in optimal DER investment decisions." *Energy* 130 (July): 99–112. ISSN: 03605442, accessed June 2, 2024. https://doi.org/10.1016/j.energy.2017.04.124. https://linkinghub.elsevier.com/retrieve/pii/S0360544217306904.
- Caves, Douglas W., Laurits R. Christensen, and Joseph A. Herriges. 1984. "Modelling alternative residential peak-load electricity rate structures." *Journal of Econometrics* 24, no. 3 (March): 249–268. ISSN: 03044076, accessed September 25, 2023. https://doi.org/10.1016/0304-4076(84)90052-6. https://linkinghub.elsevier.com/retrieve/pii/0304407684900526.
- Chang, Yoosoon, Yongok Choi, and Joon Y. Park. 2017. "A new approach to model regime switching." *Journal of Econometrics* 196, no. 1 (January): 127–143. ISSN: 03044076, accessed June 10, 2024. https://doi.org/10.1016/j.jeconom.2016.09.005. https://linkinghub.elsevier.com/retrieve/pii/S0304407616301671.
- **Choose Lethbridge.** 2020. "Agriculture | Economic Development Lethbridge." Accessed October 17, 2023. https://chooselethbridge.ca/agriculture.
- City of Lethbridge. 2023. "Utility rates." City of Lethbridge. Accessed October 17, 2023. https://www.lethbridge.ca/utilities/utility-rates/.
- CPUC. 2017. 2016 SGIP Advanced Energy Storage Impact Evaluation.

- CPUC. 2018a. 2017 SGIP Advanced Energy Storage Impact Evaluation.
- CPUC. 2018b. Revised Self-Generation Incentive Program Greenhouse Gas Staff Proposal.
- **CPUC.** 2022. "Customer-Sited Renewable Energy Generation." Accessed April 30, 2022. https://www.cpuc.ca.gov/industries-and-topics/electrical-energy/demand-side-management/net-energy-metering.
- **CPUC.** 2023. Self-Generation Incentive Program Handbook.
- Darby, Sarah J., and Eoghan McKenna. 2012. "Social implications of residential demand response in cool temperate climates." *Energy Policy* 49 (October): 759–769. ISSN: 03014215, accessed June 18, 2024. https://doi.org/10.1016/j.enpol.2012.07.026. https://linkinghub.elsevier.com/retrieve/pii/S0301421512006076.
- Darghouth, Naim, Galen Barbose, Andrew Mills, Ryan Wiser, Pieter Gagnon, and Lori Bird. 2017. Exploring Demand Charge Savings from Residential Solar. LBNL– 1007030, 1373280. NREL, July 31, 2017. Accessed October 17, 2023. https://doi.org/ 10.2172/1373280. http://www.osti.gov/servlets/purl/1373280/.
- Deniel, Joe, and Sam Gomberg. 2021. Why Does Wind Energy Get Wasted? The Myth of Wind Energy Oversupply and the True Root Causes of Wind Curtailment. Cambridge, MA (United States): Union of Concerned Scientists. https://www.ourenergypolicy.org/ resources/why-does-wind-energy-get-wasted-the-myth-of-wind-energy-oversupplyand-the-true-root-causes-of-wind-curtailment/.
- Deru, M., K. Field, D. Studer, K. Benne, B. Griffith, P. Torcellini, B. Liu, et al. 2011. U.S. Department of Energy Commercial Reference Building Models of the National Building Stock. NREL/TP-5500-46861, 1009264. February 1, 2011. Accessed June 6, 2024. https://doi.org/10.2172/1009264. http://www.osti.gov/servlets/purl/1009264pitlfN/.
- Divya, K.C., and Jacob Østergaard. 2009. "Battery energy storage technology for power systems—An overview." *Electric Power Systems Research* 79, no. 4 (April): 511–520. ISSN: 03787796, accessed June 2, 2024. https://doi.org/10.1016/j.epsr.2008.09.017. https://linkinghub.elsevier.com/retrieve/pii/S0378779608002642.
- Dutta, Goutam, and Krishnendranath Mitra. 2017. "A literature review on dynamic pricing of electricity." Journal of the Operational Research Society 68, no. 10 (October): 1131–1145. ISSN: 0160-5682, 1476-9360, accessed October 17, 2023. https://doi.org/10. 1057/s41274-016-0149-4. https://www.tandfonline.com/doi/full/10.1057/s41274-016-0149-4.
- **EIA.** 2021a. "California's curtailments of solar electricity generation continue to increase." Accessed April 30, 2022. https://www.eia.gov/todayinenergy/detail.php?id=49276.
- EIA. 2021b. Price Elasticity for Energy Use in Buildings in the United States.

- **EIA.** 2023. *EIA-860 Suevey*. Accessed May 31, 2023. https://www.eia.gov/electricity/data/eia860/.
- ElectricityMatch. 2023. "Texas Electricity Providers." Accessed April 30, 2023. https://electricitymatch.com/providers/.
- Elmallah, Salma, Anna M Brockway, and Duncan Callaway. 2022. "Can distribution grid infrastructure accommodate residential electrification and electric vehicle adoption in Northern California?" *Environmental Research: Infrastructure and Sustainability* 2, no. 4 (December 1, 2022): 045005. ISSN: 2634-4505, accessed April 12, 2024. https://doi.org/10.1088/2634-4505/ac949c. https://iopscience.iop.org/article/10.1088/2634-4505/ac949c.
- Energy Information Administration. 2018. "Use of energy in commercial buildings." Accessed October 17, 2023. https://www.eia.gov/energyexplained/use-of-energy/ commercial-buildings.php.
- Energy Sage. 2023a. "Buying solar panels in 2023: The complete guide." Accessed August 31, 2023. https://www.energysage.com/solar/.
- Energy Sage. 2023b. "How much do storage systems cost in Texas in 2023?" Accessed May 19, 2023. https://www.energysage.com/local-data/energy-storage-cost/tx/.
- EnerNOC. 2013. ENERNOC 2012 Commercial Energy Consumption Data. Accessed April 30, 2023. https://trynthink.github.io/buildingsdatasets/show.html?title_id=enernoc-2012-commercial-energy-consumption-data.
- **EPA.** 2023a. Continuous Emissions Monitoring System. Accessed May 5, 2023. https://www.epa.gov/emc/emc-continuous-emission-monitoring-systems.
- **EPA.** 2023b. Tailpipe Greenhouse Gas Emissions from a Typical Passenger Vehicle.
- **EPA.** 2024. Inventory of U.S. Greenhouse Gas Emissions and Sinks 1990-2022. Washington, D.C. (United States): United States Environmental Protection Agency.
- **ERCOT.** 2014. The Competitive Renewable Energy Zones Process.
- ERCOT. 2022. ERCOT Fuel Mix Report. Accessed May 5, 2024. https://www.ercot.com/ gridinfo/generation.
- ERCOT. 2023a. Fuel Mix Report. Accessed May 5, 2023. https://www.ercot.com/gridinfo/generation.
- **ERCOT.** 2023b. Report on Existing and Potential Electric System Constraints and Needs.
- ERCOT. 2024. Hourly Load Data Archives. Accessed May 31, 2024.

- Fabra, Natalia, and Mar Reguant. 2014. "Pass-Through of Emissions Costs in Electricity Markets." American Economic Review 104, no. 9 (September 1, 2014): 2872–2899. ISSN: 0002-8282, accessed June 2, 2024. https://doi.org/10.1257/aer.104.9.2872. https: //pubs.aeaweb.org/doi/10.1257/aer.104.9.2872.
- Fan, Cheng, Fu Xiao, and Shengwei Wang. 2014. "Development of prediction models for next-day building energy consumption and peak power demand using data mining techniques." Applied Energy 127 (August): 1–10. ISSN: 03062619, accessed October 17, 2023. https://doi.org/10.1016/j.apenergy.2014.04.016. https://linkinghub.elsevier. com/retrieve/pii/S0306261914003596.
- Faruqui, Ahmad, and Geronimo Aydin. 2017. "Moving Forward with Electricity Tariff Reform." Cato Institute, September 25, 2017. Accessed October 17, 2023. https://www. cato.org/regulation/fall-2017/moving-forward-electricity-tariff-reform.
- Faruqui, Ahmad, and Stephen George. 2005. "Quantifying Customer Response to Dynamic Pricing." *The Electricity Journal* 18, no. 4 (May): 53–63. ISSN: 10406190, accessed September 15, 2023. https://doi.org/10.1016/j.tej.2005.04.005. https://linkinghub. elsevier.com/retrieve/pii/S1040619005000412.
- Faruqui, Ahmad, Ryan Hledik, and Long Lam. 2020. Modernizing Distribution Rate Design. The Brattle Group.
- Faruqui, Ahmad, Ryan Hledik, and John Tsoukalis. 2009. "The Power of Dynamic Pricing." *The Electricity Journal* 22, no. 3 (April): 42–56. ISSN: 10406190, accessed October 16, 2023. https://doi.org/10.1016/j.tej.2009.02.011. https://linkinghub.elsevier.com/retrieve/pii/S1040619009000414.
- Faruqui, Ahmad, and Sanem Sergici. 2010. "Household response to dynamic pricing of electricity: a survey of 15 experiments." *Journal of Regulatory Economics* 38, no. 2 (October): 193–225. ISSN: 0922-680X, 1573-0468, accessed October 16, 2023. https: //doi.org/10.1007/s11149-010-9127-y. http://link.springer.com/10.1007/s11149-010-9127-y.
- Felder, Frank A., and Rasika Athawale. 2014. "The Life and Death of the Utility Death Spiral." The Electricity Journal 27, no. 6 (July): 9–16. ISSN: 10406190, accessed June 5, 2024. https://doi.org/10.1016/j.tej.2014.06.008. https://linkinghub.elsevier.com/ retrieve/pii/S1040619014001407.
- Feldman, David, Vignesh Ramasamy, Ran Fu, Ashwin Ramdas, Jal Desai, and Robert Margolis. 2021. U.S. Solar Photovoltaic System and Energy Storage Cost Benchmark (Q1 2020). NREL/TP-6A20-77324, 1764908, MainId:26270. January 1, 2021. Accessed June 6, 2024. https://doi.org/10.2172/1764908. https://www.osti. gov/servlets/purl/1764908/.

- Fell, Harrison, Daniel T. Kaffine, and Kevin Novan. 2021. "Emissions, Transmission, and the Environmental Value of Renewable Energy." *American Economic Journal: Economic Policy* 13, no. 2 (May 1, 2021): 241–272. ISSN: 1945-7731, 1945-774X, accessed June 10, 2024. https://doi.org/10.1257/pol.20190258. https://pubs.aeaweb.org/doi/10. 1257/pol.20190258.
- **FERC.** 2021. "FERC Order No. 2222 Explainer: Facilitating Participation in Electricity Markets by Distributed Energy Resources." Accessed April 30, 2022. https://www. ferc.gov/ferc-order-no-2222-explainer-facilitating-participation-electricity-marketsdistributed-energy.
- Fisher, Michael J., and Jay Apt. 2017. "Emissions and Economics of Behind-the-Meter Electricity Storage." *Environmental Science & Technology* 51, no. 3 (February 7, 2017): 1094–1101. ISSN: 0013-936X, 1520-5851, accessed June 5, 2024. https://doi.org/10. 1021/acs.est.6b03536. https://pubs.acs.org/doi/10.1021/acs.est.6b03536.
- Gentle, Jake, Kurt S Myers, Thomas Baldwin, Isaac West, Kenyon Hart, Bruce Savage, Mike Ellis, and Phil Anderson. 2012. "Concurrent Wind Cooling in Power Transmission Lines."
- Gianfreda, Angelica, and Luigi Grossi. 2012. "Forecasting Italian electricity zonal prices with exogenous variables." *Energy Economics* 34, no. 6 (November): 2228–2239. ISSN: 01409883, accessed June 10, 2024. https://doi.org/10.1016/j.eneco.2012.06.024. https://linkinghub.elsevier.com/retrieve/pii/S0140988312001338.
- Graff Zivin, Joshua S., Matthew J. Kotchen, and Erin T. Mansur. 2014. "Spatial and temporal heterogeneity of marginal emissions: Implications for electric cars and other electricity-shifting policies." *Journal of Economic Behavior & Organization* 107 (November): 248–268. ISSN: 01672681, accessed June 5, 2024. https://doi.org/10.1016/ j.jebo.2014.03.010. https://linkinghub.elsevier.com/retrieve/pii/S0167268114000808.
- Green, Richard, Iain Staffell, and Nicholas Vasilakos. 2014. "Divide and conquer? Assessing k-means clustering of demand data in simulations of the British electricity system." *IEEE Transactions on Engineering Management* 61 (2): 251–260.
- Gyamfi, Samuel, Susan Krumdieck, and Tania Urmee. 2013. "Residential peak electricity demand response—Highlights of some behavioural issues." *Renewable and Sustainable Energy Reviews* 25 (September): 71–77. ISSN: 13640321, accessed June 18, 2024. https://doi.org/10.1016/j.rser.2013.04.006. https://linkinghub.elsevier.com/retrieve/pii/S1364032113002578.
- Henderson, J. Stephen. 1983. "The Economics of Electricity Demand Charges." The Energy Journal 4, no. 1 (September 1, 1983). ISSN: 01956574, accessed September 15, 2023. https://doi.org/10.5547/ISSN0195-6574-EJ-Vol4-NoSI-8. http://www.iaee.org/en/publications/ejarticle.aspx?id=932.

- Hitaj, Claudia. 2015. "Location matters: The impact of renewable power on transmission congestion and emissions." *Energy Policy* 86 (November): 1–16. ISSN: 03014215, accessed June 10, 2024. https://doi.org/10.1016/j.enpol.2015.06.019. https://linkinghub.elsevier. com/retrieve/pii/S0301421515002372.
- Hittinger, Eric S., and Inês M. L. Azevedo. 2015. "Bulk Energy Storage Increases United States Electricity System Emissions." *Environmental Science & Technology* 49, no. 5 (March 3, 2015): 3203–3210. ISSN: 0013-936X, 1520-5851, accessed June 5, 2024. https://doi.org/10.1021/es505027p. https://pubs.acs.org/doi/10.1021/es505027p.
- Hledik, Ryan. 2014. "Rediscovering Residential Demand Charges." The Electricity Journal 27, no. 7 (August): 82–96. ISSN: 10406190, accessed September 15, 2023. https://d oi.org/10.1016/j.tej.2014.07.003. https://linkinghub.elsevier.com/retrieve/pii/ S104061901400150X.
- Hledik, Ryan, and Ahmad Faruqui. 2016. "Competing Perspectives on Demand Charges." *Public Utilities Fortnightly*, 20–25.
- Hledik, Ryan, and Gus Greenstein. 2016. "The distributional impacts of residential demand charges." The Electricity Journal 29, no. 6 (July): 33–41. ISSN: 10406190, accessed October 16, 2023. https://doi.org/10.1016/j.tej.2016.07.002. https://linkinghub. elsevier.com/retrieve/pii/S1040619016300963.
- Hochul, Kathy, and Basil Seggos. 2023. Establishing a Value of Carbon. New York, NY (United States): New York State Department of Environmental Conservation.
- Holland, Stephen P, and Erin T Mansur. 2008. "Is Real-Time Pricing Green? The Environmental Impacts of Electricity Demand Variance." The Review of Economics and Statistics.
- Holland, Stephen P., Matthew J. Kotchen, Erin T. Mansur, and Andrew J. Yates. 2022. "Why marginal CO₂ emissions are not decreasing for US electricity: Estimates and implications for climate policy." *Proceedings of the National Academy of Sciences* 119, no. 8 (February 22, 2022): e2116632119. ISSN: 0027-8424, 1091-6490, accessed June 10, 2024. https://doi.org/10.1073/pnas.2116632119. https://pnas.org/doi/full/10.1073/ pnas.2116632119.
- Holland, Stephen P., Erin T. Mansur, Nicholas Z. Muller, and Andrew J. Yates. 2016. "Are There Environmental Benefits from Driving Electric Vehicles? The Importance of Local Factors." *American Economic Review* 106, no. 12 (December 1, 2016): 3700–3729. ISSN: 0002-8282, accessed June 10, 2024. https://doi.org/10.1257/aer. 20150897. https://pubs.aeaweb.org/doi/10.1257/aer.20150897.

- IPCC. 2022. Global Warming of 1.5°C: IPCC Special Report on Impacts of Global Warming of 1.5°C above Pre-industrial Levels in Context of Strengthening Response to Climate Change, Sustainable Development, and Efforts to Eradicate Poverty. 1st ed. Cambridge University Press, June 9, 2022. ISBN: 978-1-00-915794-0 978-1-00-915795-7, accessed June 5, 2024. https://doi.org/10.1017/9781009157940. https://www.cambridge.org/ core/product/identifier/9781009157940/type/book.
- **IRENA.** 2017. *Electricity storage and renewables: costs and markets to 2030.* OCLC: 1393122335. Abu Dhabi: International Renewable Energy Agency.
- **IWG.** 2021. Technical Support Document: Social Cost of Carbon, Methane, Washington, D.C. (United States): The Interagency Working Group.
- Joskow, Paul L, and Catherine D Wolfram. 2012. "Dynamic Pricing of Electricity." Public Utilities Fortnightly 102 (3): 381–385. https://doi.org/10.1257/aer.102.3.381.
- Joskow, Paul L. 2007. "Regulation of Natural Monopoly." In Handbook of law and economics, 2:1227–1348.
- Joskow, Paul L. 2020. "Transmission Capacity Expansion Is Needed to Decarbonize the Electricity Sector Efficiently." *Joule* 4, no. 1 (January): 1–3. ISSN: 25424351, accessed June 10, 2024. https://doi.org/10.1016/j.joule.2019.10.011. https://linkinghub.elsevier. com/retrieve/pii/S2542435119305276.
- Joskow, Paul L., and Richard Schmalensee. 1983. Markets for Power: An Analysis of Electric Utility Deregulation. Cambridge, MA: Massachusetts Inst. of Technology Press.
- Kurdgelashvili, Lado, Junli Li, Cheng-Hao Shih, and Benjamin Attia. 2016. "Estimating technical potential for rooftop photovoltaics in California, Arizona and New Jersey." *Renewable Energy* 95 (September): 286–302. ISSN: 09601481, accessed June 5, 2024. https://doi.org/10.1016/j.renene.2016.03.105. https://linkinghub.elsevier.com/ retrieve/pii/S0960148116302890.
- Lazar, Jim. 2016. "Use Great Caution in Design of Residential Demand Charges." Natural Gas & Electricity 32, no. 7 (February): 13–19. ISSN: 15457893, accessed September 15, 2023. https://doi.org/10.1002/gas.21884. https://onlinelibrary.wiley.com/doi/10.1002/ gas.21884.
- Li, Mo, Timothy M. Smith, Yi Yang, and Elizabeth J. Wilson. 2017. "Marginal Emission Factors Considering Renewables: A Case Study of the U.S. Midcontinent Independent System Operator (MISO) System." *Environmental Science & Technology* 51, no. 19 (October 3, 2017): 11215–11223. ISSN: 0013-936X, 1520-5851, accessed June 5, 2024. https://doi.org/10.1021/acs.est.7b00034. https://pubs.acs.org/doi/10.1021/acs. est.7b00034.

- Lokshin, Michael, and Zurab Sajaia. 2004. "Maximum Likelihood Estimation of Endogenous Switching Regression Models." The Stata Journal: Promoting communications on statistics and Stata 4, no. 3 (August): 282–289. ISSN: 1536-867X, 1536-8734, accessed June 10, 2024. https://doi.org/10.1177/1536867X0400400306. http://journals.sagepub.com/doi/10.1177/1536867X0400400306.
- Maddala, Gangadharrao S. 1983. Limited-dependent and qualitative variables in econometrics. Cambridge University Press.
- Matek, Benjamin, and Karl Gawell. 2015. "The Benefits of Baseload Renewables: A Misunderstood Energy Technology." The Electricity Journal 28, no. 2 (March): 101– 112. ISSN: 10406190, accessed June 6, 2024. https://doi.org/10.1016/j.tej.2015.02.001. https://linkinghub.elsevier.com/retrieve/pii/S104061901500024X.
- McLaren, Joyce, Seth Mullendore, Nicholas Laws, and Kate Anderson. 2017. "Identifying Potential Markets for Behind-the-Meter Battery Energy Storage: A Survey of U.S. Demand Charges."
- Minnesota Department of Commerce. 2014. Minnesota Value of Solar: Methodology. Minnesota Department of Commerce, Division of Energy Resources. https://mn.gov/ commerce/energy/images/MN-VOS-Methodology-FINAL.pdf.
- Mongird, K, V Viswanathan, P Balducci, J Alam, V Fotedar, V Koritarov, and B Hadjerioua. 2019. "Energy Storage Technology and Cost Characterization Report" [in en].
- Mountain, Dean C., and Cheng Hsiao. 1986. "Peak and Off-Peak Industrial Demand for Electricity: The Hopkinson Rate in Ontario, Canada." *The Energy Journal* 7, no. 1 (January 1, 1986). ISSN: 01956574, accessed September 15, 2023. https://doi.org/10. 5547/ISSN0195-6574-EJ-Vol7-No1-10. http://www.iaee.org/en/publications/ejarticle. aspx?id=1755.
- **MSA.** 2014. Alberta Retail Markets for Electricity and Natural Gas: A description of basic structural features.
- Muehlenbachs, Lucija, and David Brown. 2023. The Value of Electricity Reliability: Evidence from Battery Adoption. Accessed June 5, 2024. https://doi.org/10.2139/ssrn. 4509955. https://www.ssrn.com/abstract=4509955.
- NCCETC. 2023. 50 States of Solar: 2023 Quarterly Report. Raleigh, NC (United States): North Carolina Clean Energy Technology Center. https://nccleantech.ncsu.edu/2023/ 04/19/the-50-states-of-solar-states-move-toward-time-varying-or-avoided-costcompensation-for-distributed-generation-in-q1-2023/.

- Neufeld, John L. 1987. "Price Discrimination and the Adoption of the Electricity Demand Charge." The Journal of Economic History 47, no. 3 (September): 693–709. ISSN: 0022-0507, 1471-6372, accessed September 15, 2023. https://doi.org/10.1017/S002205070004 9068. https://www.cambridge.org/core/product/identifier/S0022050700049068/type/ journal_article.
- NOAA. 2023. Environmental Information's Integrated Surface Dataset (ISD). Accessed June 30, 2023. https://www.ncei.noaa.gov/access/metadata/landing-page/bin/iso?id=gov.noaa.ncdc:C00532.
- NREL. 2014. Commercial and Residential Hourly Load Profiles for all TMY3 Locations in the United States. Accessed April 30, 2023. https://catalog.data.gov/dataset/ commercial-and-residential-hourly-load-profiles-for-all-tmy3-locations-in-the-unitedstate-bbc75.
- **NREL.** 2020. 2018 Renewable Energy Grid Integration Data Book. Golden, CO (United States): National Renewable Energy Lab.(NREL).
- NREL. 2022. Maximum demand charge rates for commercial and industrial electricity tariffs in the United States. Golden, CO (US). https://doi.org/10.7799/1392982. https: //data.nrel.gov/submissions/74.
- Öhrlund, Isak, Mårten Schultzberg, and Cajsa Bartusch. 2019. "Identifying and estimating the effects of a mandatory billing demand charge." *Applied Energy* 237 (March): 885–895. ISSN: 03062619, accessed September 15, 2023. https://doi.org/ 10.1016/j.apenergy.2019.01.028. https://linkinghub.elsevier.com/retrieve/pii/ S0306261919300285.
- **Open EI.** 2023. *The Utility Rate Database*. Accessed April 30, 2023. https://openei.org/wik i/Utility_Rate_Database#:~:text=The%20Utility%20Rate%20Database%20(URDB, of%20Energy's%20Energy%20Information%20Administration..
- Passey, Robert, Navid Haghdadi, Anna Bruce, and Iain MacGill. 2017. "Designing more cost reflective electricity network tariffs with demand charges." *Energy Policy* 109 (October): 642–649. ISSN: 03014215, accessed September 15, 2023. https://doi. org/10.1016/j.enpol.2017.07.045. https://linkinghub.elsevier.com/retrieve/pii/ S0301421517304767.
- Phillips, Tyler Bennett. 2014. "Dynamic Rating of Transmission Lines for Improved Wind Energy Integration in Complex Terrain." PhD diss., Boise State University.
- **Potomac Economics.** 2023a. 2021 State of the Market Report for the ERCOT Electricity Markets.
- **Potomac Economics.** 2023b. 2022 State of the Market Report for the ERCOT Electricity Markets.

- **Potomac Economics.** 2024. 2023 State of the Market Report for the ERCOT Electricity Markets.
- Quick Electricity. 2022. "Solar Buyback Plans and Net Metering Incentives in Texas." Accessed April 30, 2022. https://quickelectricity.com/2018-solar-panel-incentivestexas-net-metering-buyback-programs/.
- **REN21.** 2024. Renewables 2024 Global Status Report Collection, Global Overview.
- Rhodes, Joshua, Thomas Deetjen, and Caitlin Smith. 2021. Impacts of Large, Flexible Data Center Operations on the Future of ERCOT. Lancium Balancing Energy.
- Rubin, Scott J. 2015. "Moving Toward Demand-Based Residential Rates." The Electricity Journal 28, no. 9 (November): 63–71. ISSN: 10406190, accessed September 15, 2023. https://doi.org/10.1016/j.tej.2015.09.021. https://linkinghub.elsevier.com/retrieve/ pii/S1040619015002043.
- Rugolo, Jason, and Michael J. Aziz. 2012. "Electricity storage for intermittent renewable sources." *Energy & Environmental Science* 5 (5): 7151. ISSN: 1754-5692, 1754-5706, accessed June 5, 2024. https://doi.org/10.1039/c2ee02542f. https://xlink.rsc.org/?DOI=c2ee02542f.
- Sapio, Alessandro. 2015. "The effects of renewables in space and time: A regime switching model of the Italian power price." *Energy Policy* 85 (October): 487–499. ISSN: 03014215, accessed June 10, 2024. https://doi.org/10.1016/j.enpol.2015.07.025. https:// linkinghub.elsevier.com/retrieve/pii/S0301421515300367.
- Seeto, Dewey. 1997. "Time-of-use rates vs. Hopkinson tariffs redux: An analysis of the choice of rate structures in a regulated electricity distribution company." *Energy Economics* 19 (2): 169–185.
- **SEIA.** 2022. Solar Market Insight Report 2021 Year in Review. Washington, D.C. (United States): Solar Energy Industries Association.
- **Serecon.** 2014. Economic Impact of Agriculture in Lethbridge County.
- Sexton, Steven, A. Justin Kirkpatrick, Robert Harris, and Nicholas Muller. 2018. Heterogeneous Environmental and Grid Benefits from Rooftop Solar and the Costs of Inefficient Siting Decisions. w25241. Cambridge, MA: National Bureau of Economic Research, November. Accessed June 10, 2024. https://doi.org/10.3386/w25241. http: //www.nber.org/papers/w25241.pdf.
- Siler-Evans, Kyle, Inês Lima Azevedo, and M. Granger Morgan. 2012. "Marginal Emissions Factors for the U.S. Electricity System." *Environmental Science & Technology* 46, no. 9 (May 1, 2012): 4742–4748. ISSN: 0013-936X, 1520-5851, accessed June 6, 2024. https://doi.org/10.1021/es300145v. https://pubs.acs.org/doi/10.1021/es300145v.

- Smartly Energy. 2023. "The Tesla Powerwall: How Do the Alternatives Measure Up?" Accessed May 5, 2024. https://smartlyenergy.com/blog/the-tesla-powerwall-how-do-the-alternatives-measure-up.
- Statistics Canada. 2020. Regional unemployment rates used by the Employment Insurance program, three-month moving average, seasonally adjusted. Accessed May 31, 2021. htt ps://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=1410035401.
- Statistics Canada Government of Canada. 2023. "Labour force characteristics by economic region, three-month moving average, unadjusted for seasonality, last 5 months, inactive." Last Modified: 2021-01-08. Accessed October 17, 2023. https://www150. statcan.gc.ca/t1/tbl1/en/tv.action?pid=1410029301.
- Stokke, Andreas V., Gerard L. Doorman, and Torgeir Ericson. 2010. "An analysis of a demand charge electricity grid tariff in the residential sector." *Energy Efficiency* 3, no. 3 (September): 267–282. ISSN: 1570-646X, 1570-6478, accessed September 15, 2023. https://doi.org/10.1007/s12053-009-9071-9. http://link.springer.com/10.1007/s12053-009-9071-9.
- Stoll, Christian, Lena Klaaßen, Ulrich Gallersdörfer, and Alexander Neumüller. 2023. *Climate Impacts of Bitcoin Mining in the U.S.* MIT Center for Energy and Environmental Policy Research.
- Taylor, Thomas N., and Peter M. Schwarz. 1986. "A Residential Demand Charge: Evidence from the Duke Power Time-of-Day Pricing Experiment." *The Energy Journal* 7, no. 2 (April 1, 1986). ISSN: 01956574, accessed September 15, 2023. https://doi.org/ 10.5547/ISSN0195-6574-EJ-Vol7-No2-10. http://www.iaee.org/en/publications/ ejarticle.aspx?id=1768.
- Taylor, Thomas N., and Peter M. Schwarz. 1990. "The Long-Run Effects of a Time-of-Use Demand Charge." The RAND Journal of Economics 21 (3): 431. ISSN: 07416261, accessed September 15, 2023. https://doi.org/10.2307/2555618. http://doi.wiley.com/ 10.2307/2555618.
- Tibshirani, Robert. 1996. "Regression Shrinkage and Selection Via the Lasso." Journal of the Royal Statistical Society: Series B (Methodological) 58, no. 1 (January): 267–288. ISSN: 0035-9246, 2517-6161, accessed October 17, 2023. https://doi.org/10.1111/j.2517-6161.1996.tb02080.x. https://rss.onlinelibrary.wiley.com/doi/10.1111/j.2517-6161.1996.tb02080.x.
- Trabish, Herman K. 2018. "Are regulators starting to rethink fixed charges?" Utility Dive. Accessed October 17, 2023. https://www.utilitydive.com/news/are-regulators-startingto-rethink-fixed-charges/530417/.
- Turk, Graham, Tim Schittekatte, Pablo Dueñas Martínez, Paul L Joskow, and Richard L Schmalensee. n.d. "Designing distribution network tariffs under increased residential end-user electrification: Can the US learn something from Europe?"

- Veal, Michael R. 1986. "Time-of use rates and peak period electricity consumption." *Energy* economics 8 (4): 257–262.
- Weisbach, David A., and Gilbert E. Metcalf. 2009. "The Design of a Carbon Tax." SSRN Electronic Journal, ISSN: 1556-5068, accessed June 5, 2024. https://doi.org/10. 2139/ssrn.1327260. http://www.ssrn.com/abstract=1327260.
- White, Philip R., Joshua D. Rhodes, Eric J.H. Wilson, and Michael E. Webber. 2021. "Quantifying the impact of residential space heating electrification on the Texas electric grid." *Applied Energy* 298 (September): 117113. ISSN: 03062619, accessed June 5, 2024. https://doi.org/10.1016/j.apenergy.2021.117113. https://linkinghub.elsevier. com/retrieve/pii/S0306261921005559.
- Woerman, Matt. 2018. "Market Size and Market Power: Evidence from the Texas Electricity Market." *Energy Institute Working Paper.*
- Wolak, Frank A. 2010. "An Experimental Comparison of Critical Peak and Hourly Pricing: The PowerCentsDC Program."
- Wolak, Frank A. 2021. "Wholesale electricity market design." In Handbook on Electricity Markets, edited by Jean-Michel Glachant, Paul L. Joskow, and Michael G. Pollitt. Edward Elgar Publishing, November 12, 2021. ISBN: 978-1-78897-995-5 978-1-78897-994-8, accessed June 5, 2024. https://doi.org/10.4337/9781788979955.00010. https: //www.elgaronline.com/view/edcoll/9781788979948/9781788979948.00010.xml.
- Woo, C.K., Brian Horii, and Ira Horowitz. 2002. "The Hopkinson tariff alternative to TOU rates in the Israel Electric Corporation." *Managerial and Decision Economics* 23, no. 1 (January): 9–19. ISSN: 0143-6570, 1099-1468, accessed September 15, 2023. https: //doi.org/10.1002/mde.1040. https://onlinelibrary.wiley.com/doi/10.1002/mde.1040.
- Wood, Lisa, Ross Hemphill, John Howat, Ralph Cavanagh, Severin Borenstein, Jeff Deason, Lisa Schwartz, and Lisa Schwartz. 2016. Recovery of Utility Fixed Costs: Utility, Consumer, Environmental and Economist Perspectives. LBNL-1005742, 1342757. Future Electric Utility Regulation, June 14, 2016. Accessed September 15, 2023. https://doi.org/10.2172/1342757. http://www.osti.gov/servlets/purl/1342757/.
- Wooldridge, Jeffrey M. 2010. Econometric analysis of cross section and panel data. MIT press.
- Xcel Energy. 2023. "Time of Use Rates." Accessed June 1, 2023. https://tx.my.xcelenergy. com/s/business/rate-plans/time-of-use.
- Zethmayr, Jeff, and Ramandeep Singh Makhija. 2019. "Six unique load shapes: A segmentation analysis of Illinois residential electricity consumers." *The Electricity Journal* 32, no. 9 (November): 106643. ISSN: 10406190, accessed June 18, 2024. https://doi.org/10.1016/j.tej.2019.106643. https://linkinghub.elsevier.com/retrieve/pii/S1040619019301800.